SEVENTH RENEWAL TO RECORDS MANAGEMENT MASTER AGREEMENT

THIS SEVENTH RENEWAL ("7th Renewal") is entered into by and between Fort Bend County, ("County"), a body corporate and politic under the laws of the State of Texas, and Manatron, Inc., ("Manatron"), a company authorized to conduct business in the State of Texas (hereinafter each referred to as a "party" or collectively as the "parties").

WHEREAS, County and Manatron previously executed a Master Agreement, Amendment 1 to the Subscription Agreement, Amendment 2 the Subscription Agreement, Renewal to HIPA Plus Hosting Agreement, and the Second Renewal to HIPA Plus Hosting Agreement, Third Renewal To Records Management Master Agreement, Fourth Renewal to Records Management Master Agreement, a Fifth Renewal, and a Sixth Renewal (collectively referred to as the "Agreement"), attached hereto as Exhibit "2" and incorporated fully by reference; and

WHEREAS, County and Manatron now desire to renew the Agreement a seventh time, for a one (1) year term at the price of \$132,766.00, as described in Manatron's Invoice No. MANMN0002372 (dated July 15, 2022), attached hereto and incorporated fully by reference as Exhibit "1"; and

WHEREAS, Manatron is the sole source provider of the products and services described in Manatron's Invoice No. MANMN0002372 (Exhibit 1), as indicated by the letter attached hereto as Exhibit "3" and incorporated fully by reference; and

NOW THEREFORE, in consideration of the foregoing, the Agreement between the County and Manatron are hereby renewed amended as follows:

- 1. County shall renew the Agreement and the purchase the licenses and services described in Exhibit 1 at a total price of \$132,766.00. Manatron will provide license and services as described in the Agreement and in Exhibit 1.
- 2. The term of the Agreement is effective as of October 1, 2022, and shall expire no later than September 30, 2023, unless terminated sooner pursuant to the Agreement. This Agreement shall not automatically renew, but may be subsequently renewed in writing upon agreement of the parties. The parties acknowledge and agree that services were and will be supported by good and valuable consideration during the Term of this Agreement, the sufficiency of which is acknowledged by the parties.
- 3. Payment shall be made by County within thirty (30) days of receipt of invoices. Manatron may submit invoices electronically in a form acceptable to County via: apauditor@fortbendcountytx.gov. County is a body corporate and politic under the laws of the State of Texas and claims exemption from sales and use taxes. A copy of a tax-exempt certificate will be furnished upon request. County will make quarterly payments to Manatron in the amount of \$33,191.50.
- 4. Manatron clearly understands and agrees, such understanding and agreement being of the absolute essence of this Agreement, that County shall have available the total maximum sum of One Hundred Thirty-Two Thousand, Seven Hundred Sixty-Six and 00/100 dollars (\$132,766.00), specifically allocated to fully discharge any and all liabilities County may incur.

Manatron does further understand and agree, said understanding and agreement also being of the absolute essence of this Agreement, that the total maximum compensation that Manatron may become entitled to and the total maximum sum that County may become liable to pay to Manatron shall not under any conditions, circumstances, or interpretations thereof exceed One Hundred Thirty-Two Thousand, Seven Hundred Sixty-Six and 00/100 dollars (\$132,766.00). In no event will the amount paid by the County for all services under this Agreement exceed this Limit of Appropriation without an amendment executed by the parties.

- 5. It is specifically understood and agreed that in the event no funds or insufficient funds are appropriated by County under this 7th Renewal, County shall notify all necessary parties that the 7th Renewal shall thereafter terminate and be null and void on the last day of the fiscal period for which appropriations were made without penalty, liability or expense to County.
- 6. Governing Law.
 - a. The laws of the State of Texas govern all disputes arising out of or relating to the Agreement and this Addendum. The parties hereto acknowledge that venue is proper in Fort Bend County, Texas, for all legal actions or proceedings arising out of or relating to this Addendum and waive the right to sue or be sued elsewhere. Nothing in the Agreement and this Addendum shall be construed to waive the County's sovereign immunity. County does not agree to pay any and/or all attorney fees incurred by Manatron in any way associated with the Agreement.
- 7. Certain State Law Requirements for Contracts. The contents of this Section are required by Texas Law and are included by County regardless of content. For purposes of Sections 2252.152, 2271.002, and 2274.002, Texas Government Code, as amended, Manatron hereby verifies that Manatron and any parent company, wholly owned subsidiary, majority-owned subsidiary, and affiliate:
 - a. Unless affirmatively declared by the United States government to be excluded from its federal sanctions regime relating to Sudan or Iran or any federal sanctions regime relating to a foreign terrorist organization, is not identified on a list prepared and maintained by the Texas Comptroller of Public Accounts under Section 806.051, 807.051, or 2252.153 of the Texas Government Code.
 - b. If employing ten (10) or more full-time employees and this Agreement has a value of \$100,000.00 or more, Manatron does not boycott Israel and is authorized to agree in such contracts not to boycott Israel during the term of such contracts. "Boycott Israel" has the meaning provided in § 808.001 of the Texas Government Code.
 - c. If employing ten (10) or more full-time employees and this Agreement has a value of \$100,000.00 or more, Manatron does not boycott energy companies and is authorized to agree in such contracts not to boycott energy companies during the term of such contracts. "Boycott energy company" has the meaning provided in § 809.001 of the Texas Government Code.
 - d. If employing ten (10) or more full-time employees and this Agreement has a value of \$100,000.00 or more, Manatron does not have a practice, policy, guidance, or directive that discriminates against a firearm entity or firearm trade association and is authorized to agree in such contracts not to discriminate against a firearm entity or firearm trade association during the term of such contracts. "Discriminate against a firearm entity or firearm trade association" has the meaning provided in § 2274.001(3) of the Texas Government Code. "Firearm entity" and "firearm trade association" have the meanings provided in § 2274.001(6) and (7) of the Texas Government Code.

- 8. The Agreement remains in full force and effect. If there is a conflict between this 7th Renewal and the Agreement, the provisions of this 7th Renewal shall prevail.
- 9. By execution of this 7th Renewal, the parties acknowledge that they have read and understood each provision, term and obligation contained in this 7th Renewal. This 7th Renewal, although drawn by one party, shall be construed fairly and reasonably and not more strictly against the drafting party than the nondrafting party.
- 10. The parties to the Agreement and this 7th Renewal agree that the electronic and/or digital signatures of the parties included in the Agreement and this 7th Renewal are intended to authenticate this writing and to have the same force and effect as the use of manual signatures.
- 11. BY ACCEPTANCE OF AGREEMENT, MANATRON ACKNOWLEDGES THAT FORT BEND COUNTY IS OPPOSED TO HUMAN TRAFFICKING AND THAT NO COUNTY FUNDS WILL BE USED IN SUPPORT OF SERVICES OR ACTIVITIES THAT VIOLATE HUMAN TRAFFICKING LAWS.

(Execution Page Follows)

(Remainder of Page Intentionally Left Blank)

IN WITNESS WHEREOF, this 7th Renewal is signed, accepted, and agreed to by all parties by and through the parties or their agents or authorized representatives. All parties hereby acknowledge that they have read and understood this 7th Renewal and the attachments and exhibits hereto. All parties further acknowledge that they have executed this legal document voluntarily and of their own free will.

FORT BEND COUNTY	Manatron, Inc.
	700
KP George, County Judge	Authorized Agent – Signature
	Ross A. D'Elia
Date	Authorized Agent-Printed Name
	VP, Business Operations
ATTEST:	Title
	12/1/2022
Latura Richard, County Clerk	Date
•	ITOR'S CERTIFICATE Int of \$ are available to pay the obligation of contract.
	Robert Ed Sturdivant, County Auditor
and the state of t	N. S. (N. 10000) 273 (J. L. J. L. L. T. 2077)-

Exhibit 1: Manatron's Invoice No. MANMN0002372 (dated July 15, 2022);

Exhibit 22: Master Agreement, Amendment 1 to the Subscription Agreement, Amendment 2 the Subscription Agreement, Renewal to HIPA Plus Hosting Agreement, the Second Renewal to HIPA Plus Hosting Agreement, Third Renewal To Records Management Master Agreement, and Fourth Renewal to Records Management Master Agreement, a Fifth Renewal, and a Sixth Renewal previously executed by the County and Manatron; and Exhibit 3: Sole source letter.

i:\agreements\2020 agreements\county disrk\mannton agreement (20-cocle-100213-a5)\mannton user conference 2012 (20-cocle-100213-a5)\mannton agreement 7th renewal-dock 20-CoCle-100213 aw

Exhibit 1

Invoice No.

MANMN0002372

Date

07/15/2022

Due Date

10/01/2022 4308100-REC

Customer No. Page

1 of 1



TECHNOLOGIES

BIII To

FORT BEND COUNTY CLERK 301 JACKSON RICHMOND, TX, 77469 United States Ship To

FORT BEND COUNTY CLERK 301 JACKSON RICHMOND, TX 77469 United States

Contract/Project Number

Purchase Order

Payment Terms

Currency

MN OCT

HARRIS-USS

Quantity	Item No	Description	Unit Price	Amount
1.00	REC-RECORDER-S	Recorder System Spt: October 2022 to December 2022	10,552.50	10,552.50
1.00	REC-ALINDEX-S	Automated Indexing System Spt: October 2022 to	5,529.75	5,529.75
7,55	1.00	December 2022		
1.00	REC-MARRIAGE-S	Marriage Lic SW Spt: October 2022 to December 2022	1,048.50	1,048.50
1.00	REC-SLA-4	Service Level 4 Support: October 2022 to December 2022	1,572.50	1,572.50
1.00	REC-VITALS-S	Vital Records Support: October 2022 to December 2022	786,25	788.25
1.00	REC-ERECORDING-S	eRecording Support: October 2022 to December 2022	5,413.50	5,413.50
1.00	REC-PUBLIC ACCESS-S	Recorder Public Access Support: October 2022 to	5,655.00	5,855,00
1,00	(CEC-1 CIDED NODECO-C	December 2022		
4.00	REC-NETMINUTES-S	NetMinutes SW Spt. October 2022 to December 2022	2,494.00	2,494.00
1.00	REC-AGENDA LINK-S	AgendaQuick Link SW Spt: October 2022 to December	139.50	139.50
1.00	KEC-WOCHOW FIRM-0	2022		

	Subtotal	33,191.50
	Misc	0,00
Remit To:	Taxes	0.00
Manatron, Inc.	Freight	0.00
("Aumentum Technologies")	Payments/Credits	0.00
PO Box 74008484 Chicago, II 60674-8484	Total	33,191.50



Invoice Questions? Please call or small Shirley Ma at 343-804-9883 or xma@harriscomputer.com

Thank you for your business!

Exhibit 2

§

COUNTY OF FORT BEND

SIXTH RENEWAL TO RECORDS MANAGEMENT MASTER AGREEMENT

THIS SIXTH RENEWAL ("6th Renewal") is entered into by and between Fort Bend County, ("County"), a body corporate and politic under the laws of the State of Texas, and Manatron, Inc., ("Manatron"), a company authorized to conduct business in the State of Texas.

WHEREAS, County and Manatron previously executed a Master Agreement, Amendment 1 to the Subscription Agreement, Amendment 2 the Subscription Agreement, Renewal to HIPA Plus Hosting Agreement, and the Second Renewal to HIPA Plus Hosting Agreement, and Third Renewal To Records Management Master Agreement, and Fourth Renewal to Records Management Master Agreement, and a Fifth Renewal, (collectively referred to as the "Agreement"), attached hereto as Exhibit "II" and incorporated fully by reference; and

WHEREAS, County and Manatron now desire to renew the Agreement a sixth time, for a one (1) year term at the price of \$132,766.00, as described in Manatron's Invoice No. MANMN0001676 (dated December 10, 2021), attached hereto and incorporated fully by reference as Exhibit "I".

NOW THEREFORE, in consideration of the foregoing, the Agreement between the County and Manatron are hereby renewed amended as follows:

- 1. County shall renew the Agreement and the purchase the licenses and services described in Exhibit I at a total price of \$132,766.00. Manatron will provide license and services as described in the Agreement and in Exhibit I.
- 2. The term of the Agreement is effective as of October 1, 2021, and shall expire no later than September 30, 2022, unless terminated sooner pursuant to the Agreement. This Agreement shall not automatically renew, but may be subsequently renewed in writing upon agreement of the parties.
- 3. Payment shall be made by County within thirty (30) days of receipt of invoice. County is a body corporate and politic under the laws of the State of Texas and claims exemption from sales and use taxes. A copy of a tax-exempt certificate will be furnished upon request. County will make quarterly payments to Manatron in the amount of \$33,191.50.
- 4. Manatron clearly understands and agrees, such understanding and agreement being of the absolute essence of this Agreement, that County shall have available the total maximum sum of One Hundred Thirty-Two Thousand, Seven Hundred Sixty-Six and 00/100 dollars (\$132,766.00), specifically allocated to fully discharge any and all liabilities County may incur. Manatron does further understand and agree, said understanding and agreement also being of the absolute essence of this Agreement, that the total maximum compensation that Manatron may become entitled to and the total maximum sum that County may become liable to pay to Manatron shall not under any conditions, circumstances, or interpretations thereof exceed One Hundred Thirty-Two Thousand, Seven Hundred Sixty-Six and 00/100 dollars (\$132,766.00). In no event will the amount paid by the County for all services under this Agreement exceed this Limit of Appropriation without an amendment executed by the parties.
- 5. It is specifically understood and agreed that in the event no funds or insufficient funds are appropriated by County under this 6th Renewal, County shall notify all necessary parties that

the 6th Renewal shall thereafter terminate and be null and void on the last day of the fiscal period for which appropriations were made without penalty, liability or expense to County.

6. Governing Law.

- a. The laws of the State of Texas govern all disputes arising out of or relating to the Agreement and this Addendum. The parties hereto acknowledge that venue is proper in Fort Bend County, Texas, for all legal actions or proceedings arising out of or relating to this Addendum and waive the right to sue or be sued elsewhere. Nothing in the Agreement and this Addendum shall be construed to waive the County's sovereign immunity.
- 7. Certain State Law Requirements for Contracts. The contents of this Section are required by Texas Law and are included by County regardless of content. For purposes of Sections 2252.152, 2271.002, and 2274.002, Texas Government Code, as amended, Manatron hereby verifies that Manatron and any parent company, wholly owned subsidiary, majority-owned subsidiary, and affiliate:
 - a. Unless affirmatively declared by the United States government to be excluded from its federal sanctions regime relating to Sudan or Iran or any federal sanctions regime relating to a foreign terrorist organization, is not identified on a list prepared and maintained by the Texas Comptroller of Public Accounts under Section 806.051, 807.051, or 2252.153 of the Texas Government Code.
 - b. If employing ten (10) or more full-time employees and this Agreement has a value of \$100,000.00 or more, Manatron does not boycott Israel and is authorized to agree in such contracts not to boycott Israel during the term of such contracts. "Boycott Israel" has the meaning provided in § 808.001 of the Texas Government Code.
 - c. If employing ten (10) or more full-time employees and this Agreement has a value of \$100,000.00 or more, Manatron does not boycott energy companies and is authorized to agree in such contracts not to boycott energy companies during the term of such contracts. "Boycott energy company" has the meaning provided in § 809.001 of the Texas Government Code.
 - d. If employing ten (10) or more full-time employees and this Agreement has a value of \$100,000.00 or more, Manatron does not have a practice, policy, guidance, or directive that discriminates against a firearm entity or firearm trade association and is authorized to agree in such contracts not to discriminate against a firearm entity or firearm trade association during the term of such contracts. "Discriminate against a firearm entity or firearm trade association" has the meaning provided in § 2274.001(3) of the Texas Government Code. "Firearm entity" and "firearm trade association" have the meanings provided in § 2274.001(6) and (7) of the Texas Government Code.
- 8. The Agreement remains in full force and effect. If there is a conflict between this 6th Renewal and the Agreement, the provisions of this 6th Renewal shall prevail.
- 9. By execution of this 6th Renewal, the parties acknowledge that they have read and understood each provision, term and obligation contained in this 6th Renewal. This 6th Renewal, although drawn by one party, shall be construed fairly and reasonably and not more strictly against the drafting party than the nondrafting party.
- 10. The parties to the Agreement and this 6th Renewal agree that the electronic and/or digital signatures of the parties included in the Agreement and this 6th Renewal are intended to authenticate this writing and to have the same force and effect as the use of manual signatures.
- 11. BY ACCEPTANCE OF AGREEMENT, MANATRON ACKNOWLEDGES THAT FORT BEND COUNTY IS OPPOSED TO HUMAN TRAFFICKING AND THAT NO COUNTY

FUNDS WILL BE USED IN SUPPORT OF SERVICES OR ACTIVITIES THAT VIOLATE HUMAN TRAFFICKING LAWS.

(Execution Page Follows)

(Remainder of Page Intentionally Left Blank)

IN WITNESS WHEREOF, this 6th Renewal is signed, accepted, and agreed to by all parties by and through the parties or their agents or authorized representatives. All parties hereby acknowledge that they have read and understood this 6th Renewal and the attachments and exhibits hereto. All parties further acknowledge that they have executed this legal document voluntarily and of their own free will.

FORT BEND COUNTY

MANATRON, INC.

Laura Richard, County Clerk

MANATRON, INC.

Authorized Agent – Signature

Ross A. D'Elia

Authorized Agent- Printed Name

Vice President, Business Operations

Title

2/22/2022

Date

AUDITOR'S CERTIFICATE

I hereby certify that funds in the amount of \$132,766.00 are available to pay the obligation of Fort Bend County within the foregoing contract.

Robert Ed Sturdivant, County Auditor

Exhibit I: Manatron's Invoice No. MANMN0001676 (dated December 10, 2021); and Exhibit II: Master Agreement, Amendment 1 to the Subscription Agreement, Amendment 2 the Subscription Agreement, Renewal to HIPA Plus Hosting Agreement, and the Second Renewal to HIPA Plus Hosting Agreement, and Third Renewal To Records Management Master Agreement, and Fourth Renewal to Records Management Master Agreement, and a Fifth Renewal previously executed by the County and Manatron

 $i:\agreements\2020\ agreements\county\ clerk\ manatron\ agreement\ (20-coclk-100213-a5)\ manatron,\ inc.\ (20-coclk-100213-a6)\ v.3\ manatron\ agreement\ 6th\ renewal.docx\ 20-CoClk-100213\ aw$

Exhibit I

Invoice No.

MANMN0001676

Date

12/10/2021

Due Date

10/1/2021 Customer No. 4308100-REC

Page

1 of 1



TECHNOLOGIES

Bill To

FORT BEND COUNTY CLERK ALICIA LOSOYA/LAURA RICHARD CO CLK 301 JACKSON RICHMOND, TX 77469 United States

Ship To

FORT BEND COUNTY CLERK 301 JACKSON RICHMOND, TX 77469 United States

Contract/Project Number	Purchase Order	Payment Terms		Currency
	OCT-DEC2021/RBL	Start of Maint Period		HARRIS-US\$
Item No	Description	Quantity	Unit Price	Amount
REC-RECORDER-S	Recorder System Spt: October 2021 to December 202:	1 1.00	10,552.50	10,552.50
REC-AI INDEX-S	Automated Indexing System Spt: October 2021 to December 2021	1.00	5,529.75	5,529.75
REC-MARRIAGE-S	Marriage Lic SW Spt: October 2021 to December 2021	1.00	1,048.50	1,048.50
REC-SLA-4	Service Level 4 Support: October 2021 to December 20		1,572.50	1,572.50
REC-VITALS-S	Vital Records Support: October 2021 to December 202		786.25	786.25
REC-ERECORDING-S	eRecording Support: October 2021 to December 2021	1,00	5,413.50	5,413.50
REC-PUBLIC ACCESS-S	Aumentum Public Access Support: October 2021 to December 2021	1.00	5,655.00	5,655.00
REC-NETMINUTES-S	NetMinutes SW Spt: October 2021 to December 2021	1.00	2,494.00	2,494.00
REC-AGENDA LINK-S	AgendaQuick Link SW Spt: October 2021 to Decembe 2021	r 1.00	139.50	139.50

	Subtotal	33,191.50
	Misc	0.00
Remit To:	Taxes	0.00
Aumentum Technologies, a division of Manatron, Inc.	Freight	0.00
PO Box 74008484	Total	33,191.50
Chicago, II 60674-8484		



Invoice Questions? Please call or email Renee Fuller at 866-471-2900 ext. 277723 or ar aumentum@harriscomputer.com

Thank you for your business!



2290 Lucien Way #330, Maitland FL, 32751. 330-319-5393 rdelia@harriscomputer.com

ATTN: Laura Richard Fort Bend County Clerk 301 Jackson Street Richmond, Texas 77469-3108

To Whom it May Concern:

On behalf of Manatron, Inc., I wish to thank Fort Bend County for its continuing use of our suite of records management software and services.

Manatron's Aumentum Recorder suite of records management software is currently deployed in the Fort Bend County Clerk's office.

Manatron utilizes a proprietary integration program to tightly integrate workflow processing within Aumentum Recorder under a common user interface. Support of these integrated components is available only through Manatron to Fort Bend County on a sole source basis. Enhancements and augmentation to the core software or its integration components may only be obtained from Manatron on a sole source basis. Manatron does not support authorized distributor or dealership programs and supports the end users of its product directly.

In addition, Manatron offers a menu of specialized services that utilize domain knowledge to access Aumentum Recorder software modules and program code.

Thank you again for the opportunity to serve Fort Bend County.

Yours truly,

QAQQ. BOSS A D'ELIA

VP, OPERATIONS

Exhibit II

§

COUNTY OF FORT BEND

FIFTH RENEWAL TO RECORDS MANAGEMENT MASTER AGREEMENT

THIS FIFTH RENEWAL ("5th Renewal") is entered into by and between Fort Bend County, ("County"), a body corporate and politic under the laws of the State of Texas, and Manatron, Inc., ("Manatron"), a company authorized to conduct business in the State of Texas.

WHEREAS, County and Manatron previously executed a Master Agreement, Amendment 1 to the Subscription Agreement, Amendment 2 the Subscription Agreement, Renewal to HIPA Plus Hosting Agreement, and the Second Renewal to HIPA Plus Hosting Agreement and Third Renewal To Records Management Master Agreement and Fourth Renewal, collectively referred to as the "Agreement" and incorporated by reference; and

WHEREAS, County and Manatron now desire to renew the Agreement a fifth time, for a one (1) year term at the price of \$132,766 as described in Maintenance and Support Schedule and/or Note Schedule dated 10/15/2020, attached hereto and incorporated by reference as Exhibit "1".

NOW THEREFORE, in consideration of the foregoing, the Agreement between the County and Manatron are hereby renewed amended as follows:

- 1. County shall renew the Agreement and the purchase the additional licenses and services at a total price of \$132,766, as provided in Exhibit 1.
- 2. Payment shall be made by County within thirty (30) days of receipt of invoice. County is a body corporate and politic under the laws of the State of Texas and claims exemption from sales and use taxes. A copy of a tax-exempt certificate will be furnished upon request.
- 3. It is specifically understood and agreed that in the event no funds or insufficient funds are appropriated by County under this 5th Renewal, County shall notify all necessary parties that the 5th Renewal shall thereafter terminate and be null and void on the last day of the fiscal period for which appropriations were made without penalty, liability or expense to County.
- Governing Law.
 - a. The laws of the State of Texas govern all disputes arising out of or relating to the Agreement and this Addendum. The parties hereto acknowledge that venue is proper in Fort Bend County, Texas, for all legal actions or proceedings arising out of or relating to this Addendum and waive the right to sue or be sued elsewhere. Nothing in the Agreement and this Addendum shall be construed to waive the County's sovereign immunity.
 - b. As required by Chapter 2271, Government Code, Manatron hereby verifies that it does not boycott Israel and will not boycott Israel through the term of the Agreement and this Addendum. For purposes of this verification, "boycott Israel" means refusing to deal with, terminating business activities with, or otherwise taking any action that is intended to penalize, inflict economic harm on, or limit commercial relations specifically with Israel, or with a person or entity doing business in Israel or in an Israeli-controlled territory, but does not include an action made for ordinary business purposes.
 - c. By signature below, Manatron represents pursuant to Section 2252.152 of the Texas Government Code, that Manatron is not listed on the website of the Comptroller of the

State of Texas concerning the listing of companies that are identified under Section 806.051, Section 807.051 or Section 2252.153

- 5. The Agreement remains in full force and effect. If there is a conflict between this 5th Renewal and the Agreement, the provisions of this 5th Renewal shall prevail.
- 6. By execution of this 5th Renewal, the parties acknowledge that they have read and understood each provision, term and obligation contained in this 5th Renewal. This 5th Renewal, although drawn by one party, shall be construed fairly and reasonably and not more strictly against the drafting party than the nondrafting party.
- 7. The parties to the Agreement and this 5th Renewal agree that the electronic and/or digital signatures of the parties included in the Agreement and this 5th Renewal are intended to authenticate this writing and to have the same force and effect as the use of manual signatures.
- 8. BY ACCEPTANCE OF AGREEMENT, MANATRON ACKNOWLEDGES THAT FORT BEND COUNTY IS OPPOSED TO HUMAN TRAFFICKING AND THAT NO COUNTY FUNDS WILL BE USED IN SUPPORT OF SERVICES OR ACTIVITIES THAT VIOLATE HUMAN TRAFFICKING LAWS.

IN WITNESS WHEREOF, this 5th Renewal is signed, accepted, and agreed to by all parties by and through the parties or their agents or authorized representatives. All parties hereby acknowledge that they have read and understood this 5th Renewal and the attachments and exhibits hereto. All parties further acknowledge that they have executed this legal document voluntarily and of their own free will.

FORT BEND COUNTY		Manatron, Inc.
COL OD TYL		Aud Criama
Cutany magnitive countries.		Authorized Agent – Signature
KP George, County Judge		
12-1-2020		Scot Crismon
12-1-2020	NERS XV.	Authorized Agent-Printed Name
Date	No.	Ü
•	製るる情	Executive V.P.
	製物が	Title
ATTEST:	SEND COUNTRIL	
Juna Kichard		10.28,2020
		Date
Laura Richard, County Clerk		

AUDITOR'S CERTIFICATE

I hereby certify that funds in the amount of \$\frac{132,766.00}{} are available to pay the obligation of Fort Bend County within the foregoing contract.

Robert Ed Sturdivant, County Auditor

Exhibit 1



Maintenance and Support Schedule and/or Note Schedule

For Internal Use Only 0018967 10/15/2020 MAN14710QTR

ACCT#

4308100

FORT BEND COUNTY CLERK 301 JACKSON

ATTN: PURCHASING DEPT RICHMOND TX 77469

THIS IS NOT AN INVOICE!

OCT-SEPT

1.00 MQ	QUARTERLY MAINTENANCE & SUPPORT	\$0.00	\$0.00
1.00 PER	THE PERIOD 10/01/20-12/31/20 FOR	\$0.00	\$0.00
1.00 CON	CONTRACT #TX071817FBC FOR	\$0.00	\$0.00
1.00 AUM-RECORDER-S	A2 Recorder System Spt	\$10,552.50	\$10,552.50
1.00 AUM-ERECORDING-S	eRecording Support	\$5,413.50	\$5,413.50
1.00 AUM-PUBLIC ACCESS-S	Public Access Spt	\$5,655.00	\$5,655.00
1.00 AUM-VITALS-S	Vital Records Spt	\$786.25	\$786.25
1.00 AI INDEX-S	Automated Indexing System Spt	\$5,529.75	\$5,529.75
1.00 AUM-REDACTION-S	A2 Automated Redaction System Spt	\$0.00	\$0.00
1.00 NETMINUTES-S	NetMinutes SW Spt	\$2,494.00	\$2,494.00
1.00 AGENDA LINK-S	AgendaQuick Link SW Spt *INCLUDED*	\$0.00	\$0.00
1.00 VERITY-S	Verity Lic SW Spt *INCLUDED*	\$0.00	\$0.00
1.00 LEAD TOOLS-S	Lead Tools SW Spt (55) *INCLUDED*	\$0.00	\$0.00
1.00 PIXTOOL-S	Pix Tools Software Spt (26) *INCLUDED*	\$0.00	\$0.00
1.00 AUM-MARRIAGE-S	Marriage Lic SW Spt	\$1,048.50	\$1,048.50
1.00 SLA-4	Service Level 4 Support	\$1,572.50	\$1,572.50
1.00 AGENDA LINK-S	AgendaQuick Link SW Spt-Archival Module	\$139.50	\$139.50

Subtotal \$33,191.50
Tax \$0.00
Total \$33,191.50

FOURTH RENEWAL TO RECORDS MANAGEMENT MASTER AGREEMENT

THIS FOURTH RENEWAL ("4th Renewal") is entered into by and between Fort Bend County, ("County"), a body corporate and politic under the laws of the State of Texas, and TRTA, Inc. – A Thomson Reuters Business ("Thomson Reuters Tax & Accounting, Government" or "TRTA"), a company authorized to conduct business in the State of Texas.

WHEREAS, County and TRTA previously executed a Master Agreement, Amendment 1 to the Subscription Agreement, Amendment 2 the Subscription Agreement, Renewal to HIPA Plus Hosting Agreement, and the Second Renewal to HIPA Plus Hosting Agreement and Third Renewal To Records Management Master Agreement, collectively refered to as "Agreement" and incorporated by reference; and

WHEREAS, County and TRTA now desire to renew the Agreement a fourth time, for a one (1) year term at the price of \$127,655.28 as described in Maintenance and Support Schedule and/or Note Schedule dated 10/15/2019, attached hereto and incorporated by reference as Exhibit "1".

NOW THEREFORE, in consideration of the foregoing, the Agreement between the County and TRTA are hereby renewed amended as follows:

- County shall renew the Agreement and the purchase the additional licenses and services at a total price of \$127,655.28, as provided in Exhibit 1.
- Payment shall be made by County within thirty (30) days of receipt of invoice. County is a
 body corporate and politic under the laws of the State of Texas and claims exemption from
 sales and use taxes. A copy of a tax-exempt certificate will be furnished upon request.
- 3. It is specifically understood and agreed that in the event no funds or insufficient funds are appropriated by County under this 4th Renewal, County shall notify all necessary parties that the 4th Renewal shall thereafter terminate and be null and void on the last day of the fiscal period for which appropriations were made without penalty, liability or expense to County.
- 4. Governing Law.
 - a. The laws of the State of Texas govern all disputes arising out of or relating to the Agreement and this Addendum. The parties hereto acknowledge that venue is proper in Fort Bend County, Texas, for all legal actions or proceedings arising out of or relating to this Addendum and waive the right to sue or be sued elsewhere. Nothing in the Agreement and this Addendum shall be construed to waive the County's sovereign immunity.
 - b. As required by Chapter 2270, Government Code, TRTA hereby verifies that it does not boycott Israel and will not boycott Israel through the term of the Agreement and this Addendum. For purposes of this verification, "boycott Israel" means refusing to deal with, terminating business activities with, or otherwise taking any action that is intended to penalize, inflict economic harm on, or limit commercial relations specifically with Israel, or with a person or entity doing business in Israel or in an Israeli-controlled territory, but does not include an action made for ordinary business purposes.
 - c. By signature below, TRTA represents pursuant to Section 2252.152 of the Texas Government Code, that TRTA is not listed on the website of the Comptroller of the State

of Texas concerning the listing of companies that are identified under Section 806.051, Section 807.051 or Section 2253.153

- 5. The Agreement remains in full force and effect. If there is a conflict between this 4th Renewal and the Agreement, the provisions of this 4th Renewal shall prevail.
- 6. By execution of this 4th Renewal, the parties acknowledge that they have read and understood each provision, term and obligation contained in this 4th Renewal. This 4th Renewal, although drawn by one party, shall be construed fairly and reasonably and not more strictly against the drafting party than the nondrafting party.
- 7. The parties to the Agreement and this 4th Renewal agree that the electronic and/or digital signatures of the parties included in the Agreement and this 4th Renewal are intended to authenticate this writing and to have the same force and effect as the use of manual signatures.
- 8. BY ACCEPTANCE OF AGREEMENT, TRTA ACKNOWLEDGES THAT FORT BEND COUNTY IS OPPOSED TO HUMAN TRAFFICKING AND THAT NO COUNTY FUNDS WILL BE USED IN SUPPORT OF SERVICES OR ACTIVITIES THAT VIOLATE HUMAN TRAFFICKING LAWS.

IN WITNESS WHEREOF, this 4th Renewal is signed, accepted, and agreed to by all parties by and through the parties or their agents or authorized representatives. All parties hereby acknowledge that they have read and understood this 4th Renewal and the attachments and exhibits hereto. All parties further acknowledge that they have executed this legal document voluntarily and of their own free will.

FORT BEND COUNTY	MANATRON, INC A THOMSON REUTERS
1000	BUSINESS ("TRTA")
HULLOOSE	Junely Courtes
KP George, County Judge WONERS	Authorized Agént - Signature
12.17.2019	Kimberly Carter Authorized Agent- Printed Name
Date Si Si	THE THE THE TANK THE
ATTEST:	V.P., Finance
Laura Ribard Son CON	Title
Laura Richard, County Clerk	October 29, 2019
•	Date:
	CERTIFICATE 27,655,28 are available to pay the obligation of
Fort Bend County within the foregoing contract	

Robert Ed Sturdivant, County Auditor

i:\agreements\2020 agreements\county clerk\manatron agreement 4th renewal mit 10.21.docx 20-CoClk-160213

Exhibit 1



Maintenance and Support Schedule and/or Note Schedule

For Internal Use Only 0018967 10/15/2019 MAN14710QTR

ACCT#

4308100

FORT BEND COUNTY CLERK 301 JACKSON

ATTN: PURCEASING DEPT RICHMOND TX 77469

THIS IS NOT AN INVOICE!

OCT-SEPT

	QUARTERLY MAINTENANCE & SUPPORT	\$0.00	\$0.00
1.00 MQ	THE PERIOD 10/01/19-12/31/19 FOR	\$0.00	\$0.00
1.00 PER	CONTRACT #TX071817FBC FOR	\$0.00	\$0.00
1.00 CON		\$10,146.45	\$10,346.45
1.00 AUM-RECORDER-S	A2 Recorder System Spt	\$5,205.20	\$5,205.20
1.00 AUM-ERECORDING-S	eRecording Support	\$5,437.35	\$5,437.35
1.00 AUM-PUBLIC ACCESS-S	Public Access Spt	\$755.98	\$755.98
1.00 AUM-VITALS-S	Vital Records Spt Automated Indexing System Spt	\$5,317.02	\$5,317.02
1.00AI INDEX-S	Automated Indexing System Spt A2 Automated Redaction System Spt	\$0.00	\$0.00
1.00 AUM-REDACTION-S		\$2,398.01	\$2,398.01
1.00 netminutes-s	NetMinutes SW Spt	\$0,00	\$0.00
1.00 AGENDA LINK-S	AgendaQuick Link SW Spt *INCLUDED*	\$0.00	\$0.00
1.00 verity-s	Verity Lic SW Spt *INCLUDED* Lead Tools SW Spt (55) *INCLUDED*	\$0.00	\$0.00
1.00 LEAD TOOLS-S	Pix Tools Software Spt (26) *INCLUDED*	\$0.00	\$0.00
1.00 PIXTOOL-S		\$1,007.96	\$1,007.96
1.00 AUM-MARRIAGE-9	Marriage Lic SW Spt	\$1,511.95	\$1,511.95
1.00 SLA-4 1.00 AGENDA LINK-S	Service Level 4 Support AgendaQuick Link SW Spt-Archival Module	\$133,90	\$133.90

Terms:\$120,358.09-and	ual]	Ly(2017-2018)/\$123,952.77-
annually (2018-2019)	and	\$127,655.29-annually
(2019-2020)		

 subtotal
 \$31,913.82

 Tax
 \$0.00

 Total
 \$31,913.82



Laura Richard Fort Bend County, County Clerk 301 Jackson Street Richmond, Texas 77469-3108

Re: Sole Source Statement

Dear Ms. Richard:

On behalf of Manatron, Inc., I wish to thank Fort Bend County for its continuing use of our suite of records management software and services.

Manatron's Aumentum Recorder suite of records management software is currently deployed in the Fort Bend County Clerk's office.

Manatron utilizes a proprietary integration program to tightly integrate workflow processing within Aumentum Recorder under a common user interface. Support of these integrated components is available only through Manatron to Fort Bend County on a sole source basis. Enhancements and augmentation to the core software or its integration components may only be obtained from Manatron on a sole source basis. Manatron does not support authorized distributor or dealership programs and supports the end users of its product directly.

In addition, Manatron offers a menu of specialized services that utilize domain knowledge to access Aumentum Recorder software modules and program code.

If there are any questions on any of these software systems or services, contact Geoff Scheuber at 562-754-3602 or by email at Geoff.Scheuber@thomsonreuters.com

Thank you again for the opportunity to serve Fort Bend County.

Sincerely.

Matthew Henry

Lead Contract Administrator

Manatron, Inc. 269.388,2633

Matt.Henry@TR.com

510 East Milham Avenue Portage, Michigan 49002 (866) 471-2900 § S S

COUNTY OF FORT BEND

THIRD RENEWAL TO RECORDS MANAGEMENT MASTER AGREEMENT

THIS THIRD RENEWAL ("3RD Renewal") is entered into by and between Fort Bend County, ("County"), a body corporate and politic under the laws of the State of Texas, and TRTA, Inc. -A Thomson Reuters Business ("Thomson Reuters Tax & Accounting, Government" or "TRTA"), a company authorized to conduct business in the State of Texas.

WHEREAS, County and TRTA previously executed a Master Agreement, Amendment 1 to the Subscription Agreement, Amendment 2 the Subscription Agreement, Renewal to HIPA Plus Hosting Agreement, and the Second Renewal to HIPA Plus Hosting Agreement (collectively referred to as the "Agreement" attached hereto and incorporated by reference as Exhibit "1"); and

WHEREAS, County and TRTA now desire to renew the Agreement a third time, for a one (1) year term at the price of \$123,952.77, as described in LOA TX071817FBC, attached hereto and incorporated by reference as Exhibit "2".

NOW THEREFORE, in consideration of the foregoing, the Agreement between the County and TRTA are hereby renewed amended as follows:

- 1. County shall renew the Agreement and the purchase the additional licenses and services at a total price of \$123,952.77, as provided in Exhibit 2.
- 2. Payment shall be made by County within thirty (30) days of receipt of invoice. County is a body corporate and politic under the laws of the State of Texas and claims exemption from sales and use taxes. A copy of a tax-exempt certificate will be furnished upon request.
- 3. It is specifically understood and agreed that in the event no funds or insufficient funds are appropriated by County under this 3rd Renewal, County shall notify all necessary parties that the 3rd Renewal shall thereafter terminate and be null and void on the last day of the fiscal period for which appropriations were made without penalty, liability or expense to County.
- 4. Governing Law.
 - a. The laws of the State of Texas govern all disputes arising out of or relating to the Agreement and this Addendum. The parties hereto acknowledge that venue is proper in Fort Bend County, Texas, for all legal actions or proceedings arising out of or relating to this Addendum and waive the right to sue or be sued elsewhere. Nothing in the Agreement and this Addendum shall be construed to waive the County's sovereign immunity.
 - b. As required by Chapter 2270, Government Code, TRTA hereby verifies that it does not boycott Israel and will not boycott Israel through the term of the Agreement and this Addendum. For purposes of this verification, "boycott Israel" means refusing to deal with, terminating business activities with, or otherwise taking any action that is intended to penalize, inflict economic harm on, or limit commercial relations specifically with Israel, or with a person or entity doing business in Israel or in an Israeli-controlled territory, but does not include an action made for ordinary business purposes.
 - c. By signature below, TRTA represents pursuant to Section 2252.152 of the Texas Government Code, that TRTA is not listed on the website of the Comptroller of the State

of Texas concerning the listing of companies that are identified under Section 806.051, Section 807.051 or Section 2253.153

- 5. The Agreement remains in full force and effect. If there is a conflict between this 3rd Renewal and the Agreement, the provisions of this 3rd Renewal shall prevail.
- 6. By execution of this 3rd Renewal, the parties acknowledge that they have read and understood each provision, term and obligation contained in this 3rd Renewal. This 3rd Renewal, although drawn by one party, shall be construed fairly and reasonably and not more strictly against the drafting party than the nondrafting party.
- 7. The parties to the Agreement and this 3rd Renewal agree that the electronic and/or digital signatures of the parties included in the Agreement and this 3rd Renewal are intended to authenticate this writing and to have the same force and effect as the use of manual signatures.

IN WITNESS WHEREOF, this 3rd Renewal is signed, accepted, and agreed to by all parties by and through the parties or their agents or authorized representatives. All parties hereby acknowledge that they have read and understood this 3rd Renewal and the attachments and exhibits hereto. All parties further acknowledge that they have executed this legal document voluntarily and of their own free will.

FORT BEND COUNTY	Manatron, Inc. – A Thomson Reuters Business ("TRTA")
KP George, County Judge	Authorized Agent - Signature
Date	Kimberly Carter Authorized Agent- Printed Name
ATTEST:	V.P. Finance Title
Laura Richard, County Clerk	3.7.2019 Date
AUDITO	OR'S CERTIFICATE
I hereby certify that funds in the amount of Fort Bend County within the foregoing con	of \$ are available to pay the obligation of atract.
	Robert Ed Sturdivant, County Auditor

Exhibit 2



MAINTENANCE AND SUPPORT SERVICES FOR OCTOBER 1, 2018 - SEPTEMBER 30, 2019

Description	Annual Price
A CAMPAGE AND A	\$39,403.69
Aumentum Recorder System	\$20,214.35
eRecording	\$21,115.90
Public Access	\$2,935.85
Vital Records	\$20,648.64
Automated Indexing System	Included
Automated Redaction System	\$9,312.66
NetMinutes Software	Included
AgendaQuick Link Software	- a contract of the contract o
Verity License	Included
LeadTools Software (55)	Included
Pix Tools Software (26)	Included
Marriage License Software	\$3,914,42
Service Level 4	\$5,871.65
AgendaQuick Link Software - Archival Module	\$535.60
Agenda Culor Link Objects Total Annual Price	\$123,952.77

Manatron, Inc. - A Thomson Reuters Business

LOA No. TX071817FBC Expiration Date: November 30, 2017

Page 2 of 3

S

SECOND RENEWAL TO HIPA PLUS HOSTING AGREEMENT

THIS SECOND RENEWAL ("2nd Renewal") is entered into by and between Fort Bend County, ("County"), a body corporate and politic under the laws of the State of Texas, and Manatron, Inc., a Thomson Reuters Business, ("Manatron"), a company authorized to conduct business in the State of Texas.

WHEREAS, County and Manatron previously executed a Master Agreement, Amendment 1 to the Subscription Agreement, Amendment 2 the Subscription Agreement, and the Renewal to HIPA Plus Hosting Agreement (collectively referred to as the "Agreement"). County and Manatron now desire to renew the Agreement a second time, for a one (1) year term at the price described in the Letter of Autorization attached hereto as "Attachment A".

NOW, THEREFORE, in consideration of the foregoing, the Agreement between the County and Manatron are hereby renewed amended as follows:

- 1. County shall renew the Agreement, subject to the terms set forth in Attachment A.
- 2. Payment shall be made by County within thirty (30) days of receipt of invoice. County is a body corporate and politic under the laws of the State of Texas and claims exemption from sales and use taxes. A copy of a tax-exempt certificate will be furnished upon request.
- 3. It is specifically understood and agreed that in the event no funds or insufficient funds are appropriated by County under this 2nd Renewal, County shall notify all necessary parties that the 2nd Renewal shall thereafter terminate and be null and void on the last day of the fiscal period for which appropriations were made without penalty, liability or expense to County.
- The Agreement remains in full force and effect. If there is a conflict between this 2nd Renewal and the Agreement, the provisions of this 2nd Renewal shall prevail.
 This 2nd Renewal shall be effective upon execution by the County.

Robert Hebert, County Judge

12-19-2017

Date

Authorized Agent - Signature

Brian A. Wilson

Authorized Agent- Printed Name

Vice President, Finance

Title

November 30, 2017

Date

Date

AUDITOR'S CERTIFICATE

I hereby certify that funds in the amount of \$_120,358. are available to pay the obligation of Fort Bend County within the foregoing Agreement.

Robert Ed Sturdivant, County Auditor

MLG E\AGREEMENTS\2018 Agreements\County Clerk\Second Renewal to Manatron Agreement 11/17/2017

Attachment A



LETTER OF AUTHORIZATION

July 18, 2017

Ms. Laura Richard County Clerk Fort Bend County 301 Jackson Richmond, TX 77469

Dear Ms. Richard:

This Letter of Authorization ("LOA") is being provided to you because of Fort Bend County's (the "County's") long-term loyalty and commitment to utilizing the software and services of Manatron, Inc. — a Thomson Reuters Business ("Thomson Reuters Tax & Accounting, Government" or "TRTA Gov"). TRTA Gov values the partnership that exists between our respective organizations.

In accordance with the County's existing Master Agreement dated October 9, 2007 with TRTA Gov, we would like to honor our partnership by offering the County the following discounted maintenance and support services rates for a non-cancellable term of thirty-six (36) months. The County will be billed quarterly for these services.

MAINTENANCE AND SUPPORT SERVICES FOR OCTOBER 1, 2017 - SEPTEMBER 30, 2018

Description	Annual Price	
Aumentum Recorder System	\$38,256.01	
eRecording	\$19,625.58 \$20,500.87	
Public Access		
Vital Records	\$2,850.34	
Automated Indexing System	\$20,047.22	
Automated Redaction System	Included	
NetMinutes Software	\$9,041.42	
AgendaQuick Link Software	Included	
Verity License	Included	
LeadTools Software (55)	Included	
Pix Tools Software (26)	Included	
Marriage License Software	\$3,800.41	
Service Level 4	\$5,700.64	
AgendaQuick Link Software - Archival Module	\$535.60	
Total Annual Price	\$120,358.09	

Manatron, Inc. - A Thomson Reuters Business

LOA No. TX071817FBC

Expiration Date: November 30, 2017

Page 1 of 3



MAINTENANCE AND SUPPORT SERVICES FOR OCTOBER 1, 2018 - SEPTEMBER 30, 2019

Description	Annual Price	
Aumentum Recorder System	\$39,403.69	
eRecording	\$20,214.35 \$21,115.90 \$2,935.85	
Public Access		
Vital Records		
Automated Indexing System	\$20,648.64	
Automated Redaction System	Included	
NetMinutes Software	\$9,312,66	
AgendaQuick Link Software	Included	
Verity License	Included	
LeadTools Software (55)	Included	
Pix Tools Software (26)	Included	
Marriage License Software	\$3,914.42	
Service Level 4	\$5,871.66	
AgendaQuick Link Software – Archival Module	\$535.60	
Total Annual Price	\$123,952.77	

Manatron, Inc. - A Thomson Reuters Business

LOA No. TX071817FBC

Expiration Date: November 30, 2017

Page 2 of 3



MAINTENANCE AND SUPPORT SERVICES FOR OCTOBER 1, 2019 - SEPTEMBER 30, 2020

Description	\$40,585.80 \$20,820.78 \$21,749.38	
Aumentum Recorder System		
eRecording		
Public Access		
Vital Records	\$3,023.93	
Automated Indexing System	\$21,268.10	
Automated Redaction System	Included	
NetMinutes Software	\$9,592.04	
AgendaQuick Link Software	Included	
Verity License	Included	
LeadTools Software (55)	Included	
Pix Tools Software (26)	Included	
Marriage License Software	\$4,031.85	
Service Level 4	\$6,047.81	
AgendaQuick Link Software – Archival Module	\$535.60	
Total Annual Price	\$127,655.29	

Signing of this LOA constitutes acceptance of the terms and conditions herein. Upon approval and signing, please return this LOA to TRTA Gov via one of the following methods:

- Email a scanned image of the signed LOU to mary.ammar@thomsonreuters.com;
- Fax a signed copy to the attention of Mary Ammar at (269) 567-2930; or
- Return a signed copy to Manatron, Inc. A Thomson Reuters Business, Contract Administration, 510 East Milham Avenue, Portage, MI 49002.

Again, TRTA Gov appreciates the County's business and looks forward to continuing our service to you.

ACCEPTANCE			
Fort Bend County Texas	TRTA Gov		
BY: Sufortiskand	BY: 1 A A		
PRINTED NAME TO	PRINTED NAME:		
Tura Richard	Brian A. Wilson		
TITLE: COUNTY CHOK	TITLE: Vice President, Finance		
DATE: 10 /2 /2017	DATE: November 30, 2017		
1 10-10-11			

Manatron, Inc. - A Thomson Reuters Business

LOA No. TX071817FBC

Expiration Date: November 30, 2017

Page 3 of 3



Order 0018967

10/15/2017

MAN14710QTR

Page: 1

Acct# 4308100

FORT BEND COUNTY CLERK
301 JACKSON
RICHMOND TX 77469

OCT-SEPT

Qty	Item Number	Description	Unit Price \$0.00	Ext. Price \$0.00
1,00	MQ	QUARTERLY MAINTENANCE & SUPPORT	\$0.00	\$0.00
1.00	PER	THE PERIOD 10/01/17-12/31/17 FOR	\$0.00	\$0.00
1.00	CON	CONTRACT #TX071817FBC FOR	\$9,564.00	\$9,564.00
1.00	AUM-RECORDER-S	A2 Recorder System Spt	\$4,906.40	\$4,906.40
1.00	AUM-ERECORDING-S	eRecording Support	\$5,125.22	\$5,125.22
1.00	AUM-PUBLIC ACCESS	Public Access Spt	\$712.60	\$712.60
1.00	AUM-VITALS-S	vital Records Spt	\$5,011.80	\$5,011.80
1.00	AI INDEX-S	Automated Indexing System Spt	\$0.00	\$0.00
1.00	AUM-REDACTION-S	A2 Automated Redaction System Spt	\$2,260.36	\$2,260.36
1.00	NETMINUTES-S	NetMinutes SW Spt	\$0.00	\$0.00
1.00	AGENDA LINK-S	AgendaQuick Link SW Spt *INCLUDED*	\$0.00	\$0.00
1.00	VERITY-S	Verity Lic SW Spt *INCLUDED*	\$0.00	\$0.00
1.00	LEAD TOOLS-S	Lead Tools SW Spt (55) *INCLUDED*	\$0.00	\$0.00
1.00	PIXTOOL-S	pix Tools Software Spt (26) *INCLUDED*	\$950.10	\$950.10
1.00	AUM-MARRIAGE-S	Marriage Lic SW Spt	\$1,425.16	\$1,425.16
1.00	SLA-4	Service Level 4 Support	\$0.00	\$0.00
1.00	CON AGENDA LINK-S	CONTRACT #TX072413FBC for AgendaQuick Link SW Spt-Archival Module	\$133.90	\$133.90

Subtotal \$30,089.54 Tax \$0.00

Total \$30,089.54

CERTIFICATE OF INTERESTED PARTIES FORM 1295 1 of 1 OFFICE USE ONLY Complete Nos. 1 - 4 and 6 if there are interested parties. Complete Nos. 1, 2, 3, 5, and 6 if there are no interested parties. **CERTIFICATION OF FILING** Name of business entity filing form, and the city, state and country of the business entity's place Certificate Number: 2017-288865 of business. Manatron, Inc. - A Thomson Reuters Business Date Filed: Portage, MI United States Name of governmental entity or state agency that is a party to the contract for which the form is 11/30/2017 being filed. Date Acknowledged: Fort Bend County Provide the identification number used by the governmental entity or state agency to track or identify the contract, and provide a description of the services, goods, or other property to be provided under the contract, Maintenance and Support Nature of Interest City, State, Country (place of business) (check applicable) Name of Interested Party Controlling Intermediary 5 Check only if there is NO Interested Party. X I swear, or affirm, under penalty of perjury, that the above disclosure is true and correct. 6 AFFIDAVIT Signature of authorized agent of contracting business entity AFFIX NOTARY STAMP / SEAL ABOVE Sworn to and subscribed before me, by the said , to certify which, witness my hand and seal of office. PAMELA DENISE WEST Notary Public, State of Michigan County of Van Buren My Commission Bioles May 16, 2022 India the Chulmy of County of Turn Proce. Was istent Tide of officer administering oath

Forms provided by Texas Ethics Commission

Signature of officer administering oath

www.ethics.state.tx.us

Printed name of officer administering oath

Version V1.0.3337

CERTIFICATE OF INTERESTED PARTIES

FORM 1295

		4000			1 of 1
Complete Nos.	1 - 4 and 6 if there are interested parties. 1, 2, 3, 5, and 6 if there are no interested parties.			OFFICE USE	
Name of business entity filing form, and the city, state and country of the business entity's place of business. Manatron, Inc A Thomson Reuters Business				icate Number: -288865	
Portage, MI U 2 Name of gover		the contract for which the form is	Date 11/30	Filed;)/2017	
being filed. Fort Bend Co	•		12/19	Acknowledged: 9/2017	
Provide the id- description of 17515 Maintenance	entification number used by the governmental of the services, goods, or other property to be pro and Support	entity or state agency to track or identification or identification of the contract.	y the co	entract, and pro	vide a
<u></u>			***************************************		f interest
4	Name of Interested Party	City, State, Country (place of busi	City, State, Country (place of business) (check app		Intermediary
5 Check only if	there is NO Interested Party.			<u></u>	***************************************
6 AFFIDAVIT	I swear	, or affirm, under penalty of perjury, that t	ne above	e disclosure is tru	le and correct,
	-	Signature of authorized agent of c	ontractin	g business entity	·
AFFIX NOTAL	RY STAMP / SEAL ABOVE				
Swom to and 20,	subscribed before me, by the saldto certify which, witness my hand and seal of office.	, this the	··· •	day of	uananan mananan mananan kananan kanana
Signature		e of officer administering oath	Title of	officer administe	ring oath

Forms provided by Texas Ethics Commission

www.ethics.state.tx.us

S

COUNTY OF FORT BEND

S

RENEWAL TO HIPA PLUS HOSTING AGREEMENT

THIS RENEWAL is entered into by and between Fort Bend County, ("County"), a body corporate and politic under the laws of the State of Texas, and Manatron, Inc., a Thomson Reuters Business, ("Manatron"), a company authorized to conduct business in the State of Texas.

WHEREAS, County and Manatron previously executed a Master Agreement, Amendment 1 to the Subscription Agreement, and Amendment 2 the Subscription Agreement (attached hereto as "Exhibit A" and collectively referred to as the "Agreement"). County and Manatron now desire to renew the Agreement, subject to the Statement of Work attached hereto as "Exhibit B".

NOW, THEREFORE, in consideration of the foregoing, the Agreement between the County and Manatron are hereby renewed amended as follows:

- County shall renew the Agreement, subject to the terms set forth in Exhibit B, the Statement
 of Work.
- Payment shall be made by County within thirty (30) days of receipt of invoice. County is a body corporate and politic under the laws of the State of Texas and claims exemption from sales and use taxes. A copy of a tax-exempt certificate will be furnished upon request.
- 3. It is specifically understood and agreed that in the event no funds or insufficient funds are appropriated by County under this Renewal, County shall notify all necessary parties that the Renewal shall thereafter terminate and be null and void on the last day of the fiscal period for which appropriations were made without penalty, liability or expense to County.
- The Agreement remains in full force and effect. If there is a conflict between this Renewal and the Agreement, the provisions of this Renewal shall prevail.
 This Renewal shall be effective upon execution by the County.

Robert Hebert, County Judge

MANATRON, INC.

Authorized Agent – Signature

Brian Wilson

Authorized Agent- Printed Name

Vice President, Finance

Title

May 3, 2016

Date

Date

AUDITOR'S CERTIFICATE

I hereby certify that funds in the amount of \$130,750 are available to pay the obligation of Fort Bend County within the foregoing Agreement.

Robert Ed Sturdivant, County Auditor

Exhibit A



s

HIPA PLUS HOSTING

SUPPLEMENTAL AGREEMENT TO THE MANATRON RECORDS MANAGEMENT MASTER AGREEMENT

This Supplemental Agreement to the Manatron Records Management Master Agreement is by and between Manatron, Inc., a Michigan corporation ("Manatron") and Fort Bend County, Texas ("Fort Bend").

Fort Bend and Manatron entered into a Records Management Master Agreement and a Records Management Annual Software License and Maintenance Supplemental Agreement ("the Agreements") on October 9, 2007, pursuant to which Fort Bend engaged Manatron software and support services to operate the Fort Bend County Clerk's Official Records Imaging System.

In consideration of the covenants and agreements set forth herein, the parties agree as follows:

- 1. Manatron shall provide hosting services as described in Manatron Statement of Work TX030212FBC Version 1.7, "HIPA Plus Hosting," dated June 12, 2012. The Manatron Statement of Work is attached hereto as Exhibit A.
- This Supplemental Agreement is subject to the terms and conditions set forth in the above referenced Agreements. If any conflict exists between this Supplemental Agreement and the Agreements, the terms and conditions of the Agreements will govern.

Agreements will govern.	
Dated: June 24, 2012	SHAMONERS COMME
Manatron, Inc:	
By Sanet Buis Miller Name: Janet Buis-Miller	
Name: Janet Buis-Miller	BENO COUNTINE
Title: Director, Finance and Accounti	ng amminum
County Judge for Fort Bernd Gounty, TX By Ullen Clear	County Clerk for Fort Bend County, TX By: Houldon
See of the second secon	Name: DIANNE WILSON
Name: Robert E. Hebert	E-R-G-17-271 Applications 7 7 5 4 4 place the hard Grant American



AUDITOR'S CERTIFICATE

I hereby certify that funds are available in the amount of \$ \(\begin{align*} \frac{\psi \cong \cong\

Robert Edward Sturdivant, County Auditor

MANATRON

EXHIBIT A

Manatron, Inc.

Statement of Work

HIPA Plus Hosting

Fort Bend County, Texas

June 12, 2012 Version 1.7 Prepared by: John Rickerby

Document submitted by Manatron, Inc.

Manatron Records Management Division

Manatron Corporate Headquarters

1807 Braker Lane Suite 400

510 East Milham Avenue

Austin, TX 78758

Portage, M1 49002

Tel: 866-917-4354

Tel: 259-567-2900

Eax: 512-633-8342

Pax: 269-567-2930

Proprietary Notice:

This document contains confidential and proprietary information belonging exclusively to Manatron. No part of this publication may be reproduced, stored in a retrieval system, or transmitted in any form or by any means electronic, mechanical, photocopied, recorded or otherwise without prior written permission of Manatron, subject to the requirements of the Texas Public Information Act.

© 2012 Manatron, Inc.

PROJECT SUMMARY

Project:

Fort Bend County, Texas HIPA Plus hosting services for transactional replication of

production Anthem/GRM Recorder and Agenda Quick servers, hosting of

Anthem/GRM Recorder and Agenda Quick Internet Public Access with eCommerce,

and eRecording submissions.

Project Site:

Fort Bend County 301 Jackson Street Richmond, Texas 77469

County Contact(s):

Dianne Wilson, County Clerk - 281.341.8686

Target Date of Implementation: Detailed schedule with agreed-upon dates to be during the planning phase of this project. Estimated start date within 60 days of contract signing.

Scope:

Initial installation will include replication of the current Anthem and AgendaQuick software and production versions of Anthem Public Access, AgendaQuick Public Access, and effecteding. The County plans to upgrade from Anthem to GRM Recorder later in 2012. Here within, Anthem/GRM Recorder will refer to the Anthem or GRM Recorder software currently being used by the County.

Project scope will include process and requirements review, hosting software and

hardware configuration, testing, and migration support

Technology:

The County will provide all local production hardware and infrastructure. Manatron will provide all data center replication and web server hardware and

infrastructure.

Software (see following table for software licenses included) -

The software deliverables from Manatron are listed in this Statement of Work. Deliverables outside the current functionality of the standard Anthem/GRM Recorder suite and/or as installed currently in the County's production

environment may be subject to additional development and configuration, which may be accompanied by additional charges. Any changes or additions to these deliverables will need to adhere to Manatron's Change Management Process.

This Statement of Work outlines deliverables, assumptions, and a high-level scope of work. In response to the planning phase of this project, updated project documentation and a detailed schedule will be mutually agreed upon by Manatron, Inc. ("Manatron") and Fort Bond County, Texas (the "County").

PROJECT OVERVIEW

This Statement of Work (SOW) defines the ereas or scope of work for this project and identifies Manatron and County project responsibilities. It also describes the project management processes for performing this work, the criteria for project responsibilities. It also decessfully completed, and the methods to be used to control scope, quality, and costs. Mitigation is defined for any significant project risks that have been identified.

SOFTWARE:

Manatron will set up database-to-database replication of the data and images to the hosted data center. This will pranation with the current Anthem applications and planned 2012 upgrade to the Anthem/GRM Recorder suito of application software and the Microsoft SQL Server 2008 database and replication of AgendaQuick. The County's or appreciation of Agenda Recorder Public Access with a Commerce module, Agenda Quick Internet Public Access and Internet Anthem/CIM Recorder Public Access and eRecording module will also be directly hosted from the data center and will include real-time communication to the local Anthem/GRM Recorder and AgendaQuick production servers located in the County.

em nthem/GRM Recorder pplication Software	Description No changes to current Anthem/GRM Recorder functionality	Number of Users N/A	Comments Replicate to data center
annia Orick	No changes to current Aganda Quiele functionality	N/A	Replicated to data center.
pplication Software old Fusion, Version 10	Required for Agenda Quick	N/A	License required for data center and production system
igndaQuick Public	Public Access for Internet-based researchers	Site license (unlimited users)	Hosted at data center
Access Anthem/GRM Recorder Public Access with	Public Access for Internet-based researchers	Sita licensa (unlimited users)	Hosted at data center
eCommerce Anthem/GRM Recorder eRecording	Receive electronic filing of land records documents	County-approved submitters	Hosted at data center
Microsoft SQL Server 2008 Database Software	Database software, including transactional replication functionality	N/A	Data center license provided by Manatron
SSL Certificato	128 SSL Cartificate for eCommerce	N/A	SSL certificate provided by County

RECOMMENDED HARDWARE:

Manatron will provide all data center server hardware, software, and infrastructure required for replicating the Manation win provide an data with the country's Anthem/GRM Recorder system, Agenda Quick, and hosting internet Public Access and effecting. The County will be responsible for local production server hardware, software, infrastructure for Anthem/GRM Recorder county will be responsible for providing any new peripheral hardware. The County will be responsible for providing any recommended changes to the network and bandwidth.

200		Description	Quantity	Comments
900	Data Censes con and		1	Manetron will provide.
		AgendaQuicks erver and production web server for internet Public Access and oracording-		
			Andrew Control of the	

1	- 1:
	- 1
F	F
,	
7	
and the later of	4446
	فميشدن

IMPLEMENTATION SERVICES:

Per this SOW, the following services will be provided for this project.

Description Project Management Services	Comments Manages overall project schedule, directs all Manatron resources, and acts as escalation point for project.
Hardware Installation	Manatron will provide all hardware and infrastructure at the Time Warner indianapolis data center. The County will be responsible for local hardware and infrastructure.
Software Configuration, Localization, and Quality	Incudes configuration and testing of transactional replication to the data center and hosting internet Public Access and eRecording.
Assurance Database Configuration and Quality Assurance	Includes set-up and testing of index and image databases.
Training	No user training required. County IS will receive instruction on system management.
On-Site Support	No on-site support required. Go-Live will be initiated remotely following testing.
Documentation	Standard Manatron soft documentation for County IS.

PROJECT IMPLEMENTATION

PURPOSE STATEMENT

The purpose of this project is to provide full off-site transactional replication of the County's Anthem/GRM Recorder and AgendaQuick software, indexes, and images at Manatron's data conter located in the Time Warner facility in Indianapolis, indiana. The data center will provide layered redundancy to the County's local production system and fallover in the event of an interruption in access to the production system or a full disaster.

in addition to Anthem/GRM Recorder replication, the data center will host County's internet Public Assess and eRecording modules to manage Internet researcher inquiries of County Indexes and images stored in Anthem/GRM executions involves to the receiving of execording submissions for transmission to the County production systems. This configuration will ensure high availability for these internet-based functions within consistently managed response conneguration was a single the local production configuration and ensuring consistent access even in the event of a disaster at the County's site.

This Statement of Work describes the project that Manatron is responsible for implementing. Upon contract signing, a subsequent project execution pian will be created jointly by the Manatron and County project managers. The project execution plan will detail how the project will be managed, including detailed communication, risk, and scheduling plans.

The project will be administered in the following phases:

- Project planning—commences upon contract signing (or earlier);
- Data center system set-up, configuration, and connection to the County:
- Configuration and testing of software, index data, and images;
- Go-Live of Anthem/GRM and Agenda Quick replication, and switchover to hosted Internet Public Access and e Recording.

SCOPE STATEMENT

This project is responsible for the Manatron Hosted Internet Public Access service solution. These services include the Installation, configuration, and set-up for hosting from the Manatron data center. Specifically, this project entails: Configuration of hosted server hardware at the Manatron data center;

- Set-up and testing of real-time database and image replication to the data center;
- Go-Live support within the terms of the agreement.

ASSUMPTIONS AND CONSTRAINTS

- 1. A 36-month commitment is required for service for Manatron applications. If the County decides to terminate the service at any point prior to the end of the 36-month commitment, the County will be responsible for the fees for the Manatron applications for the balance of the commitment.
- 2. Manatron will replicate the existing indexes, images, Anthem/GRM Recorder software, and Microsoft SQL Server database to the remote data center configuration. The County's current Anthem/GRM Recorder release level and configuration will not be changed.
- 3. Manatron will replicate the existing AgendaQuick data, images, and software to the remote data center and configure for disaster recovery. Public users will access AgendaQuick documents via the remote data conter just like Anthem/GRM public users. The County's current Agenda Quick release level and configuration will not be changed except for the production and data center Cold Fusion licenses, which will be apgraded to Version 10.
- Public Access for Anthem/GRM Recorder and AgendaQuick will allow researchers to view documents using the HIPA Plus replication server. Requests to purchase Anthem/GRM Recorder document copies will be redirected to the County/production server to complete the transaction. For eRecording, documents will be received from submitters in the hosted application and will be pushed to the production server for normal processing as batches are accessed by users. All database changes occur on the production server. Replication is from the production server to the hosted server. In the event of a disaster event impacting the County production server, the hosted replication server can be enabled to allow ecommerce Public Access purchases, eRecording processing, and any other standard County Clerk business transaction. When the production server is back online, Manatron will coordinate through County IT to update the production server.
- 5. The County will be provided with 2TB of data storage, which should be sufficient based on current estimates. If usage exceeds this estimate and additional storage has to be added, the County will be notified and asked to approve the cost of adding additional storage for the balance of the contract. Space will be monitored as needed to ensure there is no limitation in functionality.
- 6. The term "real-time replication" can take a few minutes to complete as network/server traffic will affect the timing of the replication between sites.
- 7. Data replication can be impacted by the telecommunication provided between the County and the data center.
- The County is responsible for providing broadband communication access and any necessary firewall from the production site to an internet service provider in the case of a LAN-to-LAN VPN tunnel.
- Manatron will provide the County with a one-time delivery of images and database stored in the hosted facility at no charge to the County. This will include both database and images for the Manatron AgendaQuickapplications. If the County requests that Manatron Technical Services restore the delivered backup data and images on the

- local server, then a separate billable work order would be created based on the time and materials required to
- 10. Data and images housed in the Anthem/GRM Recorder system are of a non-proprietary nature. All data will be Data and things a standard SQL database structure, and images will continue to be stored as Group 4 tiff. All data and images hosted in the data center are the property of the Fort Bend County Clerk.
- 11. The project schedule is subject to resource availability (both Manatron and County). Once contract acceptance has occurred, a project execution plan with a project schedule will be provided to the County for acceptance.
- 12. The County will be responsible for providing all Anthem/GRM Recorder production system hardware, the country with the environmental space, power requirements, and access by Manatron for installation testing to infrastructure and environmental space, power requirements, and access by Manatron for installation testing to the data center and operation.
- 13. The County will provide the SSL Certificate for use with eCommerce and eRecording.
- 14. High-speed remote access to application and database servar(s) must be provided to the Manatron project team rugar-specture required access members at agreed upon times for set-up, configuration and testing. The date and times of the required access members at any control and the County in advance by the respective project managers.
- 15. The County will be responsible for testing and reviewing the data on the remote server and providing approval for cut-over production.
- 16. All documentation provided by Manatron is provided "as-is."
- 17. Manatron will provide training for County IS personnel as it relates to the configuration of the local SQL server database for transactional replication and troubleshooting. This will allow for a working knowledge of the solution and any interdependencies.
- 18. Manatron recommends running 100mbs Ethernet connections to the desktop.
- 19. Future technology refreshes for the County have not been included in the scape of this project. Technology ruture account to a peripheral hardware upgrades or replacement of equipment

DEFINITION OF SEVERITY LEVELS

Manatron shall respond to any errors reported by the County based on the severity level assigned to such error. Severity Levels and Manatron's responses are detailed in the Records Management Master Agreement dated October 9, 2007 between the County and Manatron.

PROJECT STAKEHOLDERS

Same .	Rolla Courty Spansor	Contact Information 281:341.8686	Accepts deliverables; approves
Dianne Wilson, County Clerk	County aparent	dianne wilson @co.fort: bend.tx.us	changes
Wade Peikert	Manatron Sponsor	866.917.4354 x7181 wado.peikert@manatron.com	Point of escalation; approves change.
Clay Elliott IT	County Project Manager	281.341.4588 clay.elliott@co.fort-bend.tx.us	Monitors schedule and deliverables; coordinates County responsibilities,
Operations Manager TBD	Manatron Project Manager		Monitors schedule and deliverables; coordinates Manatron responsibilities.
TRD	Manatron Product Manager	ndo redu art date color company and an extra color design of the c	Provides guidance to professional services staff.
TBD	Mariatron Consultant(s)		Business analysis, training, and on- site support.
LBD	Manatron System Engineer(s)	the market before the control of the	Consultation and/or configuration of hosted servers, and County pertpherals.
TBD	Manatron Development Lead	and the second of the second o	Testing and installation of software conversion.

IMPLEMENTATION DELIVERABLES

Deliverables refer to what a project is to produce. With this HIPA Plus implementation project, the deliverables fall penyerance categories: hardware deliverables, software deliverables, remote data center deliverables, training nno acratal cata, image conversion deliverables, documentation deliverables, and project management deliverables.

HARDWARE AND SOFTWARE DELIVERABLES

Hardware and software deliverables are listed at the beginning of this document. The County will confirm with Managron all peripheral hardware in the existing configuration that will need to be utilized in the new configuration. Manatron will confirm with the County that any existing devices to be used in the new configuration meet minimum specifications and will function properly.

REMOTE DATA CENTER DELIVERABLES

Remote Data Center deliverables are described in "Attachment 3 - Managed Services Hosting Agreement" located at the end of this document.

All remote hardware and software and data center infrastructure are provided within the proposed HIPA Plushosting services described in this Statement of Work. The County will be provided with 2 TB of data storage which should be services described on current estimates. Manatron will be responsible for system set-up, configuration, and establishing communication with the County's network.

Fort Bend County TX030212FBC - V 1.7

PROJECT MANAGEMENT DELIVERABLES

A detailed project plan and schedule will specify how the project will be managed. It will include detailed communication plans, roles and responsibilities, risk mitigation plans, and the change management process, including information on

- Meeting minutes and status reports;
- Change requests;
- Risk management;
- Issue tracking.

COUNTY RESPONSIBILITIES

The County shall cooperate with Manatron and promptly perform the County's responsibilities to assist Manatron in its installation of the Manatron HIPA PLUS solution, including, but not limited to, those responsibilities set forth he low.

SITE PREPARATION AND MAINTENANCE

The County shall:

- Prepare the local production site to facilitate communication between the data center and the Manatron Anthem/GRM Recorder and Agenda Quicksolutions in accordance with instructions provided by Manatron;
- Determine that the HIPA Plus hosting system meets requirements; and
- Provide access to the local production system as needed.

The County is responsible for providing and maintaining an environment which meets manufacturer recommendations for computer and network systems, all wiring required for hardware, and the protection of wiring from any damage. This includes environments for testing, training, and production support.

COUNTY PROJECT MANAGER

The County shall appoint a project manager with sufficient time and technical expertise to ensure the timely and country such approximation and with sufficient authority to sign the acceptance criteria. Any delays caused by delays in appointing a project manager will extend Manatron's delivery and installation deadlines. The project manager will in appointment of the County and make decisions regarding the day-to-day operations, provide information and materials to Manatron, provide access to the County's staff to answer questions, coordinate the County activities and responsibilities for the project, and communicate with Manatron concerning the County's performance. The County may change their project manager upon prior written notice to Manatron.

NON-MANATRON HARDWARE AND NON-MANATRON SOFTWARE ACQUISITION

The County shall acquire all necessary hardware, third-party software for the server (Le., workstations, printers, scanners, firewall appliance, SSL certificates, etc.), and any other necessary third-party software not specified in this SUBMINION A hardware manufacturer, vendor, or other supplier within the time frames required by the project timeline.

ON-SITE OFFICE VISITS

Access to the County's facilities must be available to the Manatron project team members for the duration of any onsite visits. The date and times of the required access will be scheduled between Manatron and County project managers in advance of the visits. The County shall make office space available for Manatron personnel while on-site (a minimum of one (1) desk, one (1) telephone, and electrical outlets), and provide an analog phone line for remote connection.

The County will provide a secure connection for remote access to facilitate Manatron's support of the system. This can be accomplished via VPN connection for Manatron personnel on a 7-day/24-hour basis during the installation or as otherwise arranged by both parties.

The County shall provide a Local Area Network (LAN) with Internet Protocol (IP) switched 100Mb Ethernet with wiring to the Network Interface Card (NIC).

NETWORK ADMINISTRATION AND OPERATING SYSTEM SECURITY

The County shall provide trained personnel to administer its network and manage its domain, including security access.

PROJECT CONTROL PROCESSES AND PROJECT MANAGEMENT **PROCEDURES**

COMMUNICATIONS PLAN

In order to keep the County and Manatron project managers and the project team informed on the progress of the in order to acep the communications plan will be created for the project execution plan. The communications plan specifies:

- Meeting schedule;
- What information the weekly status reports will contain; and
- Distribution.

Project Status Meetings

The County and Manatron project managers will meet regularly to update the project's progress, discuss and approve deliverables, resolve issues, discuss and approve change requests, determine appropriate management actions, and ensure the success of the project. Project team members will also meet as required.

LOCATION OF PROJECT DOCUMENTS, DELIVERABLES AND FILES

Soft copies of project documents, deliverables, status reports, meeting notes, etc. will be kept by the Manatron project manager electronically and are available upon request.

ISSUE-TRACKING AND RESOLUTION PROCEDURES

In the course of the project, issues will arise that will require documentation and resolution. An issue is defined as a in the course of the project from progressing or is inconsistent with the requirements of the

ISSUE-TRACKING STRATEGY

Issues may be identified by any project team member and escalated to the County and Manatron project managers for review.

- All issues will be logged, tracked and maintained by the Manatron project manager;
- The lastes Log (using Manatron's tracking system) will be made available to each stakeholder upon request;
- Project spousors will be notified of any issues with potential risk to project scope, schedule, or cost.

Fort Bend County TX030212FBC -- V 1.7

ISSUE RESPONSE AND RESOLUTION

;

The Manatron and County project managers will determine activities necessary for responding to issues. They will assign these activities to appropriate team members and are responsible for monitoring whether or not shese assign these are being completed, and whether or not they are effective in reducing the impact of an issue. They will activities are being completed, and whether or not they are effective in reducing the impact of an issue. report progress at team meetings, including the effectiveness of previous resolution plans.

CHANGE MANAGEMENT PROCESS

The Change Management Process is put in place to control scope. If processes are not set to handle change in a structured manner, projects will fail to meet expectations and goals, such as budgets, estimates and schedules.

A Change of Scope is defined as a change to any of the following:

- Hardware configuration affecting the performance or capacity of the system;
- Third-party software configuration affecting the performance or capacity of the system;
- A change in the software or hardware configuration;
- A change in the form or functionality of the Manatron application software that deviates from the mutually agreed upon final software requirements; or
- Any other change that could affect the project schedule or budget.

Changes to the project, such as delays, changes in scope, change in estimates, etc., will be documented in Manatron's Change Management System. The County or Manatron can initiate these Change Requests. The party shall identify the nature of the proposed change and reasons for the proposed change.

Manatron shall evaluate the effect of the change set forth in the Change Request with respect to the feasibility, manation such training, acceptance criteria and implementation date of the project. The results of Manatron's evaluation shall be added to and become part of the Change Request, If Manatron's evaluation of the request is positive, Manatron will propose a specific implementation and specify any additional time and charges necessary for positive, manager than properties and the scope change. If Manatron's evaluation is negative, Manatron will provide their rationale for not recommending the change.

Manatron will work jointly with the County to determine mutual interest in pursuing the Change Request. The County manager or reject the proposed solution. Should mutual agreement be reached, Manatron shall submit feedback to the County, including impact to timing and price of implementation and maintenance,

RISK CONTINGENCY OUTLINE

A Risk Contingency Outline defines potential risks associated with a project. Its purpose is to provide suggested tesolutions ahead of time for the situations that may affect the project timeline and/or budget. By being proactive in identifying possible risk issues and solutions, the impact to the project can be minimized if one of these issues occurs,

OUTLINED RISKS

The project plan will provide a description of the potential risks ("Risk Register") associated with this project, as well as suggested mitigation. A Risk Register is developed as a result of the findings completed in the planning phase. The as suggested integration and a supplied that the county project manager to detail specific known risks as a deliverable during the analysis phase. Risks may include those pertaining to environment, resources, and performance, among others. Execution of the suggested mitigation would require approval by the County project manager and the Manatron project manager and may increase the price of the project and/or require added time.

PRICE

The following outlines the HIPA Plus hosting services proposed.

The Machine Services	Extended Price
MAHAUTON HIPA Plus Hosting Services HIPA Plus Hosting Services HIPA Plus transactional data replication of Anthem/GRM Recorder and AgendaQuick production server to the Indianapolis data center HIPA Plus Hosted Internet Public Access with eCommerce and eRecording County to provide SSL certificate for eCommerce and eRecording 2TB of storage Includes all Anthem/GRM Recorder and AgendaQuick server and web server data storage hardware located at the data center, Microsoft SQL Server database software, and all supporting infrastructure to ensure high availability. Includes Manatron Project Management, Technical Services and Database Services. One-time set-up fee Recurring Manatron applications monthly fee for 36 months (\$3,632.00 per month, or \$43,584.00 per year)	\$ 10,079.00 \$130,752.00
Total Fees for HIPA Plus Hosting Services	\$140,831.00

ATTACHMENT 1- AGREEMENT TO SOW

Statement of Work Agreed and Accepted:

We, the undersigned, accept this document as a stable work product to be used in the delivery of the project described we, the under the Statement of Work is to be handled through Manatron's Change Management Process.

This Statement of Work will confirm all requests for software and/or services as outlined and at the prices indicated. This will be an addendum to the Records Management Master Agreement dated October 9, 2007 between the County and Manatron. All terms and conditions of that agreement will pertain.

Billing for HIPA Plus:

The set-up service fees (\$10,079.00) are due upon completion of system set-up.

Monthly billing of \$3,632.00 for HIPA Plus will begin upon the signing of Acceptance Form 1.

The total amount of one-time fees that will be billed to the County for set-up services is \$10,079.00. This rotal includes travel expenses. HIPA Plus hosting fees are \$3,632.00 per month for 36 months. The total fees for the full 36 months are \$140,831.00.

Fort Bend County, TX:

Manatron:

Fort Bend County 301 Jacksen Street Richmond, TX 77469	Manatron, Inc. 1807 Braker Lane, Suite 400 Austin, TX 78758
BY: (Kahout Lebert	BY: Spart Beni Miller
PRINTED NAME: Robert E. Hebert	PRINTEDNAME: Janet Buis-Müler TITLE:
TITLE:	Director of Finance and Accounting
County Judge DATE: June 26, 2012	DATE: June 25, 2012

This agreement is not effective until executed by all parties.

ATTACHMENT 2- ACCEPTANCE FORMS

ACCEPTANCE FORM 1 - HIPA PLUS COMPLETION OF SET-UP

ruxpose.
The purpose of this acceptance form is for the County to sign off on that the remote data center is set up, to agree that Purpose: the purpose of any for testing with the County, to agree that Manatron should proceed to the next phase through the life cycle of this project, and to accept billing for this phase.

outputs:

- 1. Manatron has conducted a project kick-off meeting with the County.
- The project schedule has been delivered for the County to review.
- The County server and infrastructure configuration at the HIPA Plus remote data center has been set up, and the index and image data is available for replication testing.
- the inues and the Microsoft SQL server database have been 4. The Anthem/GRM Recorder and AgendaQuick applications and the Microsoft SQL server database have been loaded and configured at the data center.

rhaco services \	vere complete	d on	
1 1130525 3431 4 4444 4	, , , , , , , , , , , , , , , , , , ,		

We, the undersigned, agree that this work is complete and that under the conditions of this Statement of Work and the Records Management Master Agreement dated October 9, 2007, the County will be billed 100% of the HIPA Plus the nectors remarked (\$10.070.00). Monthly billing of \$3.632.00 for HIPA Plus recurring fees will begin upon set-up services price (\$10.070.00). signing this acceptance. It is agreed to proceed to the next phase of this project.

Fort Bend County, TX:

Manatron:

Manatron, Inc. 1807 Braker Lane, Suite 400 Austin, TX 78758
BY:
PRINTED NAME:
TYTLE:
DATE:

ACCEPTANCE FORM 2 - GO-LIVE

The purpose of this acceptance form is for the County to sign off on the Go-Live phase and to agree that the HIPA Plus service is ready for production use.

Outputs:

1. Manatron has completed the set-up and testing of the replication of indexes and images with the Anthem/GRM Recorder and Agenda Quick production system.

Anthemy Store Access according to the 2. Manatron has configured and tested Anthem/GRM and Agenda Quick Internet Public Access according to the County configuration.

3. The County has viewed and tested each of the system components and accepts the system.

4. The system is functioning as described in this Statement of Work with no Severity Level ONE issues outstanding. Any other outstanding issues have been documented for follow up with the support teams.

5. The County authorizes the system to move into production mode.

These items were	completed on	White the second
------------------	--------------	--

We, the undersigned, agree that this work is complete under the conditions of this Statement of Work and the Records we, the university of the country agrees that this project is complete.

Fort Bend County, TX:

Manatron:

Fort Bend County 301 Jackson Street Richmond, TX 77469	Manatron, Inc. 1807 Braker Lane, Suite 400 Austin, TX 78758
DY;	BY:
PRINTED NAME:	PRINTED NAME:
TITLE:	TITLE
DATE:	DATE:

ATTACHMENT 3 - MANAGED SERVICES HOSTING AGREEMENT

An incident is defined as any time the system is inaccessible to the public. The County will own the telecommunication (VPN) line that provides the data to the data center and will be first-line support; however, there may be an issue that is out of Manatron's control that will temporarily affect the database replication. Manatron will initiate a support call, monitor progress, and monitor the relecommunication line. Database replication is not included in the credit percentage section of this SOW. The telecommunication within the data center that provides availability to the records/website once on the server is included in the credit percentages and is owned by the data center.

The SOW applies only when the County is agreeing to a term commitment of thirty-six (36) months for the Manatron applications. If the County decides to terminate the service, the County will be responsible for data center charges through the Manatron application's thirty-six-month commitment. Manatron reserves the right to amend the SOW from time to time effective upon notice to County, provided that in the event of any amendment resulting in a material reduction of the service levels, the change request sets forth County's sole remedies for any claim relating to the services, including any failure to meet any guarantee set forth in the change request. The Data Center records and data shall be the basis for all change request calculations and determinations. Notwithstanding anything to the contrary, the maximum amount of credit in a month shall not exceed 10% of the monthly recurring fee.

Without limiting the foregoing, County agrees that neither it nor its employees, agents, contractors, or representatives shall attempt in any way to circumvent or otherwise interfere with any security precautions, procedural controls, Acceptable Use Policy, Change Management, or other Data Center policies relating to the Data Center service offering. Any such actions may cause a disruption in service. Any disruption of services which results in the violation of these provisions shall be excluded from the unsuppose of a suppose of the County will have no right to any service credit or other remedy under a infrastructure availability, and the County will have no right to any service credit or other remedy under a change request or otherwise with respect to such disruption. The County will pay Manatron at the current published rates for reasonable remedial services resulting from the County's actions.

Manatron will provide the County with written notice before performing scheduled maintenance and as much notice as possible when performing emergency change controls. The Data Center designs its services around an N+ 1-redundancy philosophy that is used to minimize outages during planned change controls. The Data Center will use commercially reasonable efforts to minimize the impact of any change control and, when possible, schedule the change control to eliminate or minimize impact to the County's service. However, the Data Center reserves the right to proceed with any change control if it is determined by the nowever, in its sole discretion, that a change control is necessary to maintain the overall integrity of the services and/or the change control will not cause harm to the County.

Infrastructure Availability:

Infrastructure shall consist of the following components that are provided as part of the Data Center services:

- Co-location space reserved for use by the County in the Data Center, including quarter, half, or full cabinets and cage space;
- Power provided to County's co-location space; and
- Internet access provided to the County from the Data Center Internet Backbone.

In the event that Manatron fails to meet its stated Infrastructure Availability in any given month during the service term, County shall be entitled to receive a credit (as the sole and exclusive remedy hereunder) by providing Manatron with a written request for a service credit within sixty (60) days after receipt of an invoice for the period in which the infrastructure was unavailable. The amount of the service credit shall be determined by the percentage in accordance with the table shown below; the credit percentage will be off the monthly recurring fee.

Incident Management Availability Matrix

Infrastructure Availability Percentage	Credit Percentage
100%	0%
<99,99% > 99.5%	196
<99.5% > 99.0%	2%
<99,0% > 98.5%	3.5%
~98.5% > 95.0%	5%
<95,096	1096



Maintenance and Support Schedule and/or Note Schedule

For Internal Use Only Dolasco 10/15/2015 MANI47100TR

ACCES

4306100

NOTE DEND COUNTY CLERK ATTN: PURCHASING DEPT 301 JACKSON

RICHMOND TX

77469

THIS IS NOT AN INVOICE!

oct-sept

	·			
- ~^	3303	QUARTERLY MAINTENANCE & SUPPORT	\$0.00	\$0.00
1.00	NO	THE PERIOD 10/01/15-12/91/15 FOR	\$0.00	\$0.00
1.00	PSR	CONTRACT #TX110311FBC FOR	\$0.00	\$0.00
1.00	CON AUM-RECORDER-S	AZ Recorder System Spt	\$8,928.31	\$8,928.31
1.00	ALM-ERECORDING-S	eRecording Support	\$4,580.28	\$4,580.28
1.00	THE RESERVE TO SECURE OF A	Public Access Spt	\$4,784.56	94,784:56
1.00	VOM-BORDE WOODS	Vital Records Spt	\$665.22	\$665.22
1.00	aum-vitals-5	Automated Indexing System Spt	\$4,678.68	\$4,678.68
1.00	ai index-s	A2 Automated Redaction System Spt	\$0.00	\$0.00
1.00	AUM-REDACTION-S	Nothinutes SN Spt	\$2,110.12	\$2,110,12
1,00	netminutes-5	AgendaQuick Link EW Spt *INCLUDED*	\$0.00	\$0.00
1.00	AGENDA LINK-S	verity bic sw spt *INCLODED*	\$0.00	\$0.00
1.00	VERITY-S	ViewOne Sys Spt (2)	\$243.00	\$243.00
1.00	JAVA VIEWER-S	The state of the s	\$137.75	\$137.75
1.00	PRINT ACCELERATOR-S	Frint Accelerator Spt (2)	\$0.00	\$0.00
1.00	LEAD TOOLS-S	Lead Tools SW Spt (55) *INCLUDED*	\$0.00	\$0.00
1.00	PIXTOOL-S	Pix Tools Software Spt (26) *INCLUDED*	\$896.95	
1.00		Marriage Lic SW Spt	· · ·	\$886.95
1.00		Service Level 4 Support	\$1,330.43	\$1,330.43
1.00		CONTRACT #TX072413FBC for	\$0.00	\$0.00
1.00	AGENDA LINK-S	AgendaQuick Link SW Spt-Archival Module	\$133.90	\$133.90

QTV X 4 Oths = 113916.80 Ahould be

amount

per a fagging

subtount

subtount

for y 928, 479, 20

Tox

south on y 100

subtount

south on y 100

south on y 1

· particular services	. 0
ACORE	

17515

	general gr	SOTIF	:IC/	ATE OF LIA	BILIT	Y INS	IURAN	ICE	63/23/	19005 2015
CORD			,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	W interchase Tople Mill V	ZAND CO	NEERS NO	RIGHTS UP	ON THE MELLING	TE HOL	DER THIS
The second residual to	- 12 12 12 1 2 1 2 1 1 1 1 1 1 1 1 1 1 1		# B L C ** No.	PROPER NOT COMO	2.64 Kr. Strate.					
THE THEFT IS	W. C. S.		M 1 C T 1 L	ARTIPALATE ENGLISHED					*****	(n) his ili n han alm conservation via de de de
PRESENTATIV	ne sertificato i	holder is a	n ADD	INIONAL NEURED, the plicles may require an a	ndorseme	nt Astate	nont on this	cartificate does not	confer t	ghts to the
portanti	rditions of the	policy, cer	MAIN P	Olderico straf terdénte ser a				enjectus (hanna promocety moderne kondunent proper desides desides designa	Antonio de montos feren	genggagge, yfulldig e blikssens unis euskelsbeteilsbeilsbeils
e leans and cor othicate helder	in House of Stach	ATTACK SHIP	H1144 311-	the second secon	NAME:		more thanks the great by the barrows man			Meridian (Alaba parita parita prima Landra (Alaba)
DUCER					PHONE 65C No. E E-MAIL 600REDS:	· W	collect a tree of our programmer characterist	IAG, No	j:	
44 KK ANDUNIN 14 I	the Americas				E-MAIL ADORESS:				in and object to the second	and the state of t
Fax (212) 3453	G23				and an annual of the state of t	孙德 (1)	HERIS) AFFORD	mo coverage	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	<u> </u>
					IMPOURER!	ACE America	n Insurance Con	steers Commercial management	***************************************	20099
1671-AL-CAWL-15-16			HISUNER !	LACE Propers	à Casudiy kuu	It is a resident of the section of t		***************************************		
ured Manaron, Inc.,					MOUNER !	<u> </u>	Methodosopous since conservations	White the section of the second of the second section of the second of t	under ett die 130 stere	AND THE PERSON NAMED IN COLUMN
9 Thouseon Pay.	ters Birthess				INSURER!		maco di tojirgo i e e i i i majojo, e i pane i Modello i	Medica II bolis and proposition described and construction against a second	ijiniyacaate a joani a assidh	enterior contration to the contration of
530 taka mana Pashaga Mil 46	002				INSURER	***************************************	· · · · · · · · · · · · · · · · · · ·	a encourage de la fact	onid-description of the control of t	a gagga gyanyemmandatto-eleksandeksindiktorindiroksi.
					HISURER	f ·		EVISION NUMBER:	6	a griconstruction to the contract of the production of the contract of the con
and the second second	AND THE PROPERTY OF THE PERSON NAMED IN COLUMN TWO IS NOT THE PERSON NAMED IN COLUMN TO THE PERS	CERT	FICAT	ENUMBER:	UNITED SHOW	ISSUED TO	THE INSURE	NAMED ABOVE FOR	ne Po	LICY PERIOD
OVERAGES THIS IS TO CERT HUBICATED NO	NEY THAT THE	POLICIES O	JE INSU JUIREMI BRTAIN	E NUMBER: FRANCE LISTED BELOW H ENT, TERM OR CONDITION THE INSURANCE AFFOR BURITS SHOWN MAY HAV	N OF ANY COED BY T	CONTRACT HE POLICIES	OR OTHER DESCRIBED	OCUMENT WITH RESI HEREIN IS SUBJECT	PECT TO TO ALL	WHICH THIS THE TERMS,
CERTIFICATE M	O CONDITIONS IA BE ISSUED.	as a control of		3. Patelit a me wers 1	AE BEEN KI	POLICY SEF.	POLICY EXP.	is a second construction of the second construct	MITS.	Commence and Arthur Larry 19 (Arthur 1952)
EXCITIZIONES VI	e of Heurance	Tr.	STEEN'S	MIXO G27/39/400	· · · · · · · · · · · · · · · · · · ·	0/31/2015	03/31/2016	EACH OCCURRENCE	5	1,000,00
The second second second second	LITY			LINIO ACLAS MAN	ľ	- 4 " " "		DAMAGE TO REATED PREMISES IE & SUITENEL	ls.	500,00
onovery	LEAGUE LANGUAGE	עונוון		ļ.				MED EXP (Any time person)	3	10,D
Commen	ME NAME X OF	CCUR	1		1			PERSONAL LAGV BERDV	<u> </u>	1,000,0
			- [<u> </u>				GENERAL AGGREGATE	manacas program comm	2,000,0
]		marjula comunidatesidoskususia.]			PRODUCTS - COMPACE AL		2,000,0
LOCKY ACCORD	CATE LIMIT APPLIE	sper [ĺ						\$	opines (il large engineer) Mennigrania antiqueeripho et li interest
X Lecture	1986	roc	weens	ISA 148853083		03/31/2016	03/31/2016	COMBINED SINGLE LAW		0,000,0
AUTOMOBILE	LIABILITY	1			- 1	:		DODKY MURY (Per pane)	· †	
TUA YUG X	_	EQUAEO	. 1					BOOKY BUILTY (Per accord	to too real pages strates o	down the second
ALL CWA		OS LOWNED	.	appropriet.				PROPERTY DAMAGE LEW 1962-1953	\$	Phononecon delamata Nobel Selbert (qui (***), (**), (**),
MARDA	WOO LINE	OS S				with the same and	January and Sangar Propries Contraction Co	Control of the Contro	5	5000.0
		····	-	XCO 027638223	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	(C)31/2015	03/31/2018	FACH OCCUPACE CE		5,0X0,0
B X UNBILES	- interest	OCCUP						A06950315	***********	CARTALS Best colored to separate particles in the security of
EXCEDS	Contactor accounts in the contactor accounts to	CIVING WINE						TISS STANKS I	ny s	ijijajimetterojomiogreenosopeoomposidomjairi(q),
Oro .	RETENTIONS	Provident principal in special days with the second	-	VALICAB 146120 (AOS)		03/31/2015	03/31/2016	- State of particular contract of the state	##	1000
PROPERTY OF THE PARTY OF THE PA	OMPENSATION VERN LIABILITY	YIN	1 1	RSC C48146132 (WI)		03/31/2015	Data legita	IL FACHACCIDENT		ray and anima property of the factors.
A WORKERSE			\$4.1 M	1		ł	1	HE WHICHISH - BA EMPLE		
A WORKERS C A AND EMPLO A MY PROPRI	ETORUPARTHER EXE	ECOUAL N	1	1		l .	1	the same of the sa	3	1.000
A AND EMPLU A ANY PROPRI OFFICEROAS	MACR EXCLUDED?	Ernitae M			######################################	.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		E.L. DISEASE - POLICY L	3	1,000/
A ANY PROPRI OFFICEROAS	YERD LIABRITY STORPAPTHEREXE MBER EXCLUDED? A HIM A CADD! H OF CHERATIONS	EPPHARE N		anno 1800 de 1	######################################			S.L. DISEASE - POLICY L	3	1,0,00
A AND EMPLU A ANY PROPRI OFFICEROAS	MACR EXCLUDED?	Epploy N		A 1940 - 1940	eles de la companie d			el disease - Policy L	3	
AND EMPLOY OFFICERAL	MAER EXCLUDED? IN NHI IN OF CHERATIONS	DDIM					1 and the state of	al disease Policy L	3	
AND EMPLO ANY PROPRI COPICERANG GRANDARY Fees do sond CESCTRICTY	MAER EXCLUDED? H MI) X LANGE H OF COTERATIONS	D0/0W		tach ACORD 101, Additional Rem	nanka belasak	a, if more space	is required)	B.L. BUSTASE - POLICY L	3	
AHO EMPLO A PALY PROPER CYPICEPANE HERMANY I I WEST MONTH COSTORITIES	ASERSATIONS 1 LOS	AUTONE LAZHIL	THE PART OF	tisch ACORD (&), Additional Rem pol workers' compensation where t	nechined by seri	a. If more space	is required)	B.L. BUSTASE - POLICY L	3	
AHO EMPLO A PALY PROPER CYPICEPANE HERMANY I I WE HOME COSTANTINE	ASERSATIONS 1 LOS	AUTONE LAZHIL	THE PART OF	tisch ACORD for, Additional Rem pit workers' compensation) where r red with respect to General Usbilly	equired by sei	iten contract	is required)	B.L. BUSSASE - POLICY L	3	
AND EMPLO A PROPER CYPICEPING REMOTE IN THE PART OF THE PART OF T	ASERSATIONS 1 LOS	AUTONE LAZHIL	THE PART OF	tach ACORD 60°C, Additional Rem apt workers' compensation) where r rad with respect to General Diability	nacks detaction regulard by bei	a. If more space iten contact	ls required)	B.L. BUSEASE - POLICY L	3	
AHO EMPLO A PALY PROPER CYPICEPANE HERMANY I I WE HOME COSTANTINE	ASERSATIONS 1 LOS	AUTONE LAZHIL	THE PART OF	usch ACORD 160. Additional Rem apt workers' compensation) where o rod with respect to General Usability	narks debuchi required by wil	a. If more space lien contract.	je tednytod)	B. DESASE - POLICY L	3	
AHO EMPLO A PHY PROPRI OFFICERATE REMEMBERS I WAS ASSAULTED.	ASERSATIONS 1 LOS	AUTONE LAZHIL	THE PART OF	tech ACORD 160. Additional Rem apt workers' compensation) where r and with respect to General Uability	narks destaction required by seri	a. If more space lien contract.	je tednytod)	B. MESASE - POLICY L	3	
AHO EMPLO A PALY PROPER CYPICEPANE HERMANY I I WE HOME COSTANTINE	ASERSATIONS 1 LOS	AUTONE LAZHIL	THE PART OF	tech ACORD (10), Additional Rem polyworkers' compensation) where r rod with respect to General Liability	i.			B. MESASE - POLICY L	3	
AND EMPLOY AND PROCESS OFFICEROAL GRANDERS OFFICEROAL GRANDERS OFFICEROAL GRANDERS OFFICEROAL GRANDERS OFFICEROAL GRANDERS OFFICEROAL OFFICEROA	MERICANO LIBERT AND	AUTONE LAZHIL	THE PART OF	tech ACORD (10), Additional Rem polyworkers' compensation) where r rod with respect to General Liability	Ti	ICELLATIO	Marie Control of the			
AND EMPLOY AND PROCESS OFFICEROAL GRANDERS OFFICEROAL GRANDERS OFFICEROAL GRANDERS OFFICEROAL GRANDERS OFFICEROAL GRANDERS OFFICEROAL OFFICEROA	ASERSATIONS 1 LOS	AUTONE LAZHIL	THE PART OF	tech ACORD (10), Additional Rem polyworkers' compensation) where r rod with respect to General Liability	mana a signat si ngawan	ICELLATIO	N	DESCRIPTION OF ICES	AFF S	
AND EMPLOY A MY PROPRI OFFICERONAL GRANDERS OF PROPRI DESCRIPTE D	MERICAL DEPORT AND	AUTONE LAZHIL	THE PART OF	use Acord for Additional Rem apt workers' compensation) whose o and with respect to General Liability	CAN	ICELLATIO	N F THE ABOVE	DESCRIBED POLICIES	AFF S	
AND EMPLOY PROCESS OF	OPERATIONS (LIPS SING) SENDENCE AND SENDER AND SEN	AUTONE LAZHIL	THE PART OF	usch ACORD (so.), Additional Rem apt workers' compensation) whose o apt with respect to General Liability	CAN	ICELLATIO	N F THE ABOVE	DESCRIPTION OF ICES	AFF S	
AND EMPLOY PROCESS OF	MERICAL DEPORT AND	AUTONE LAZHIL	THE PART OF	usch ACORD (sot, Additional Rem apt workers' compensation) whose o rad with respect to General Liab My	CAP	ICELLATIO IOULD ANY O IE EXPRATI CORDANCE	n P THE ABOVE ON DAYE T WITH THE POI	DESCRIBED POLICIES	AFF S	
AND EMPLOY PROPRIES OF PROPRIES OF PROPRIES OF PROPRIES OF PROPRIES OF PORT BEND CASALTIFICATION OF POR	OPERATIONS (LIPS SING) SENDENCE AND SENDER AND SEN	AUTONE LAZHIL	THE PART OF	usch ACORD fact, Additional Rem apt workers' compensation) whose o rad with respect to General Liab My	CAN SET THE AC	ICELLATIO	n P THE ABOVE ON DAYE T WITH THE POI	DESCRIBED POLICIES HEREOF, NOTICE WILLY PROVISIONS.	DE CANC	CELLED BEFOR
AND EMPLOY PROPRIES OF PROPRIE	OPERATIONS (LIPS SING) SENDENCE AND SENDER AND SEN	AUTONE LAZHIL	THE PART OF	use ACORD fat, Additional Rem pot mothers' compensation) whose o rod with respect to General Liab My	CAN SE TH AC	ICELLATIO IOULD ANY O IE EXPRATI CORDANCE HORIZEO REPRE BORN USA Inc. Tapia Grassito	N P THE ABOVE ON DATE T WITH THE POI	DESCRIBED POLICIES	AFE S DE CANCELL BE	CELLER BEFORED BELIVERED BELIVER BEL

ACORD 25 (2010/05)

The ACORD name and logo are registered marks of ACORD

AGENCY CUST	ONER ID:	101674	
*	LOC #:	New York	

PATALISA INC.		ARKS SCHEDULE Page 2 of 2 NAMED INSURED Manageman Manageman		
FIRE REPORT	HAIGCOPE			
ER	ettivisiiseettivisiiseevat seevastateliseevateliseevatelises saaaasaa kaska	REFECTIVE DATE:		
THE PERSON NAMED IN COLUMN TO THE PE				
TIONAL REMARKS ADDITIONAL REMARKS FORM IS A SCHEDUL	E TO ACORD FORM,	·		
ADDITIONAL REMARKS FORM IS A SUPERIOR HOLE: Centification NUMBER: 25 FORM TITLE: 25	cale of Liability Insur			
end of the assurance positive referenced shows provides: that sheald su exists company will review to mor 30 days winter notice that our to representatives will not extend any policy cancellation data and will not representatives.	ch poky be conceded by the in- ne untilizate nuclei, led felter i segate any concedes of the pr	isser belong the replicition date of the premium, thousant for any reason other than non-payment of premium, the o provide such motion shall impose no californium or fability of any hind upon the insurer or its agents or idey.		
,				
		•		
1				
Table de la constant				
1				

Agreement Number

RECORDS MANAGEMENT MASTER AGREEMENT

Signature Page

WHEREAS, Ft. Bend County, Texas ("Client"), a political subdivision of the State of Texas, and Hart InterCivic, Inc. ("Hart") executed the "Fort Bend County Sublicense Agreement" on December 22, 1993; and

WHEREAS, Manatron, Inc. ("Manatron"), a Michigan corporation authorized to do business in the State of Texas, acquired Hart's Records Management Solutions Business, including the "Fort Bend County Sublicense Agreement' on September 1, 2007;

WHEREAS, Client and Manatron wish for this Records Management Master Agreement to supercede the "Fort Bend County Sublicense Agreement,"

NOW, THEREFORE, Client and Manatron desire to execute this Records Management Master Agreement

This Records Management Master Agreement ("Master Agreement") is entered into by and between Managen and Client. This Muster Agreement sets forth the general terms under which Client will purchase, license or sublicense products and services from Manatron. Manatron and Client will from time to time enter into one or more Integrated Systems Installation, Annual Software License and Maintenance and/or Professional Services Supplemental Agreement(s) ("Supplemental Agreements") which will refer to this Master Agreement and describe the specific products, services, pricing, and additional terms under which Client will purchase, license or sublicense products and services from Manatron. The combination of this Master Agreement and the Supplemental Agreements constitute the entire agreement between Manatron and Client.

The Effective Date of this Master Agreement is OCTOVEL 9

Client acknowledges it has read and understands this Master Agreement (including all Supplemental Agreements, schedules and amendments).

Agreed and Accepted:

Name:

Address:

ATTESE"

Client

Fort Bend County, Texas

301 Jackson Street, Suite 719

Richmond, Texas 77469

281-341⁾-860**8** Telephone:

281-341-8609 Facsimile:

Executed by: Robert E. Hebert

Name:

County Judge Title:

Manatron

Manatron, Inc.

510 East Milham Avenue/

Portage, Michigan 49002

John R. Hansen

Director of Risk Management

This Agreement is not effective until executed by both parties.

Dianne Wilson, County Clerk

Page 1 of 14

Version 01-06-06

L DEFINITIONS.

1 1 , 1

- 1.4 "Client" has the meaning set forth in the signature page.
- document, as amended from time to time, prepared by Manairon describing the scope and terms of Maintenance Services available from Manairon under Annual Software License and Maintenance Supplemental Agreements.
- 1.3 "Effective Date" has the meaning set forth in the signature page.
- 1.4 "Hardware" means the hardware identified in one or more integrated Systems Installation Supplemental Agreement(s).
- 1.5 "Manatron" means Manatron, Inc., a Michigan corporation authorized to do business in the State of Texas.
- 1.6 "Manatron Proprietary Software" means the executable version of computer programs and computer code owned by Manatron which are licensed to Client pursuant to one or more Annual Software License and Maintenance Supplemental Agreement(s), and all updates, upgrades, versions, new releases, derivatives, revisions, corrections, improvements, rewrites, bug fixes, enhancements and other modifications, including any custom modifications, to such computer programs and code which are provided to Client, and all copies of the foregoing. Manatron Proprietary Software also includes all documentation provided by Manation to Client with respect to these computer programs and code, excluding maintenance diagnostics, and the source code version of the programs and code when provided pursuant to a Supplemental A'greement, and all copies of the foregoing. Manairon Proprietary Software licensed to Client is identified as "Licensed Managon Proprietary Software" in Section 2.1 (Licensed Manatron Proprietary Software) of Exhibit A (Pricing and Inventory) of one or more Annual Maintenance Supplemental and License Software Agreement(s).
 - 1.7 "Installation Date" means (a) the date Manatron completes installation (as determined by Manatron), or (b) if the relevant Supplemental Agreement specifically designates such date, the installation Date as specified in such Supplemental Agreement, or, (c) if Hardware or Suftware is to be installed by Client, the tenth calendar day following receipt by Client.
 - 1.8 "Integrated System" means Hardware and/or Software that is integrated and installed by Manatron for Client pursuant to one or more integrated System Installation Supplemental Agreement(s).
 - 1.9 "Licensed Location" means the location where an item of Software is licensed for Use as designated in an Annual Software License and Maintenance Supplemental Agreement.
 - 1.10 "Licensed Server" means the Hardware on which an item of Software is licensed for Use as designated in an Version 01-06-06 Page 2 of 14

Annual Software License and Maintenance Supplemental Agreement.

- 1.11 "Licensor" means the licensor(s), respectively, of the Non-Manatron Software, as listed in Section 2.2 (Non-Manatron Sublicensed Software) and 2.3 (Non-Manatron Other Software) of Exhibit A (Pricing and Inventory) of one or more Annual Software License and Maintenance Supplemental Agreement(s).
- 1.12 "Maintenance Services" means the level of maintenance service (Level One, Basic Level of Service (Mandatory); Level Two, Extended Service (Optional); or, Level Three, Extended Service (Optional) in Exhibit A (Pricing and Inventory) of an Annual Software License and Maintenance Supplemental Agreement selected by Client, which Manaton will provide to Client under such Supplemental Agreement. The Maintenance Services are more specifically described in the Description of Maintenance Services attached to such Supplemental Agreement.
- 1.13 "Master Agreement" has the meaning sot forth in the signature page.
- 1.14 "Non-Manatron Software" means the executable version of computer programs owned by third parties that are provided by Manatron to Client pursuant to sublicense(s) under one or more Annual Software License and Maintenance Supplemental Agreement(s) or license(s) directly from the third party Licensor, and all updates, upgrades, versions, new releases, derivatives, revisions, corrections, improvements, rewrites, bug fixes, enhancements and other modifications to such computer programs which are provided to Client, and all copies of the foregoing. Non-Manatron Software also includes all documentation provided to Client with respect to these computer programs. Non-Manatron Software provided to Client is identified as "Non-Manatron Sublicensed Software" or "Non-Manatron Other Software" in Sections 2.2 and 2.3, respectively, of Exhibit A (Pricing and Inventory) of one or more Annual Software License and Maintenance Supplemental Agreement(s).
 - 1.15 "Non-Manatron Other Software" means Non-Manatron Software that is licensed by a third party Licensor directly to Client. Non-Manatron Other Software is identified in Section 2.3 (Non-Manatron Other Software) of Exhibit A (Pricing and Inventory) of one or more Annual Software License and Maintenance Supplemental Agreement(s).
 - 1.16 "Non-Manatron Sublicensed Software" means Non-Manatron Software that is sublicensed by Manatron to Client pursuant to sublicense(s) under one or more Annual Software License and Maintenance Supplemental Agreement(s). Non-Manatron Sublicensed Software is identified in Section 2.2 (Non-Manatron Sublicensed Software) of Exhibit A (Pricing and Inventory) of one or more Annual Software License and Maintenance Supplemental Agreement(s).

- 1.17 "Number of Licensed Users" means the number of users licensed to use an item of Manatron Proprietary Software or Non-Manatron Subdicensed Software as set forth in Exhibit A to the applicable Annual Software License and Maintenance Supplemental Agreement.
- 1.18 "Products" means the Hardware, Software, and all other documentation provided by Manatron to Client under this Master Agreement and any Supplemental Agreements.
- 1.19 "Proprietary and Confidential Information" means Software, diagnostics, documentation (including manuals), Hardware and Software configuration, Integrated Systems design and configuration, training materials, user guides, trade secrets, source code and related documentation, and any other information confidential to Manatron or its Proprietary and Confidential suppliers or Licensons. Information shall be deerned not to include information that (a) is or becomes (other than by disclosure by Client) publicly known or is contained in a publicly available document; (b) is furnished by Manatron to others without restrictions similar to those imposed by this Master Agreement and the Supplemental Agreements; (c) is rightfully in Client's possession without the obligation of nondisclosure prior to the time of its disclosure under this Master Agreement and the Supplemental Agreements or any prior agreements between Manatron and Client; or (d) is independently developed by employees or agents of Client who can be shown to have had no access to the Proprietary and Confidential Information.
 - 1.20 "Services" means the services to be performed by Manatron for Client as identified on one or more Supplemental Agreement(s).
 - 1.21 "Software" means the Manatron Proprietary Software and Non-Manatron Software.
 - forth on the signature page. The types of Supplemental Agreements are: Integrated Systems Installation Supplemental Agreement, Annual Software License and Maintenance Supplemental Agreement and Professional Services Supplemental Agreement.
 - 1.23 "Use" means reading the Manatron Proprietary Software or Non-Manatron Sublicoused Software into and out of memory and the execution of such Software, in whole or in part, by the Licensed Server.

2. SUPPLEMENTAL AGREEMENTS.

2.1 Supplemental Agreements. Manatron will furnish to Client and Client will accept and pay for Products, Services and Integrated Systems itemized on Supplemental Agreements entered into by Client and Manatron, which together with the terms in the Supplemental Agreements, are an integral part of this Master Agreement. Supplemental Agreements will refer to this Master Agreement by number and will be signed by Client and Manatron. All references to Products, Services and Integrated Systems in this Master Agreement are to the Products, Services and Integrated

Version 01-06-06 Page 3 of 14

- Systems listed on any Supplemental Agreements submitted to and accepted by Manairon pursuant to Sections 2.2 and 2.3, as modified by any Change Requests entered into by Client and Manairon pursuant to Section 2.4.
- 2.2 Additional Requests. Client may order additional Products, Services and Integrated Systems under this Master Agreement by submitting properly completed Supplemental Agreements referencing this Master Agreement, signed by an authorized representative of Client. Under no circumstances will additional orders be effective until accepted in writing by Manatron.
- 2.3 Subject to Acceptance. All Supplemental Agreements are subject to acceptance by Manatron. Manatron's acceptance will be effective when Manatron signs the Supplemental Agreements. Manatron's receipt or deposit of a Client purchase order or down payment will not constitute acceptance of a Supplemental Agreement. Manatron will return any down payment received from Client if Manatron does not accept the Supplemental Agreement.
- 2.4 Change Request. Client and Manatron may at any time modify a Supplemental Agreement by written Change Request, signed by both parties, identifying the modified Supplemental Agreement and specifying the modifications to at least the same degree of specificity as in the original specifications. The Change Request will include all changes and additions being made to the terms of the applicable Supplemental Agreement. Manatron will not be bound by any modifications to a Supplemental Agreement unless made by written Change Request signed by authorized representatives of both parties. A Change Request, when signed by both parties, will be subject to the terms of the applicable Supplemental Agreement, as modified by the Change Request, and this Master Agreement.
- 2.5 Substitution. With notification to Client and Client's prior approval, such approval not to be unreasonably denied, Manatron may substitute Product(s) of equivalent or superior functionality and performance in the event that any of the Product(s) ordered are not available. If Manatron reasonably determines that the substitute Product(s) would be more suitable, this substitute will be documented on the Change Request Form, modifying the Hardware or Software listed in Eschibit A. (Pricing and Inventory) of the applicable Supplemental Agreement(s) and submitted to Client for approval, which approval shall not be unreasonably withheld.
- 2.6 Training. Manatron will provide training to Client's personnel as specified in one or more Integrated System Installation Supplemental Agreement(s) or Professional Services Supplemental Agreement(s). Client will be responsible for all training not specifically provided for in Supplemental Agreement(s).

3, LICENSES AND SUBLICENSES.

3.1 <u>Supplemental Agreements</u>. Manatron will provide Client with licenses and sublicenses under Section 3.2 to the Manatron Proprietary Software and Non-Manatron

Sublicensed Software, respectively, which is specified by Manatron and Client in Exhibit A (Pricing and Inventory) of one or more Annual Software License and Maintenance Supplemental Agreement(s). Licenses and sublicenses provided under an Annual Software License and Maintenance Supplemental Agreement will begin with the date of the Annual Software License and Maintenance Supplemental Agreement and end upon termination or expiration of such agreement. The terms of this Agreement, including but not limited to this Section 3, and the terms of the applicable Annual Software License and Maintenance Supplemental Agreement will apply to all licenses of Manatron Proprietary Software and to all sublicenses of Non-Manairon Sublicensed In some cases, Non-Manatron Software (consisting of Non-Manatron Other Software) may be provided subject to a license directly from the Licensor to Client. If a separate license agreement applies to or accompanies Non-Manatron Software, then the separate license agreement terms will apply and supersede the license terms in this Master Agreement and the Supplemental Agreements for that Non-Manatron Software. Client agrees to comply with the terms of all licenses governing Manatron Proprietary Software and Non-Manatron Software.

3.2 Licenses and Sublicenses. Manatron grants to Client a personal, non-exclusive, nontransferable limited license or sublicense to Use the Manarron Proprietary Software and Non-Manatron Sublicensed Software, respectively, which is identified as licensed or sublicensed to Client in Exhibit A (Pricing and Inventory) of one or more Annual Software License and Maintenance Supplemental Agreement(s). Client is licensed or sublicensed to Use the Manatron Proprietary Software and Non-Manatron Sublicensed Software only for Client's internal data processing requirements at the Licensed Location on the Licensed Server by the Number of Licensed Users, each as specified in the applicable Annual Software License and Maintenance Supplemental Agreement. Client may temporarily transfer the Manatron Proprietary Software and Non-Manatron Sublicensed Software to a back-up server at an alternative location within Client's county of operation if the Licensed Server is inoperative or the Licensed Location is temporarily unavailable. Unless otherwise provided in the applicable Annual Software License and Maintenance Supplemental Agreement, Client will only be provided and permitted to use the executable form of Manatron Proprietary Software and Non-Manatron Sublicensed Software and such use must be in connection with the application package provide by Manatron. Client agrees that Manatron or its representatives may periodically inspect and audit, at rhutually agreed upon times during normal business hours, the consputer site, Integrated Systems and appropriate records of Client to verify Client's compliance with the terms of this Master Agreement and all applicable Supplemental Agreement(s) with respect to the Software supplied by Manatron.

3.3 Protection of Software.

- (a) Client shall not; under any circumstances, cause or permit the adaptation, conversion, reverse engineering, disassembly, modification or de-compilation of any Software. (b) Client shall not modify, assign, transfer, sublicense, time-share, rent, copy or duplicate the Software; provided, Client may have in its possession a reasonable number of copies of the Software for inactive archival or back-up purposes. All copies of the Software, in whole or in part, must contain all of Manatron's and the third party Licensor's titles, trademarks, copyright notices and other restrictive and proprietary notices and legends (including government restricted rights) as they appear on the copies of the Software provided to Client. Client shall notify Manatron of the following:
- (i) the location of all Software and all copies thereof, and
- (ii) any circumstances known to Client regarding any unauthorized possession or use of the Software.
- (c) Upon termination of Client's license or sublicense of Software, Client shall immediately discontinue all use of the Software and return to Manatron or destroy, at Manatron's option, the Software (and all related documentation and Proprietary and Confidential Information) and all archival, back-up and other copies thereof, and provide certification to Manatron of such return or destruction.
- (d) Client shall not publish any results of benchmark tests run on any Software.
- (e) Client shall not relocate the Manatron Proprietary Software or Non-Manatron Sublicensed Software from the Licensed Location, except as permitted in Section 3.2.
- (f) Client shall maintain the Manatron Proprietary Software and Non-Manatron Sublicensed Software in confidence and comply with the terms of Section 7. Protection of Proprietary and Confidential Information, with respect to such Software.
- (g) The terms of this Section 3.3 will survive the termination or expiration of this Agreement and the applicable Annual Software License and Maintenance Supplemental Agreement.
- 3.4 No Transfer of Title. This Agreement does not transfer to Client title to any Software, intellectual property contained in any Software, or Proprietary and Confidential Information. Title to Manatron Proprietary Software and all copies thereof, and all associated intellectual property rights therein, will remain in Manatron. Title to Non-Manatron Software and all copies thereof, and all associated intellectual property rights therein, will remain in the applicable third party Licensor.
- 3.5 Inherently Dangerous Applications. The Software is not developed or licensed for use in any nuclear, aviation, mass transit, or medical application or in any other inherently dangerous applications. Client shall not use the Software in any inherently dangerous application and agrees that

Manatron and any third party licensor will not be liable for any claims or damages arising from such use.

4. MAINTENANCE SERVICES

- 4.1 Maintenance. Manatron will provide Maintenance Services to the extent provided for in Annual Software License and Maintenance Supplemental Agreement(s) or Professional Services Supplemental Agreement(s) and pursuant to the terms and conditions of Manatron's Description of Maintsnance Services, as amended from time to time. Manatron will make best commercially reasonable efforts to remedy or provide a reasonable work-around for defects, errors or malfunctions in Manatron Proprietary Software which have a significant adverse affect upon operation of the Manatron Proprietary Software or Integrated System, as applicable, and which are promptly reported by Client to Manatron. Because not all errors or defects can or meed be corrected, it is possible that some errors or defects will not be corrected. Errors or defects must be reported on Manatron's Client Service Request Form, Exhibit C to the applicable Annual Software License and Maintenance Supplemental Agreement and be accompanied with sufficient detail, if available, to enable Manatron to reproduce the error and provide a remedy or suitable work-around. The remedies and obligations set forth in this Section 4.0 are the full extent of Client's remedies and the full extent of Manatron's obligations with respect to errors, defects or malfunctions of the Integrated System, Manatron Proprietary Software and Non-Manatron Sublicensed Software.
 - 4.2 Limitation. Manatron will not provide Maintenance Services if alterations to Products or Integrated Systems which are not made by or coordinated with Manatron or attachments to Products or Integrated Systems which are not provided and installed by or coordinated with Manatron directly or indirectly result in any malfunction, nonperformance or degradation of performance of Products or Integrated Systems.

4.3 Exclusions.

- (a) Maintenance Services apply only to properly This exclusion for improper configured Products. configuration is not applicable to Maintenance Services for any Hardware or Software installed by Manatron or under Manatron supervision, unless Client or any third party has changed the configuration without Manatron's supervision. Maintenance Services also apply only at the minimum Hardware and Software levels designated by Menauron for support of the applicable Product specifications.
- (b) Maintenance Services do not include correction or repair of defects, errors or malfunctions, including any related to date data functionality, in the design, manufacture, materials or workmanship of either (i) Non-Manatron Other Software, or (ii) Hardware.
- (c) Maintenance Services described in this Section 4 and under Annual Software License and Maintenance Supplemental Agreement(s), including the Description of

Version 01-06-06

Page 5 of 14

- Maintenance Services exhibit, do not cover defects, errors or malfunctions which are not attributable to the relevant Manatron Proprietary Software or Integrated System or which are caused by any of the following: (i) de-installation, reinstallation or relocation of any item of Hardware by Client or any third party; (ii) Client's failure to follow operational or maintenance instructions as set forth in applicable documentation; (iii) the use of non-compatible media or supplies; (iv) repair, maintenance, modification or alteration of the Manatron Proprietary Software, Hardware or Integrated System by Client or third parties; (v) use of hardware or software not supplied or authorized by Manatron; (vi) external factors (e.g., power failure, surges or electrical damage, fire or water damage, air conditioning failure, humidity control failure, or corrosive atmosphere harmful to electronic circuitry); (vii) failure to maintain proper site specifications and onvironmental conditions; (viii) negligence, accidents, neglect, misuse or tampering; (ix) improper or abnormal use or use under abnormal conditions; (x) use in a manner not authorized by this Master Agreement and any Supplemental Agreement(s) or use inconsistent with Manatron's specifications; (xi) Client's failure to comply with Client's responsibilities under Section 6; (xii) attachments or alterations not provided and installed by Manatron as further described in Section 4.4 or (xiii) the introduction of software viruses.
 - (d) Manatron reserves the right to charge on a time and materials basis for efforts expended due to problems caused by these maintenance exclusions, using Manatron's standard time and materials charges, subject to Client's prior approval.

4.4 Alterations and Attachments.

- (a) Client will be solely responsible for infringement, personal injury or damage to property, Products and Integrated Systems resulting from alterations to Products or Integrated Systems that are not made by Manatron or attachments to Products or Integrated Systems that are not provided and installed by Manatron.
- (b) Client will give Manatron prior written notice of any proposed alterations or attachments to Products or Integrated Systems subject to Maintenance Services. Manatron has no obligation to provide Maintenance Services for Products or Integrated Systems containing alterations not made by Manatron or attachments not provided and installed by Manatron. If Manatron agrees to maintain, support or correct altered Products or Integrated Systems, Manatron may impose additional fees. Manatron is not responsible for a mulfunction, nonperformance or degradation of performance of Products or Integrated Systems caused by or resulting directly or indirectly from any alteration or attachment unless Manatron has provided and installed the alteration or attachment and has informed Client that such adverse conditions will not occur. For purposes of this Master Agreement "alterations" includes, but is not limited to, the incorporation of components, boards and subassemblies not provided by Manatron into Products or Integrated Systems, as

well as modifications to Products or Integrated Systems that are not made by Manatron. "Attachments" includes any hardware, software, components or devices which are connected to Products or Integrated Systems and which are not provided by Manatron.

5. CHARGES; PAYMENT.

- 5.1 Charges. Charges for Products, Services and Integrated Systems will be identified and payable in accordance with the terms set forth in the relevant Supplemental Agreement(s). All payments are to be made to Manatron at its principal office in Austin, Texas, as set forth on the signature page or to such other location as may be designated by Manatron in a notice to Client.
- 5.2 Late Charges. All uncontested charges must be paid as agreed in a Supplemental Agreement. If invoiced, all payments are due no later than thirty (30) days from the date of receipt of invoice by Client. Manatron may impose a late payment charge on past due payments equal to the lesser of (a) 1% per month or (b) the maximum rate allowed by law.
- 5.3 Additional Charges. Subject to Manatron's notification to Client and Client's prior approval, additional charges may be assessed for services rendered outside contracted hours or boyond normal coverage at Client's request, including but not limited to travel expenses, premium and minimum charges.
- 5.4 Payment Disputes. If any dispute exists between the parties concerning the amount due or due date of any payment, Client shall promptly pay the undisputed portion. Such payment will not constitute a waiver by Client or Manatron of any of their respective legal rights and remedies against each other.

5.5 Price Protection.

- (a) The prices for Products in any Supplemental Agreement will remain firm through delivery, unless through no fault of Manatron shipment takes place more than one year after the date of the Supplemental Agreement. If Manatron notifies Client that an increase in prices will apply to its order and the affected part of the order is not part of an Integrated System, Client may terminate the affected part of its order by giving written notice to Manatron within ten (10) days of the date of notification of the increase.
- (b) Fees for Maintenance Services under an Annual Software License and Maintenance Supplemental Agreement will not be increased during the twenty-four (24) month period beginning with the Installation Date, but may be increased on the second (2nd) anniversary of the Installation Date and on each subsequent anniversary of the Installation Date with thirty (30) days prior written notice to Client.
- (c) Fees for Software licenses, Software sublicenses and Services under Supplemental Agreements of at least one year will not be increased during the twenty-four (24) month period beginning with the Installation Date, but may be increased on the second (2nd) anniversary of the Installation

Date and on each subsequent anniversary of the Installation Date with thirty (30) days prior written notice to Client. If Software or Services are contracted on a month-to-month basis, the fees may be increased at any time with sixty (60) days prior written notice to Client.

- (d) Manatron will limit the increase in annual license and support fees to 5% or the CPI, whichever is greater. The base for computing the adjustment is the Consumer Price Index All Urban Consumers for the Houston/Harris County area published by the United States Department of Labor, Eureau of Labor Statistics ("Index"). The Index published most immediately preceding the Adjustment Date in question ("Extension Index") is to be used in determining the amount of the increase.
- 5.6 Taxes. If Client is tax exempt, Client will provide Manatron with proof of its tax-exempt status. If Client is not tax-exempt, (a) Client will pay any tax Manatron becomes obligated to pay in connection with this Agreement, exclusive of taxes based on the net income of Manatron, and (b) Client will pay all personal property and similar taxes assessed after shipment. If Client challenges the applicability of any such tax, Client shall pay the tax and may thereafter seek a refund.
- 5.7 <u>Delivery</u>. Unless otherwise specified in the relevant Supplemental Agreement, Manatron will arrange for delivery of Integrated Systems and Products to Client and delivery charges will be included in Manatron's pricing.
- 5.8 Installation. Unless otherwise provided in the relevant Supplemental Agreement, if the Supplemental Agreement provides for Manatron to install Integrated Systems or Products, (a) installation will be performed during Manatron normal working hours, (b) all installation will be subject to the then-current standard Manatron charges and conditions, and (c) if additional labor and rigging is required for installation due to Client's special site requirements, Client will pay those costs including costs to meet union or local law requirements.

6. CLIENT RESPONSIBILITIES.

- 6.1 Independent Determination. Client acknowledges that, based on Manatron's representations and Client's own due diligence, Client has determined that the Products, Services and Integrated Systems ordered under this Master Agreement and Supplemental Agreement(s) meet its requirements.
- 6.2 <u>Cooperation</u>. Client agrees to cooperate with Manatron and promptly perform Client's responsibilities under this Master Agreement, including but not limited to those set forth in any Supplemental Agreement. Client shall:
- (a) provide adequate working and storage space for use by Manatron personnel near Hardware and Integrated Systems;
- (b) provide Manaron full access to the Hardware, Software and Integrated Systems and sufficient computer time, subject only to Client's security rules;

- (c) follow Manatron's procedures for placing service requests and determining if remedial service is required;
- (d) follow Manatron's or manufacturer's instructions for operator maintenance and obtaining services;
- (e) provide a memory dump and additional data in machine-readable form if requested;
- (f) reproduce suspected errors or malfunctions in Software;
- (g) install all Manatron Software and Non-Manatron Software releases supplied by Manatron within ninety (90) days after receipt of such release by Client or, if the applicable Supplemental Agreement provides for Manatron to install the releases, then Client shall fully cooperate with Manatron's installation of the Manatron Software and Non-Manatron Software releases, as applicable;
- (h) provide timely access to key customer personnel and timely response to Manarron's questions; and
- (i) otherwise cooperate with Manatron in its performance under this Master Agreement and Supplemental Agreements.
- 6.3 Site Preparation. Client is responsible for compliance with all local labor concerns and building codes. If Manatron is to install Products, Client shall prepare and maintain the installation site in accordance with the manufacturer's instructions and instructions provided by Manatron and ensure that these instructions are not in violation of labor laws or building ordinances. Client is responsible for environmental requirements, electrical interconnections and modifications to facilities for proper installation. Any delays in preparation of the installation site will correspondingly extend Manatron's delivery and installation deadlines.
- 6.4 Site Maintenance. Client shall maintain the appropriate operating environment; in accordance with the manufacturer's specifications and Manatron's specifications, for the Hardware, Software and Integrated Systems and all communications hardware, telephone lines, electric lines, cabling, moderns, air conditioning and all other hardware and utilities necessary for the Hardware, Software and integrated Systems to operate properly.
- 6.5 Use. Client is exclusively responsible for supervising, managing and controlling its use of the Hardware, Software and Integrated Systems, including but not limited to, establishing operating procedures and audit controls, supervising its employees, making daily backups, providing virus protection, inputting data, ensuring the accuracy and security of data input and data output, monitoring the accuracy of information obtained, and managing the use of information and data obtained. Client will ensure that its personnel are, at all times, educated and trained in the proper use and operation of the Products and that the Products are used in accordance with applicable manuals, instructions and specifications.

- 6.6 Backups. Client will maintain back-up data necessary to replace critical Client data in the event of loss or damage to data from any cause.
- 6.7 Compliance with Third Party Contracts. Client represents and warrants to Manatron that it does not have any contracts or other obligations to third parties, including but not limited to any license agreements or confidentiality obligations, that will be violated in any respect by Client's or Manatron's performance under this Master Agreement or any Supplemental Agreement(s).
- 7. PROTECTION OF PROPRIETARY AND CONFIDENTIAL INFORMATION.
- 7.1 Manatron Proprietary and Confidential Information, Client shall keep in confidence and protect Proprietary and Confidential Information from disclosure to third parties and restrict its use to uses expressly permitted under this Muster Agreement and the Supplemental Agreements subject to the requirements of the Texas Public Information Act. Client shall take all reasonable steps to ensure that Proprietary and Confidential Information is not disclosed, copied, duplicated. misappropriated or used in any manner not expressly permitted by the terms of this Master Agreement and Supplemental Agreement(s). Client shall keep the Software and all tapes, CDs, diskettes and other physical embodiments of the software, and all copies thereof, at a secure location and limit access to those employees who must have access to enable Client to use Client acknowledges that unauthorized the Software. disclosure of Proprietary and Confidential Information may cause substantial economic loss to Manatron or its suppliers and licensors. Client agrees not to copy Proprietary and Confidential Information, in whole or in part, except as expressly authorized by this Master Agreement and any Supplemental Agreement(s). Each permitted copy of Proprietary and Confidential Information, including its storage media, will be marked by Client to include all notices and legends (including government restricted rights) that appear on the original. Title, copyright and all other proprietary rights in and to the Software, at all times remain vested exclusively in Manatron or, as applicable, the respective third party licensors.
 - Client Confidential Information. 7.2 acknowledges that it and its employees or agents may, in the course of performing their responsibilities under this Master Agreement and the Supplemental Agreements, be exposed to or acquire information that is confidential to the Client. Any and all information of any form obtained by Manairon or its employees or agents in the performance of this Master Agreement and the Supplemental Agreements shall be deemed to be confidential information of the Client. Any reports or other documents or items (including software) that result from the use of the confidential information by Manatron shall be treated with respect to confidentiality in the same manner as the Proprietary and Confidential Information. Confidential information shall be deemed not to include information that (a) is or becomes (other than by disclosure by Manatron) publicly known or is contained in a publicly.

available document, (b) is furnished by the Client to others without restrictions similar to those imposed by this Master Agreement and the Supplemental Agreements; (c) is rightfully in Manatron's possession without the obligation of mondisclosure prior to the time of its disclosure under this Master Agreement and the Supplemental Agreements; or (d) is independently developed by employees or agents of Manatron who can be shown to have had no access to the confidential information.

- Information. Upon termination of cancellation of this Master Agreement, Client shall immediately destroy or return to Manatron, at Manatron's option, all Proprietary and Confidential Information in Client's possession or control (including Proprietary and Confidential Information incorporated in other software or writings) and provide certification to Manatron of such return or destruction. Upon termination of a Supplemental Agreement, Client shall immediately destroy or return, at Manatron's option, all Proprietary and Confidential Information in Client's possession or control (including Proprietary and Confidential Information incorporated in other software or writings) relating to such Supplemental Agreement, which is no longer licensed or sublicensed to Client pursuant to another Supplemental Agreement.
 - 7.4 Intellectual Properties. All ideas, concepts, know-how, data processing techniques, Software, documentation, diagrams, schematics, blueprints and trade secrets developed by Manatron personnel (alone or jointly with Client) in connection with Proprietary and Confidential Information or Products, Services and Integrated Systems provided to Client will be the exclusive property of Manatron. Manatron grants to Client a non-exclusive, royalty-free license to use of any of the foregoing in connection with Client's use of the Products and/or Integrated System as permitted by the terms of this Master Agreement and the Supplemental Agreement(s).
 - 7.5 Support Maierials. Client acknowledges that all support materials, including without limitation, diagnostic software, are the property of and include Proprietary and Confidential Information of Manatron. Client will not use such materials. Manatron has the right to remove such materials from Client's facility at any time. This provision applies unless Client has specifically purchased or licensed such support materials pursuant to a Supplemental Agreement.
 - 7.6 Client Employees. Client will inform its employees of their obligations under this Section 7 to ensure such obligations are met.
 - 7.7 Survival. This Section 7 will survive termination or cancellation of this Agreement and the Supplemental Agreement(s).
 - 8, TITLES; RISK OF LOSS.

8.1 Hardware. Title to Hardware and risk of loss of or damage to Hardware will pass to Client upon delivery to Client of Manatron as Client's agent.

8.2 Software.

- (a) Manatron Proprietary Software. Title to Manatron Proprietary Software and all associated intellectual property rights will remain in Manatron, including but not limited to all copyrights, trade secrets, trademarks and other proprietary rights in and to the Manatron Proprietary Software.
- (b) Non-Manatron Software. Title to Non-Manatron Software and all associated intellectual property rights will remain in the applicable third party Licensor including but not limited to all copyrights, trade secrets, trademarks and other proprietary rights in and to the Non-Manatron Software.
- (c) <u>Risk of Loss</u>. Risk of loss to Software will pass to Client on the Installation Date, except to the extent covered by the limited warranties in Section 9 or Maintenance Services pursuant to an Annual Software License and Maintenance Supplemental Agreement, as applicable.
- 8.3 Data. Client will retain all title, rights, and ownership of all images and associated indexes, and other data created antifor acquired by use of the Hardware, Software and Integrated Systems as stored on magnetic disk, magnetic tape, optical disk, optical tape, CD-ROM disk (or other "like" electronic media that may be used).
- 8.4 Proprietary and Confidential Information. Title to Manatron's Proprietary and Confidential Information will remain in Manatron. Title to Proprietary and Confidential Information of Manatron's suppliers and licensors will remain in the relevant suppliers and licensors. Title to Client's confidential information will remain in Client.
 - 9. REPRESENTATIONS AND WARRANTIES.

9.1 Software.

- (a) <u>Title</u>. <u>Manatron</u> represents and warrants that it owns or has the right to license or sublicense the Manatron Proprietary Software and Non-Manatron Sublicensed Software licensed or sublicensed by Manatron to Client as provided for under Annual Software License and Maintenance Supplemental Agreement(s) or other prior Agreements.
- (b) Services. Manatron warrants to Client that Manatron has the skill and knowledge ordinarily possessed by well-informed members of its trade and profession and Manatron will apply that skill and knowledge with care and diligence to perform Services under the Supplemental Agreements in accordance with best commercially reasonable professional standards and in accordance with industry standards. If Client believes Manatron has breached this warranty, Client shall promptly, and in any event within fourteen (14) days after Client becomes aware of the breach, notify Manatron in writing of the claimed breach, with substantiating documentation, and request that Manatron re-perform the

Services subject to the breach. If there is a breach of this warranty, Manatron shall re-perform the Services for which there has been a breach. Manatron will not be responsible under this warranty if Client fails to timely notify Manatron of the claimed breach or if Client fails to substantiate the breach with documentation. The remedy set forth in this section is Client's sole and exclusive remedy, and the full extent of Manatron's liability, for Manatron's breach of this warranty.

(c) Conformance to Specifications. Monatron warrants to Client that upon the Installation Date or, if there is an acceptance testing process upon acceptance by Client, the Hardware, Software and Integrated System, as applicable, will conform and perform materially to the specifications in the related Supplemental Agreement. Client shall inspect the Hardware Software and Integrated System for compliance with this warranty promptly upon receipt and installation. If Client believes Manatron has breached this warranty, Client shall within fourteen (14) days after Client knew or should have known of the breach, and in any event no later than ninety (90) days after the Installation Date (or date of acceptance if there is an acceptance testing process), notify Manatron in writing of the claimed breach, with substantiating documentation, and request that Manatron provide compliant Hardware, Software and Integrated System, as applicable, or, at Manatron's election, refund the amount paid for the Hardware, Software and Integrated System subject to the breach. If there is a material breach of this warranty, Manatron will, at Manatron's election, either provide materially compliant Hardware, Software and Integrated System, as applicable, for which there has been a material breach or refund the amount paid by Client with respect to the Hardware, Software and Integrated System, as applicable, for which there has been a material breach. Because not all errors or defects can or need to be corrected, it is possible that some errors or defects will not be corrected. Errors or defects relating to Software must be reported on Manatron's Client Service Request Form, Exhibit C to the applicable Annual Software License and Maintenance Supplemental Agreement and he accompanied with sufficient detail, if available, to enable Manarron to reproduce the error and provide a remedy or suitable work-around, Manatron will not be responsible under this warranty if Client fails to promptly and timely notify Manatron of the claimed breach or if Chient fails to substantiate the breach with documentation. Client further acknowledges and agrees that this warranty will be deemed to be satisfied upon acceptance by Client of the Hardware, Software and/or Integrated System, as applicable, pursuant to Client's inspection or pursuant to acceptance testing done with respect to the Hardware. Software and/or Integrated System, as applicable. This warranty does not apply if the failure of the Hardware, Software or Integrated System to conform to this warranty is caused by any of the exclusions from Maintenance Services that are set forth in Section 4.2, Section 4.3, and/or Section 4.4 which are incorporated herein by reference as exceptions

to this warranty. The remedy set forth in this section is Client's sole and exclusive remedy, and the full extent of Manatron's liability, for Manatron's breach of this warranty.

- 9.2 Initial Maintenance Period at No Extra Charge. During the one-year period beginning with the Installation Date, in lieu of a one (1) year warranty, Manatron will provide to Client, at no extra charge, Maintenance Services at Level One, Basic Level of Service on the terms and conditions (including the exclusions, limitations and restrictions) set forth in the applicable Annual Software License and Maintenance Supplemental Agreement, Manatron's Description of Maintenance Services, as amended from time to time, and Sections 4, and 9.8 of this Master Agreement. Client's remedies are limited to Maintenance Services as set forth in this Section 9.2.
- 9.3 Conversion. If Manatron converts Client's legacy data pursuant to a Supplemental Agreement, Manatron warrants, for the benefit of Client only, that Manatron has converted the legacy data as extracted by Client into the Integrated System as directed by Client. Client shall review the converted data and Client's sole remedy and Manatron's sole obligation for conversion services will be to correct any errors caused by conversion of the data by Manatron, as detected by Client. Manatron will not be obligated to correct errors in the data provided to Manatron. The remedies and obligations set forth in this Section 9.3 are the full extent of Client's remedies and the full extent of Manatron's obligations with respect to errors caused by the conversion of data by Manatron.
- 9.4 Non-Manatron Software. EXCEPT AS OTHERWISE SPECIFICALLY PROVIDED IN SECTION 9.1 (c); MANATRON MAKES NO OTHER REPRESENTATIONS WARRANTIES AS TO NON-MANATRON SOFTWARE, ALL OF WHICH IS SOLD OR LICENSED TO CLIENT "AS IS." Manatron will pass through to Client, on a non-exclusive basis and without recourse to Manatron. any third party manufacturer's and Licensor's warranties covering Non-Manatron Software, but only to the extent, if any, permitted by the third party manufacturer and Licensor. Client may independently seek to obtain directly from the manufacturers or Licensors of the Non-Manatron Other Software maintenance of the Non-Manatron Other Software under any warranty or guarantee provided by such third party manufacturer or Licensor. Client agrees to look solely to the warranties and remedies, if any, provided by the third party manufacturer or Licensor. The remedies and obligations set forth in this Section 9.4 are the full extent of Client's remedies and the full extent of Manatron's obligations with respect to warranties of Non-Manatron Software.
- 9.5 Hardware. EXCEPT AS OTHERWISE SPECIFICALLY PROVIDED IN SECTION 9.1 (c), MANATRON MAKES NO OTHER REPRESENTATIONS OR WARRANTIES AS TO HARDWARE, ALL OF WHICH IS SOLD OR LICENSED TO CLIENT "AS IS."

MANATRON will pass through to Client, on a non-exclusive basis and without recourse to MANATRON, any third party manufacturer's warranties covering the Hardware, but only to the extent, if any, permitted by the third party manufacturer. Client may independently seek to obtain directly from the manufacturers of the Hardware maintenance of the Hardware under any warranty or guarantee provided by such third party manufacturer. Client agrees to look solely to the warranties and remedies, if any, provided by the manufacturer. The remedies and obligations set forth in this Section 9.5 are the full extent of Client's remedies and the full extent of MANATRON's obligations with respect to warranties of Hardware.

9.6 Compliance with Third Party Contracts. MANATRON represents and warrants to Client that it does not have any contracts or other obligations to third parties, including but not limited to any license agreements or confidentiality obligations, that will be violated in any respect by MANATRON's or Client's performance under this Master Agreement or any Supplemental Agreement(s).

9.7 Proof of Insurance.

- Manairon shall obtain and maintain, throughout the term of the Master Agreement and the Supplemental Agreements, insurance of the types and in the minimum amounts set forth below. Manatron shall turnish certificates of insurance to Client evidencing compliance with the insurance requirements hereof. Certificates shall indicate Menatron's name, name of insurance company, policy number, term of coverage and limits of coverage. Manatron shall cause its insurance companies to provide Client with at least thirty (30) days prior written notice of any reduction in the limit of liability by endorsement of the policy, cancellation or non-tenewal of the insurance coverage required under this Agreement. Manatron shall obtain such insurance from such companies having a Bests rating of B+/VII or better, licensed or approved to transact business in the State of Texas, and shall obtain such insurance of the following types and minimum limits:
 - (i) Workers' Compensation insurance in accordance with the laws of the State of Texas, or state of hire/location of Services, and Employers' Liability coverage with a limit of not less than \$500,000 each employee for Occupational Disease, \$500,000 policy limit for Occupational Disease; and Employer's Liability of \$500,000 each accident.
 - (ii) Employers' Liability insurance with limits not less than \$1,000,000 per injury by accident, \$1,000,000 per injury by disease, and \$1,000,000 per bodily injury by disease.
 - (iii) Commercial General Liability insurance including coverage for Products/Completed Operations, Blanket Contractual, Contractor's Protective Liability Broad Form Property Damage, Personal Injury/Advertising Liability, and Bodily Injury and Property Damage with limits not less than:

\$2,000,000 annual aggregate limit

Version 01-06-06 Page 10 of 14

- \$1,000,000 cach occurrence, combined single limit
- (iv) Business Automobile Liability coverage applying to owned, non-owned and hired automobiles with limits not less than \$1,000,000 each occurrence combined single limit for Bodily injury and Property Damage combined.
- (v) Professional Liability insurance with limits not less than \$1,000,000 each claim/annual aggregate.
- (b) Client shall be named as additional insured to all coverages required above. All policies written on behalf of Manatron shall contain a waiver of subrogation in favor of Client.
- MANATRON DOES NOT 9.8 DISCLAIMER. REPRESENT OR WARRANT THAT THE FUNCTIONS MANATRON PROPRIETARY IN CONTAINED SOFTWARE OR INTEGRATED SYSTEMS WILL MEET THAT ARE NOT CLIENT'S REQUIREMENTS FORTH IM APPLICABLE **EXPRESSLY** SET SUPPLEMENTAL AGREEMENTS. MANATRON DOES NOT REPRESENT OR WARRANT THAT SOFTWARE PROPRIETARY OR MANATRON WILL OPERATE IN INTEGRATED SYSTEMS COMBINATIONS SELECTED FOR USE BY CLIENT WITH HARDWARE OR SOFTWARE NOT APPROVED EXCEPT FOR THE EXPRESS BY MANATRON. WARRANTIES IN THIS MASTER AGREEMENT, TO PERMITTED EXTENT MUMIXAM APPLICABLE LAW: (A) THERE ARE NO WARRANTIES, IMPLIED BY OPERATION OF LAW OR OTHERWISE, AND (B) MANATRON DISCLAIMS ALL WARRANTIES, INCLUDING BUT NOT IMPLIED LIMITED TO, THE IMPLIED WARRANTIES OF FITNESS FOR A PARTICULAR PURPOSE AND MERCHANTABILITY FOR MANATRON AND NON-MANATRON PRODUCTS AND SERVICES. EXPRESS LIMITED WARRANTIES EXTEND SOLELY TO CLIENT.

10. LIMITATION OF DAMAGES.

- 10.1 EXCLUSIVE REMEDY. MANATRON'S ENTIRE LIABILITY AND CLIENT'S EXCLUSIVE REMEDY FOR ANY CLAIM CONCERNING THIS MASTER AGREEMENT AND SUPPLEMENTAL AGREEMENTS, AND THE PRODUCTS, SERVICES AND INTEGRATED SYSTEMS PROVIDED UNDER THIS MASTER AGREEMENT AND SUPPLEMENTAL AGREEMENTS, ARE SET FORTH IN THIS SECTION.
- 10.2. <u>DISCLAIMER</u>. CLIENT IS RESPONSIBLE FOR ASSURING AND MAINTAINING THE BACK-UP OF ALL CLIENT DATA. UNDER NO CIRCUMSTANCES WILL MANATRON BE LIABLE TO CLIENT OR ANY THIRD PARTY FOR THE LOSS OF OR DAMAGE TO CLIENT DATA.
- 10.3 INDEMNITY. MANATRON SHALL SAVE HARMLESS CLIENT FROM AND AGAINST ALL CLAIMS, LIABILITY, AND EXPENSES, INCLUDING

REASONABLE ATTORNEYS FEES FOR INJURY TO PERSONS OR DAMAGE TO TANGIBLE PROPERTY, ARISING FROM ACTIVITIES OF MANATRON, ITS EMPLOYEES, CONSULTANTS OR. PERFORMED UNDER THIS AGREEMENT THAT RESULT FROM THE NEGLIGENT ACT, ERROR, OR MANATRON OF OMISSION MANATRON'S AGENTS, SERVANTS OR EMPLOYEES. MANATRON'S OBLIGATION TO SAVE HARMLESS CLIENT IS CONDITIONED UPON CLIENT: (A) GIVING MANATRON PROMPT WRITTEN NOTICE OF ALL SUCH CLAIMS FOLLOWING RECEIPT OF SUCH CLAIMS BY CLIENT, (B) PERMITTING MANATRON TO CONTROL THE DEPENSE AND SETTLEMENT OF ALL SUCH CLAIMS, AND (C) REASONABLY COOPERATING WITH MANATRON IN THE DEFENSE AND SETTLEMENT OF ALL SUCH CLAIMS. IN NO EVENT WILL MANATRON BE LIABLE FOR ANY TUOHTIW INTO ENTERED SETTLEMENTS MANATRON'S WRITTEN CONSENT.

NOTWITHSTANDING LIMITATION. ANYTHING TO THE CONTRARY IN THIS MASTER SUPPLEMENTAL OR ANY AGREEMENT THE MAXIMUM EXTENT TO AGREEMENT, PERMITTED BY APPLICABLE LAW: MANATRON AND ANY PARTY INVOLVED IN THE CREATION, MANUFACTURE OR DISTRIBUTION THE OF PRODUCTS AND INTEGRATED SYSTEMS PERFORMANCE OF THE SERVICES, WILL NOT BE LIABLE TO CLIENT FOR ANY SPECIAL, INDIRECT, INCIDENTAL, PUNITIVE, OR CONSEQUENTIAL DAMAGES (INCLUDING LOST PROFITS) OR FOR LOST DATA SUSTAINED OR INCURRED CONNECTION WITH THIS MASTER AGREEMENT OR SUPPLEMENTAL AGREEMENTS, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGES, REGARDLESS OF THE FORM OF ACTION AND SUCH DAMAGES WHETHER OR NOT FORESEEABLE. IN ADDITION, TO THE MAXIMUM EXTENT PERMITTED BY APPLICABLE LAW, MANATRON'S LIABILITY TO CLIENT FOR DIRECT DAMAGES ARISING OUT OF OR RELATING TO THIS MASTER AGREEMENT AND SUPPLEMENTAL AGREEMENTS WILL IN NO EVENT EXCEED THE TOTAL AMOUNT ACTUALLY PAID BY CLIENT TO SUPPLEMENTAL THE UNDER MANATRON AGREEMENT WITH RESPECT TO WHICH THE DIRECT DAMAGES WERE INCURRED TIMES 2.0.

10.5 Referrals. Manatron may direct Client to third parties having products or services that may be of interest to Client for use in conjunction with the Products, Services or Integrated Systems. Notwithstanding any Manatron recommendation, referral or introduction, Client will independently investigate and test non-Manatron products and services and will have sole responsibility for determining suitability for use of non-Manatron products and services. Manatron has no liability with respect to claims relating to or Version 01-06-06

arising from use of non-Manatron products and services, including, without limitation, claims arising from failure of non-Manatron products to provide proper time and date functionality.

11. INFRINGEMENT INDEMNITY.

11.1 Indemnity. Manaron, at its own expense, will defend and indemnify Client against claims that Manaron Proprietary Software furnished under this Master Agreement or Supplemental Agreements infringe any patent or copyright or misappropriate trade secrets protected under United States law, provided Client (a) gives Manatron prompt written notice of such claims pursuant to Section 14.11, (b) permits Manaron to comirol the defense and settlement of the claims, and (c) provides all reasonable assistance to Manatron in defending and settling the claims.

11.2 Remedies. As to any Manatron Proprietary Software which is subject to a claim of infringement or misappropriation, Manatron may (a) obtain the right of continued use of the Manatron Proprietary Software for Client or (b) replace or modify the Manatron Proprietary Software to avoid the claim. If noither alternative is available on commercially reasonable terms, then, at the request of Manatron, any applicable Software license and its charges will end, Client will stop using the Manatron Proprietary Software, and Client will return to Manatron or destroy all copies of the Manatron Proprietary Software, and will certify in writing to Manatron that such return or destruction has been completed. Upon return or Manutron's receipt of certification of destruction of the Manatron Proprietary Software, Manatron will give Client a credit for the price paid to Managron, less a reasonable offset for use and obsolescence.

11.3 Exclusions. Manatron will not defend or indemnify Client if any claim of infringement or misappropriation (a) is asserted by an affiliate of Client, (b) results from Client's design or alteration of any Manatron Proprietary Software, (c) results from use of any Manatron Proprietary Software in combination with any non-Manatron product, except to the extent, if any, that such use in combination is part of an Integrated System designed and installed by Manatron for Client, or (d) relates to a non-Manatron Product alone.

11.4 Exclusive Remedies. This Section 11 states the entire liability of Manatron and Client's sole and exclusive remedies for patent or copyright infringement and trade secret misappropriation.

12. TERMINATION.

12.1 Term. The term of this Master Agreement will be for one (1) year from the Effective Date (the "Initial Term"). This Master Agreement will automatically renew for consecutive one (1) year terms thereafter ("Renewal Terms") unless either party notifies the other of its election not to renew the terms of this Master Agreement at least sixty (60) days prior to the expiration of the Initial Term or any Renewal Term. Notwithstanding the termination of this

Master Agreement upon the expiration of the Initial Term or eny Renewal Torm, this Master Agreement will remain in effect with respect to any Supplemental Agreements then in progress, but not then completed, until such Supplemental Agreements terminate or another Master Agreement is entered into by Client and Manatron.

- 12.2 <u>Defaults</u>. The following events are deemed to be defaults:
- (a) A party committing a material breach of any term of this Master Agreement or any Supplemental Agreement, if such breach has not been cured within thirty days after written notice of such breach has been given by the non-defaulting party to the defaulting party;
- (b) A party filing bankruptcy, becoming insolvent, or having its business placed in the hands of a receiver, assignee or trustee, whether by voluntary act or otherwise;
- (c) A party failing to comply in any material respect with any federal, state or local laws applicable to a party's performance under this Master Agreement or any Supplemental Agreement.
- 12.3. Termination for Default. A party may terminate this Master Agreement and the relevant Supplemental Agreement(s) before expiration of their respective term(s) for default by the other party. If default occurs, the parties will have all remedies provided in this Master Agreement and otherwise available by statute, law or equity.
- 12.4. <u>Survival</u>. Any terms of this Agreement, which by their nature extend beyond its termination, remain in effect until fulfilled, and apply to successors and permitted assigns.
- 12.5 Suspension of Performance. If any payment due to Manatron under this Master Agreement or any Supplemental Agreement is past due more than thirty days, Manatron may suspend performance under this Master Agreement and any or all Supplemental Agreements until all amounts due are current.
- 12.6 Fiscal Funding. Manatron or Client may terminate any Supplemental Agreement upon thirty (30) days written notification due to the lack of fiscal funding. Client will be responsible for payment of all labor, costs and expenses incurred by Manatron through the date of the revelpt of written notification.

12.7 Termination of Maintenance Services.

- (a) Client may terminate this Master Agreement or any Supplemental Agreement(s) at any time after the first anniversary of the Installation Date by providing at least sixty (60) days prior written notice of termination to Manatron.
- (b) Manatron may not terminate an Annual Software License and Maintenance Supplemental Agreement during the first thirty-six (36) months of paid maintenance. Thereafter, Manatron may terminate Maintenance Services provided to the Client for any Product or Integrated System upon written notice six (6) months prior to termination.

Version 01-06-06

Page 12 of 14

- (c) If Manatron determines that any alterations, attachments, or modifications not made by Manatron will interfere with the provision of Maintenance Services, then Manatron may notify Client of its intention to terminate Maintenance Services. If Client does not cure within thirty (30) days of such notice, Maintenance Services will be terminated.
- (d) Notwithstanding anything in this Master Agreement, if Client is in default of Section 6.2(g), Manatron may terminate Maintenance Services for such Manatron Software or Non-Manatron Software for which Client is in default.

13. DISPUTE RESOLUTION.

- 15.1 Disputes and Demands. The parties will attempt to resolve any claim or controversy related to or arising out of this Master Agreement or Supplemental Agreements, whether in contract or in tort ("Dispute"), on a confidential basis according to the following process, which either party may start by delivering to the other party a written notice describing the dispute and the amount involved ("Demand").
- 13.7 Negotiation and Meditation. After receipt of a Demand, authorized representatives of the parties will meet at a mutually agreed upon time and place to try to resolve the Dispute by negotiation. If the Dispute remains unresolved after this meeting, either party may start mandatory non-binding mediation under the commercial mediation rules of the American Arbitration Association ("AAA").
- 13.3 Injunctive Relief. Notwithstanding the provisions of Section 12.2(a) and this Section 13, if either party seeks injunctive relief, such relief may be sought in a court of competent jurisdiction without the requirement for prior notice and opportunity to cure under Section 12.2(a) and without complying with the negotiation and mediation provisions of this Section 13.
- 13.4 Time Limit. Meither mediation under this section nor any legal action, regardless of its form, related to or arising out of this Master Agreement or Supplemental Agreements may be brought more than four (4) years after the cause of action first accrued or within the statutory limits prescribed by law.

14. GENERAL PROVISIONS.

14.1 Entire Agreement. This Muster Agreement, the Supplemental Agreements and the attachments and exhibits thereto are the entire agreement and supersede all prior negotiations and oral agreements. Manatron has made no representations or warranties with respect to this Master Agreement or the Supplemental Agreements that are not included herein or therein. This Master Agreement and the Supplemental Agreements may not be amended or waived except in writing signed by an officer of the party to be bound thereby. If any conflict exists between the terms of this Master Agreement and any Supplemental Agreement, the terms of the Supplemental Agreement will control.

14.2 Preprinted Forms. The use of preprinted forms, such as purchase orders or acknowledgments, in connection with this Master Agreement and the Supplemental Agreements is for convenience only and all preprinted terms and conditions stated thereon are void and of no effect. The terms of this Master Agreement, including but not limited to this Section 14.2, and the terms of Supplemental Agreements cannot be amended, modified or altered by any conflicting preprinted terms, provisions or conditions contained in a preprinted form, such as purchase orders or acknowledgements. If any conflict exists between this Master Agreement or Supplemental Agreements and any terms and conditions on a purchase order, acknowledgment or other preprinted form, the terms and conditions of this Master Agreement and Supplemental Agreements will govern.

14.3 Interpretation. This Master Agreement and the Supplemental Agreements will be construed according to their fair meaning and not for or against either party. Headings are for reference purposes only and are not to be used in construing the Master Agreement or Supplemental Agreements. All words and phrases in this Master Agreement and the Supplemental Agreements are to be construed to include the singular or plural number and the masculine, feminine, or neuter gender as the context requires.

14.4 Governing Law. THIS MASTER AGREEMENT AND THE SUPPLEMENTAL AGREEMENTS WILL BE GOVERNED BY THE LAWS OF THE STATE OF TEXAS, WITHOUT REGARD TO ITS CONFLICT OF LAWS PROVISIONS, UNLESS CLIENT IS A GOVERNMENTAL SUBDIVISION OF ANOTHER STATE, IN WHICH CASE THE LAWS OF THE STATE IN WHICH CLIENT IS A GOVERNMENTAL SUBDIVISION WILL CONTROL.

this Master Agreement and the Supplemental Agreements will be interpreted to be effective and valid under applicable law, but if any provision is found to be invalid, illegal or unenforceable, then such provision or portion thereof will be modified to the extent necessary to render it legal, valid and enforceable and have the intent and economic effect as close as possible to the invalid, illegal or unenforceable provision. If it is not possible to modify the provision to render it legal, valid and enforceable, then the provision will be severed from the rest of the Master Agreement or Supplemental Agreement, as applicable, and ignored. The invalidity, illegality or unenforceability of any provision will not affect the validity, legality or enforceability of any other provision of this Agreement, which will remain valid and biriding.

14.6 Delays. Manatron is not responsible for failure to fulfill its obligations when due to causes beyond its reasonable control, including the failure of third parties to timely provide Software, Hardware, Services, materials or labor contemplated herein or in any Supplemental Agreement(s). Manatron will notify Client in writing of any such delay, and the time for Manatron's performance will be extended for a period corresponding to the delay. Manatron

and Client will determine alternative procedures to minimize project delays.

14.7 Force Majeure. "Force Majeure" means a delay encountered by a party in the performance of its obligations under this Master Agreement or Supplemental Agreements which is caused by an event beyond the reasonable control of the party, but does not include any delays in the payment of monics due by either party. Without limiting the generality of the foregoing. Force Majeure includes but is not restricted to the following types of events: acts of God or public enemy; nots of governmental or regulatory authorities (other than the Client and its governing entities); fires, floods, enidemics or serious accidents; unusually severe weather conditions; strikes, lockouts, or other labor disputes. If any event constituting Force Majeure occurs, the affected party shall notify the other party in writing, disclosing the estimated length of the delay, and the cause of the delay. If a Force Majeure occurs, the affected party will not be deemed to have violated its obligations under this Master Agreement or any Supplemental Agreements, and time for performance of any obligations of that party will be extended by a period of time necessary to overcome the effects of the Force

14.8 Compliance with Laws. Client and Manatron shall comply with all federal, state and local laws in the performance of this Master Agreement and the Supplemental Agreements, including those governing use of the Hardware, Software and Integrated Systems. Products provided under this Master Agreement may be subject to U.S. and other government export control regulations. Client shall not export or re-export any Products,

14.9 <u>Assignments</u>. Neither party may assign its interest in this Master Agreement or the Supplemental Agreements without the prior written consent of the other, such consent not to be unreasonably denied.

14.10 Independent Contractors. Client and Manatron are independent contractors and are not agents or partners of each other. Manatron's employees, agents and subcontractors will not be entitled to any privileges or benefits of Client employment. Client's employees, agents and contractors will not be entitled to any privileges or benefits of Manatron employment.

14.11 Notices. Any notice required or permitted to be given under this Agreement by one party to the other must be in writing and shall be given and deemed to have been given immediately if delivered in person to the address set forth on the Signature Page for the party to whom the notice is given, or on the fifth (5th) business day following mailing if placed in the United States Mail, postage prepaid, by registered or certified mail with return receipt requested, addressed to the party at their address set forth on the Signature Page. Each purty may change its address for notice by giving written notice of the change to the other party.

Manawon, Inc. Master Agreement

Signature P	age
-------------	-----

Agreement l	lumber:
-------------	---------

Re: Master Agreement Number: FB0002

RECORDS MANAGEMENT ANNUAL SOFTWARE LICENSE

AND MAINTENANCE SUPPLEMENTAL AGREEMENT

This Records Management Annual Software License and Maintenance Supplemental Agreement ("Supplemental Agreement") is entered into by and between Manatron, Inc. ("Manatron"), a Michigan corporation authorized to do business in the State of Texas, and Fort Bend County ("Client"), a political subdivision of the State of Texas. Manatron and Client have entered into a Records Management Master Agreement (the "Master Agreement") number F80002. This Supplemental Agreement is entered into under the terms of the Master Agreement and constitutes a "Supplemental Agreement" as defined in the Master Agreement. The terms of the Master Agreement and constitutes a "Supplemental agreement" at its Supplemental Agreement. The Master Agreement and this Supplemental Agreement constitute the entire agreement between Manatron and Client with respect to the subject matter of this Supplemental Agreement.

The following Exhibits are attached to this Supplemental Agreement and made a part hereof.

Exhibit A Pricing and Inventory

Exhibit B Description of Maintenance Services

Exhibit C Client Service Request Form

Exhibit D Change Request Form

Exhibit E Non-Manatron Sublicensed Software License Agreements

The date of this Supplemental Agreement is OCTOUR 9 2007. Client acknowledges it has read and understands the Master Agreement and this Supplemental Agreement (including all exhibits, schedules and amendments).

Agreed and Accepted:

Client

Name: Address: Fort Bend County, Texas 301 Jackson Street, Suite 719

Richmond, Texas 77469

Primary Phone:

Facsimile:

Executed by: Name: Title: 281/341-8608

281-341-86091

Robert E. Hebert County Judge_ Manatron

Managon, Inc.

510 East Milham Avenue

Portage, Michigan 49002

John R. Hansen

Director of Risk Management

This Agreement is not effective until executed by both parties.

ATTEST:

Dianne Wilson, County Clerk

Page 1 of 12

Manatrón, Inc. Annual Software License and Maintenance Agreement

1. DEFINITIONS.

Defined terms used in this Supplemental Agreement will have the same meaning given to such terms in the Master Agreement, except to the extent modified or otherwise defined herein. As used in this Supplemental Agreement, the following additional definitions apply:

- 1.1 "Anniversary Date" means each anniversary of the Installation Date.
- 1.2 "Amunal Fee" means the combined annual license, sublicense and Maintenance Services fees payable by Client to Manatron as described in Section 4.
- 1.3 "Maintenance Services" means the level of maintenance service selected (Level One, Basic Level of Service (Mandatory); Level Two, Extended Service (Optional); or, Level Three, Extended Service (Optional) in Exhibit A (Pricing and Inventory) of this Supplemental Agreement that Manatron will provide to Client under this Supplemental Agreement. The Maintenance Services are more specifically described in Exhibit B, Description of Maintenance Services.
- 1.4 "Maintenance Phase Start Date" means the date the Client signs off on the final Project Acceptance Criteria. This final sign-off begins the Maintenance Phase of the Agreement.

2. SOFTWARE LICENSES AND SUBLICENSES.

- 2.1 Manatron Proprietary Software License. The Manatron Proprietary Software licensed to Client under this Supplemental Agreement is identified in Section 2.1 (Manatron Proprietary Software) of Exhibit A (Pricing and Inventory). The Number of Licensed Users, Licensed Server and Licensed Location with respect to each item of Manatron Proprietary Software are also specified in Exhibit A. The Manatron Proprietary Software is licensed to Client on the terms and conditions set forth in the Master Agreement and this Supplemental Agreement. Client is permitted to use the Licensed Software only in connection with the Integrated System with which the Manatron Proprietary Software is provided. 2.2 Non-Manatron Sublicensed Software.
 - (a) The Non-Manatron Sublicensed Software sublicensed to Client under this Supplemental Agreement is identified in Section 2.2 (Non-Manatron Sublicensed Software) of Exhibit A (Pricing and Inventory). The Licensor, Number of Licensed Users, Licensed Server and Licensed Location with respect to each item of Non-Manatron Sublicensed Software are also specified in Exhibit A. The Non-Manatron Sublicensed Software is sublicensed to Client on the terms and conditions set forth in the Master Agreement and this Supplemental Agreement. Client is permitted to use the Non-Manatron Sublicensed Software only in connection with the Integrated System with which the

Non-Manatron Sublicensed Software is provided. Client acknowledges that this sublicense is subject to the terms of the Licensors' respective license agreements for the Non-Manatron Sublicensed Software, which are incorporated bosoin by reference, and attached hereto as Exhibit E. The Licensor(s) of the Non-Manatron Sublicensed Software are third party beneficiaries of the sublicense terms of the Master Agreement and this Supplemental Agreement to the extent permitted by applicable law.

- 2.3 Non-Manatron Other Software Not Under This Agreement. The Non-Manatron Other Software identified in Section 2.3 (Non-Manatron Other Software) of Exhibit A (Pricing and Inventory) is not included in or covered by this Supplemental Agreement. The Non-Manatron Other Software is listed in Exhibit A solely as a matter of record keeping convenience and to identify the Integrated System with which the Manatron Proprietary Software and Non-Manatron Sublicensed Software may be used. If Client has any rights with respect to the Non-Manatron Other Software, such rights would be under a separate agreement with Manatron or the Licensor of such software.
- 2.4 Hardware Not Under This Agreement. The Hardware identified in Exhibit A (Pricing and Inventory) is not included in or covered by this Supplemental Agreement. The Hardware is listed in Exhibit A solely as a record keeping convenience and to identify the Integrated System with which the Manatron Proprietary Software and Non-Manatron Sublicensed Software may be used. If Client has any rights with respect to the Hardware, such rights would be under a separate agreement with Manatron or the manufacturer or supplier of such Hardware.

3. MAINTENANCE SERVICES.

3.1 First Year's Maintenance Services at No Extra Charge. Manatron will provide to Client Maintenance Services for no extra charge for one (1) year beginning with the Installation Date on the terms and conditions set forth in Section 9 of the Master Agreement. The Maintenance Services to be provided during this one year (1-year) period are described in Section 3 of Exhibit B (Description of Maintenance Services) as Level ONE, Basic Level of Service. The Maintenance Services described in Section 3 of Exhibit B as Level ONE, Basic Level of Service constitute the complete list of services provided by Manatron during the first year, unless Client elects to pay for a higher level of Maintenance Services, which payment will be due on the Installation Date. If Client pays Manatron on or before the installation Date for a Level Two or Level Three level of service during the first year, Manatron will provide Client during such year the level of service paid for by Client. Manatron will provide one (1) week

Annual Software License and Maintenance Agreement

of on-site support upon commencement of the first year's Maintenance Services.

- 3.2 Selection of Maintenance Service Level. Pursuant to the terms and conditions set forth in the Master Agreement and this Supplemental Agreement, if the term of this Supplemental Agreement renews as provided in Section 5.1(b), beginning with the first Anniversary Date Manatron will provide Client with Maintenance Services at the level selected and paid for by Client (Level One, Level Two, or Level Three) as described in Exhibit B, upon payment of the Annual Fee. The Maintenance Services as described in Exhibit B for the level selected and paid for by Client constitute the complete list of maintenance and support to be provided by Manatron. Manatron will not be obligated to provide any Maintenance Services not described in the level of Maintenance Services selected and paid for by Client.
 - 3.3 Additional Services on Time and Materials Basis. If Manatron with Client's prior written approval, provides maintenance and support or other services requested by Client that are not covered by the level of Maintenance Services selected and paid for by Client, Client shall pay Manatron for all such maintenance, support and services on a time and materials basis, plus expenses, at Manatron's their prevailing rates, unless otherwise agreed in writing by Manatron and Client.

4. ANNUAL FEE.

- 4.1 <u>Date Payable</u>. An Annual Fee is due and payable by Client to Manatron on each Anniversary Date of this Agreement, Client must pay each invoiced Annual Fee by the due date.
- A2 Invoicing. Manatron will invoice Client for the Annual Fee at least ninety calendar days before the Anniversary Date on which the Annual Fee is due. Manatron may periodically review and adjust the amount of the Annual Fee, subject to the terms of Section 5.5 of the Master Agreement. Manatron will notify Client of any changes in the Annual Fee with the invoice.
- 43 Nonrenewal. If Client fails to timely pay the Annual Fee, this Supplemental Agreement will not automatically renew in accordance with Section 5.1(b) of this Agreement.
- 5. TERM OF AGREEMENT; TERMINATION.

5.1 <u>Term</u>

(a) The initial term of this Supplemental Agreement will commence on the date of this Supplemental Agreement, with the licenses and sublicenses granted herein becoming effective on the

date of this Supplemental Agreement. This Supplemental Agreement and the licenses and sublicenses granted herein will automatically terminate on the first Anniversary Date unless renewed as provided in Section 5.1(b).

- (b) This Supplemental Agreement will renew for a one year renewal term on each Anniversary Date if Manatron has received, on or before such Anniversary Date, payment of the invoiced Annual Fee due on such Anniversary Date. Unless this Supplemental Agreement renews at the end of a renewal term as provided in the previous sentence, this Supplemental Agreement and the licenses and sublicenses granted herein will automatically terminate upon the end of such renewal term.
- 5.2 <u>Termination</u>. This Agreement may be terminated as provided in Section 12 of the Master Agreement. Maintenance and support services under this Agreement may be terminated as provided for in Section 12.7 of the Master Agreement.
- 5.3 Effect of Termination. Upon termination or cancellation of this Supplemental Agreement, Client shall return to Manatron or destroy all Manatron Proprietary Software, Non-Manatron Sublicensed Software and Proprietary and Confidential Information in accordance with Sections 3.3(c) and 7.2 of the Master Agreement.
- 6. LIMITED WARRANTY TERMS.

THE DISCLAIMERS SET FORTH IN SECTION 9 OF THE MASTER AGREEMENT ARE INCORPORATED HEREIN BY REFERENCE.

7. Limitation of Damages.

THE LIMITATION OF DAMAGES SET PORTH IN SECTION 10 OF THE MASTER AGREEMENT ARE INCORPORATED HEREIN BY REFERENCE.

8. CHANGE REQUESTS.

Any amendment of this Supplemental Agreement must follow the Change Request procedure stated in Section 2.4 of the Master Agreement.

Manatron, Inc.

Annual Software License and Maintenance Agreement

AUDITOR'S CERTIFICATE

I hereby certify that funds in the amount of \$92,010.00 are available to pay the obligation of fort bend county within the foregoing agreement.

Roben Ed Sturdivant, County Auditor

(The rest of this page is intentionally left blank.)

Manatron, Inc. Exhibit A - Pricing and Inventory

I. ANNUAL FEE

1.1 License Fees and Basic Level ONE Service - Mandatory

Client shall pay Manatron the fees set forth below for the license of Manatron Proprietary Software, sublicense of Non-Manatron Sublicensed Software and Manatron's Level ONE Basic Level of Service:

reased Software and Manatron's Level One Basic Level of Service.	Total Price
Service Option Software License and Sublicense Fees plus	\$92,010.00
Maintenance Level ONE Basic Level of Service	1
Anthom ^{rad} Licenses (\$16,880.00)	A constitution of the second o
Anthem TM Official Public Records (also includes UCC, Assumed Names,	
Anthem Contrar Fund Management Reporting) — (105 users) Export/Archive, and Cash Management Reporting) — (105 users) Anthem Vitals for Birth and Death	
- Anthom Marriage Licenses	ľ
- Anthem Map/Plat - Advanced Backfile	un de la companya de
- Anthein TM Commissioners Court - Anthein TM Courts (for Juvenile, Criminal, Civil & Probate)	
- Anthem TM Public Access (local and internet)	
- ImagenTM Licenses (\$27,365.00)	
*Web Support (\$15,190.00) Anthematic Recording (\$12,700.00)	
Antheman Automated Indexing (\$18,375.00)	
-Anthem** Toll Road (\$800.00) -Anthem** Texas Online Support (\$700.00)	
- Lead Tools Image Viewer (105 users included)	
- Java Viewer	
Total for Level ONE Service (10/1/07 - 9/30/08)	

1.2 Extended levels (optional - select one)

Client shall pay Manatron the fees set forth below for extended levels of service selected by Client:

GII SURII DAY MANIMINA CO	1	Additional Price
Optional - Select One	Check one	Additional Price
Mandatory fees above plus Level TWO -	İ]
Evtended Service (On site 2x per year)		
Mandatory fees above plus Level THREE-	į	
Mandatory tess about page 120	1	1
Extended Service (On site 4x per year)		
The second secon		

1,3 Total Armual Fee

The Annual Fee consists of the sum of the fees in Section 1.1 and 1.2 above. Manatron will bill Client quarterly for the Annual Fee. Manatron may periodically adjust the amount of the Annual Fee, subject to the terms of Section 5.5 of the Muster Agreement.

2. LICENSED SOFTWARE INVENTORY

2.1 Manatron Proprietary Software

N. 1 Islanten on a Johnson 1	A CONTRACTOR OF THE PROPERTY O	
	Version: 6.8.x.	
Licensed Software: - Anthem Official Public Records (also includes UCC, Assumed Names,		
- Anthem Official Public Records (also includes Cot.) Published Property		
Export/Archive and Cash Management Reporting)		
- Anthem Vitals for Birth and Death		
- Anthem Marriage Licenses		ĺ
Allinent mantage	1	ı
- Anthem Map/Plat	ł	Ĺ
- Advanced Backfile		į

Manatron, Inc., Amnual Software License and Maintenance Agreement Exhibit A - Pricing and Inventory

	AN	
- Authem Commissi	mers Court	
- Anthem Courts (for	Juvenile, Criminal, Civil & Probate)	
- Anthem Toll Road		
Township	ntral Processing Unit) Anthem production ser	KEL.
Licensed Server. (C-	Cat Dand County	the first of the state of the s
Licensed Location:	Alexand 102	Kristinas karagoninga, ini maningana manggapa karagoning panggapangan bandangan pangan minadan karagon
Number of Licensed	OSCISCION	many normalistics in the first the contest of the c
Licensed Software:	· A STATE OF THE PROPERTY OF T	Version: 6.8.x
Amthorn TM Paiblic	Access (local and internet)	L
- Anthem the eRecor	ling	**************************************
- Anthem Texas	Inline	fr _e
T - margat Corver (C	entral Processing Unit) Anthem™ PA internet	on Anthem web server, Anthem
an acarding and All	hem TX Online on Anthem eRecording serv	er
Licensed Location:	Fort Rend County	
Number of Licenses	Hiters: unlimited	《大龙沙沙·沙·沙·沙·沙·沙·沙·沙·沙·沙·沙·沙·沙·沙·沙·沙·沙·沙·沙
Milliogi, or Priceuso.		
		Contracting the contracting the contraction of the
Licensed Software:		Version: 6.8,x
- AnthemTM Autom	ated Indexing	
		The second secon
Licensed Server: (C	Central Processing Unit) Anthem Al server	
Licensed Location:	Fort Bond County	
Number of License	d Users: 6 users plus 1 supervisor	
The same of the sa	**************************************	
2.2 Non-Manatron St	utilisanced Suffragre	
(a) Database Soft	ware	
***	77	
ensed Software		Version
ensed Server (Centra	processing Unit)	
ensed Location	Annual Art Was a contract and a cont	
mber of Licensed Use	**************************************	
Hibel Of The House oc.		
	1	
(h) Image Softwa	are	
3 C Garage at L	ead Tools ImageViewer	Version:
Licensed Soliwate.	ntral Processing Unit) Fort Bend County Clerk	
Lacensed Server: (Cel	HI of Territorians while the distille beinger bother.	# CARPERTON CONTRACTOR AND
Licensed Location: Fi	ort Bend County	, to the state of
Number of Licensed	JSGTST 192	***************************************
Licensed Software: Ja	***	I Version:
Licenced Coffware Is	IAS ATOMOL	A GURTOTT.

2.3 Non-Manatron Other Software

Licensed Location: Fort Bend County
Number of Licensed Users: 2 processor licenses

web server

(a) Sheint arrain software

	(a) Dirlik-Mich sortum.			
	And the second s	A		
i	1 Themson's Software Maine	Version	Quantity of Licenses	
	La Taylor Marie Control of the Contr			
	F Control of the Cont	i .	1 i	

Licensed Server. (Central Processing Unit) Fort Bend County Clerk Anthem production server and Anthem

Manatron, Inc., Amuni Software License and Maintenance Agreement Exhibit A - Pricing and Inventory

(b) Operating systems shipped with hardware

(i) Desktop operating systems

(f) Deskiop operating systems.	VIII. 114 VIII.	
	Version	Quantity of Licenses
Licensed Software Name		
The state of the s		j
The state of the s	Lamman	
The state of the s		
(ii) Server operating systems	tal establishment to be a second contract of the second se	Market and a second
Norma Norma	Version	Quantity of Licenses
Licensed Software Name		- Later Company of the Company of th
Land to the second seco		

Page 8 of 12

Chranfity		al gener proce source special general		Quantity	- Berry M	- ignorif	Saint,	w-vie	1948	pica	
	Model Number Description		r or Other Supplier:	Model Number Description							
	Medel		ware Manufacture	200							
3.1 Hardware Purchased From Manatrout.	Composint		3.2 Hardware Purchased Directly From Hardware Manufactarer or Other Supplier:		111721711117						
3.1 Hardware	Subsystem		3.2 Hardware	,	Subsystem						

Manatron, Inc., Andual Sostware License and Maintenance Agreement

Exhibit A – Pricing and Inventory 3, HARDWARE INVENTORY

Manatron, Inc. Exhibit B - Description of Maintenance Services

y & (s

Manatron, Inc. Exhibit C - Client Service Request Form

.

CLIENT SERVICE REQUEST FORM

(FAX # 800-396-4278) (HELP DESK # 800-750-4278)

ተረጓሮ ልፕ	TON:	······································	promocratically and the second	(CLIEN	T) REGISTER OF DEEDS
ትፖርት ስማርያ ነ	5 × 71	TIME:		_[A.M./P.M.]	
NAME	OF USER/DEPT, HAVING	THE PRO	BLEM:		The state of the s
TAUTANT	NUMBER:	ananananananananananananananananananan			was ground the control of the contro
CEVLES	OF USERIDER: P	RIORITY	Die ASAP, HIGH, ROUTINE	FY]):	AND THE PROPERTY OF THE PROPER
STA A TOWN	are - Check one of the followin	et.			
II. ZOTIM	His m Manha arms an succession	pro-			
a#FN.	DOCUMENT RECEPTION	Q	SCANNING	ā	ENHANGEMENT
0	INDEXING	O	CODE MAINT/FEE SETUP	. 5	
ä	REPORTS		UCC	0	
ō	MARRIAGE	u D	PUBLIC		
	SYSTEM ADM.	heelt	O) nan		
	or a manage the following	irri or+			
if Fland	ware - Check one of the following	. .	SCANNER		MONITOR
a	HP LASER PRINTER LABEL PRINTER	ŭ	PG		SERVER
<u>u</u>	RECEIPT PRINTER		CASH DRAWER		
ü	OTHER (specity)				
Section of the sectio	And the second s	educial parameter pary active or a second side of the last of the	NO 201-201-201-201-201-201-201-201-201-201-		and gift in the contract of th
A	March 1 Control of the Control of th			AND THE RESIDENCE OF THE PROPERTY OF THE PROPE	
3464-3511	ution: (i.e. who was contacted, s	ituation was	resolved in what way, sec.)	<u> </u>	and the state of t
-	and the state of t				
	-	####**********************************	· · · · · · · · · · · · · · · · · · ·	***************************************	
,		***************************************			en ege egene krytingskipp det de state de state en egene i ne blande en en egene egene krytingskept en blande e
***************************************		~*************************************		· ····································	end:
مستن بياريا _ي ه					
					acidematicidady magazine incidental particular enterior and acidematic and acidematic and acidematic acidemati
		,,, 	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	and the second s	
					nerithentic (to 15-17) - 1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-

Manatron, Inc. Exhibit D - Change Request Form

φ ... t. ... _{ξ φ} ... c.

CHANGE REQUEST FORM

Part I - Request

(Check one)	
D Initial Project Requirement U Project Scope Change D Post-Acceptance Enhancement	
Project Name:	and the second s
Product / Release / Version:	Received Date:
Request Number:	Phone:
Request Originator: Name:	Email:
Title / Role:	Citan:
Solution area / module / component:	Requirement Priority:
Requirement Severity:	KCLINI PIRCH, F. M. H. I.
Request Abstract or Title:	the state of the s
	Marini internativa continuation and the second distribution of the second d
and the same of th	
	which the same of
And included the second of the second	and the second s
The state of the s	
Charles and the second	
Request description: (Objective to be satisfied or issue	to be resolved. Please attach any relevant examples.)
Request descriptions: (Confession to on agents	A to the special section of the sect
And the second s	
And the state of t	
the control of the co	4.
templeterary is a reference and among the reference before the reference and among the reference and among the reference and and a reference a	
A CONTRACTOR OF THE PROPERTY O	Str. Str. Str. Str. Str. Str. Str. Str.
No object to the second	
Justification: (What is the significance of this request?	What are the benefits?)
Marie C. I. S. to the St. Co.	
and the state of t	
paracopping and the second programme and programme and the second progr	
Married Comments of the Commen	
- And the state of	
OPTIONAL Solution Proposal (Used to clarify issue	description)
All the state of t	
Implementation Acceptance Criteria: (What will you	use as your acceptance criteria if this change of scope is
implemented?)	
The second secon	
\$10000 All 10000	**************************************
	and the state of t
Branch Control of the	
the state of the s	

Manatron, Inc., Annual Software License and Maintenance Agreement
-Exhibit D - Change Request Form

Part 2 - Response

Response: (One of the following. Also shown as status once decision is made) Proposal created. Implementation not proposed (Cost) = The cost of implementing this request would far exceed the value to be added to the solution. Implementation not proposed (Conflict) = The implementation would conflict with current or future functions or objectives of the solution. On the solution of proposed (Comein) = The issue to be addressed, and/or the implementation for this request is outside the domain of this solution.
Description of the proposed solution: (Detailed implementation proposal attached)
Estimated cost to modify the solution: (HW/SW infrastructure, solution documentation, training materials, training, and support requirements)
Estimated schedule impact to accommodate the solution:
Manatron response approval: Signature: Phone:
Comments:
Client acceptance of change to project scope, schedule, and cost: Signature: Phone: Purchase order/invoice number for the revised project scope:
Comments:

I. PURPOSES AND SCOPE.

6 10 1

This Service Level Agreement sets forth the terms of the maintenance services available to Client from Manatron, Inc. ("Manatron"), a Michigan corporation authorized to do business in the State of Texas, Manatron will perform the maintenance services selected in and on the terms and conditions set forth in the Records Management Moster Agreement and the Records Management Annual Software License and Maintenance Supplemental Agreement between Manatron and Client (the "License Agreements") and as described herein.

2. AVAILABLE SERVICES.

Manatron offers three (3) levels of maimenance services.

- 2.1 Level ONE, Basic Lovel of Service. This is the most basic level of maintenance services available to Client under this Service Level Agreement and is further described in Section 3, Level ONE, Basic Level of Service.
- 2.2 Level TWO. Optional Extended Support. Manairon provides optional on-site maintenance services. Maintenance services under Level TWO. Optional Extended Support, include all maintenance services under Level ONE, Basic Level of Service, and the additional maintenance services described in Section 4, Level TWO, Optional Extended Support.
- 2.3 Level TFIREE Optional Extended Support. Manaron provides optional on-site maintenance services. Maintenance services under Level TFIREE Optional Extended Support, include all maintenance services under Level ONE, Basic Level of Service, and the additional maintenance services described in Section 5, Level TFIREE, Optional Extended Support.

3. LEVEL ONE - BASIC LEVEL OF SERVICE.

- 3.1 Base Mandatory Services. Level ONE, Basic Level of Service, is included in the Annual License/Maintenance/Support Fees and Annual Sublicense/Maintenance Fees payable pursuant to the License Agreements. During the term of the License Agreements, Manatron will provide Level ONE, Basic Level of Service, with respect to the Manatron Proprietary Software and Sublicensed Software.
- 3.2 Project and Support Manager, Manaton will designate a Project and Support Manager ("PSMPSM"). The PSM will maintain close contact with Client through frequent communication. The PSM will be responsible for managing delivery of the maintenance services.
- 3.3 Client Support Center. The Client Support Center ("CSC") is the primary point of Client contact

for all support. CSC consultants provide responses to support requests received from system users and system administration personnel. When initiating a support request, Client should communicate to the CSC the information in the Client Service Request ("CSR") Form.

- (a) The primary means of contacting Manatton's CSC during normal operating hours is via telephone through the toll-free client support line. Outside of narmal operating hours or if all CSC consultants are busy, the client support line will prompt callers to leave a voice mail message that will, in turn, activate a page to a CSC consultant.
- (b) A dedicated, toll-free client support fax line is available 24 hours a day, 7 days a week, as is e-mail access.
- 3.4 <u>CSC Hours of Operation</u>. Normal operating hours for the CSC are 7:00 AM to 7:00 PM Central Time, Monday through Friday, except for Manatron company holidays.

3.5 CSC Response Goals.

(a) Upon receipt of a CSR, a CSC consultant will review the information and assign a severity for urgency of response according to the following list:

Severity	Lype of Problem/Request.
1/Critical	A system-wide problem, one that
	prevents the recorder's office from
	continuing fundamental business
	processes. Some examples might be the
	system servers being down, users unable
	to record documents, unable to view
	images on the Clerk system, etc.
2/High	A problem that affects one or more
	modules of the Manatron system. A
	problem that prevents the recorder's
	office from performing an important
	function of the office's normal business
Metasser matrices and an appearance and	processes.
3/Medium	System feature or minor hardware is
	malfunctioning or inoperative, but a
	alternative procedure exists to achieve
	business needs. A problem that impacts
	individual users or workstations.
	Examples would include receipts
	requiring adjustment, usors receiving
	error messages that do not otherwise
	prevent business activities, data
	corrections, etc.
4/Low	The "Low" category includes cosmetic
	issues such as misspellings, parts of
	letters falling off the screen or report
	print outs, incorrect punctuation, etc.
	"Low" also includes problems that
	happen intermittently, for which root
	causes are being determined or which
	cannot be reproduced. This category is

Version 01-06-06

also used to characterize information requests.

(b) A CSC consultant will communicate to Client a Response based upon the severity of the problem. "Response" is defined as a communication with Client of the status of problem, malysis or potential remedies, or workarounds. The Response goals for a CSR received during normal working hours are shown in the following table:

Severity	Response Goal
1/Critical	Within I hour
2/High	Within 4 hours
3/Medium	Within 2 business days
4/1.ow	Assumption is these will be fixed in
•• •	the next Maintenance Release.

- (c) Responses to a CSR received via voice mail/pager, fax, or e-mail during other than normal operating hours may be delayed unless previous arrangements have been made for standby support resources.
 - 3.6 CSC Request Escalation,
- (a) Upon receipt of a Severity I/Critical CSR, the CSC manager will be notified to insure that appropriate Manatron resources are focused on returning the affected system to operation as soon as possible.
- (b) A severity 2/High CSR not resolved within eight (8) hours of notification to the CSC will be escalated for assistance by other subject matter experts or Manatron functional area supervisor/manager to determine next steps.
- (c) Client will be notified of the current status and projected closure target on each unresolved CSR, which will be tracked and reported until resolved.
- 3.7 Remote Diagnostics. The CSC consultant, subject matter expert, PSM, or other Client support personnel may utilize remote dial-in capability to assist with system diagnosis and/or corrective action. Client direct participation may or may not be required during remote dial-in operations. However, in either case, all use of remote dial-in capability will be coordinated with the Client in advance.
- 3.3 Supplements and Custom Programming are Excluded.
- (a) From time to time, Manatron may make available computer programs that are compatible with the Manatron Proprietary Software and that supplement the Manatron Proprietary Software. Also, third parties may make available computer programs that are

compatible with the Sublicensed Software and that supplement the Sublicensed Software. Supplements do not include programs necessary for the proper functioning of the Hardware, Software, or Integrated System. SUPPLEMENTS ARE NOT LICENSED OR LICENSE THE. SUBLICENSED UNDER AGREEMENTS AND WILL NOT BE PROVIDED WITH MAINTENANCE RELEASES. Subject to availability and compatibility, Client may license or sublicense supplements by written amendment to such License Agreements. All licenses and sublicenses of supplements will include additional charges.

- (b) Maintenance services do <u>not</u> include custom programming.
- 3.9 Manatron Proprietary Software Maintenance and Support. The terms of this section apply to maintenance of Manatron Proprietary Software.
- (a) Client's designated PSM will manage delivery of Manatron Proprietary Software maintenance releases or updates in accordance with the provisions of the applicable License Agreements and this Description of Maintenance Services.
- (b) Maintenance releases will be deployed on an "as-required" basis as determined by Manatron. Maintenance releases for Manatron interface programs und/or supplementary applications, that are not part of the main application (e.g., interfaces with mainframe programs, index or image conversion programs, export programs, etc.), will also be developed and deployed on an "as required" basis as determined by Manatron.
- (e) Manatron may include, at its sole discretion, in its maintenance releases, software modifications, and enhancements, which enhance the functionality of the software.
- (d) Release notes will be provided simultaneously with delivery of the release to Client to include all issues and corresponding resolutions contained in the maintenance release.
- (e) Client may submit recommended software application enhancements to be considered for inclusion in future software maintenance releases.
- (f) Manatron reserves the right to decline acceptance of software modifications recommended or requested by Client. Manatron also reserves the right to determine the conditions under which approved modifications will be delivered.
- (g) Manatron Proprietary Software maintenance includes the correction of material defects, maifunctions, or failures that result in the Manatron Proprietary Software failing to perform substantially according to the performance specifications provided by

Manatron when used properly under normal use and conditions.

- (i) Client shall fully inform Manatron immediately of any such defects, malfunctions or failures. [Upon receipt of such notice, Manatron will make best commercially reasonable efforts to fix or replace the Manatron Proprietary Software or provide a suitable workaround, as herein provided. Manatron will make best commercially reasonable efforts to provide the fix, replacement, or workaround as soon as is reasonably possible, taking into consideration the applicable Severity level.].
- (ii) Client shall provide Manatron with a list of output and any other data, including databases and backup systems, that Manatron reasonably may request to reproduce operating conditions similar to those present when the error occurred.
- (iii) Client shall provide Manatron and its agents access to all Client's facilities, hardware, personnel, and data, physically at the hardware site and, if requested by Manatron, through modem telephone connection, to permit Manatron to perform its maintenance services.
- (h) Manatron Proprietary Software maintenance includes any updates to the Manatron Proprietary Software developed by Manatron. Updates consist of any enhancements, corrections, modifications, and additions to the Manatron Proprietary Software. Use of updates with or in place of the Manatron Proprietary Software will be fully governed by and subject to the terms of the applicable License Agreements and this Description of Maintenance Services. Any portion of the Manatron Proprietary Software replaced by updates, and all copies thereof, will be destroyed (with certification of destruction provided to Manatron) or returned to Manatron, at Manatron's option.
- 3.10 Non-Monetron Sublicensed Software Maintenance and Support. The terms of this section apply to maintenance of Sublicensed Software. Manatron does not provide maintenance of any other non-Manatron software.
- (a) Client's designated PSM will manage delivery of Sublicensed Software maintenance releases or updates in accordance with the provisions of the applicable License Agreements and this Description of Maintenance Services.
- (b) Maintenance of Sublicensed Software will be accomplished on an "as required" basis as determined by Manatron and the software licensor.
- (c) Sublicensed Software maintenance will be provided only to the extent offered by the licensor of the Sublicensed Software. Manatron will not be responsible for any software programming with respect to the

Sublicensed Software or for software fixes or replacements except to the extent available from the licensor.

- (d) Client shall fully inform Manatron immediately of any defects, malfunctions, or failures in the Sublicensed Software. Upon receipt of such notice. Manatron will contact the ficensor and scok a fix or replacement of the Sublicensed Software. Client shall provide Manatron and the licensor with a list of output and any other data, including databases and backup systems, that Manatron reasonably may request to reproduce operating conditions similar to those present when the error occurred. Client shall provide Manatron, the Licensor, and their agents access to all Client's facilities, hardware, personnel and data, physically at the hardware site and, if requested by Manatron, through modern telephone connection, to permit Manatron and the licensor to perform the maintenance services.
- (e) If a Sublicensed Software failure occurs, Manatron will make best commercially reasonable efforts to obtain a fix, replacement, or suitable workaround of the Sublicensed Software from the Licensor as soon as is reasonably possible, taking into consideration the applicable Severity level.
- (f) Sublicensed Software maintenance includes any updates to the Sublicensed Software developed by the Licensor and that are made available to the Client. Updates consist of any enhancements, corrections, modifications, and additions to the Sublicensed Software. Use of updates with or in place of the Sublicensed Software will be fully governed by and subject to the terms of the License Agreements and this Service Level Agreement. Any portion of the Sublicensed Software replaced by updates, and all copies thereof, will be destroyed (with destruction certified to Manatron) or returned to Manatron, at Manatron's option.
- 3.11 Exclusions. Manatron will not provide maintenance or support of any hardware or non-Manatron software (unless it is Sublicensed Software or hardware purchased directly by Manatron which is still under manufacturer's warranty).
- 3.12 Other Services. Any additional support and professional services will be under a separate Services Agreement and will be priced and billed as provided for in the Services Agreement, in addition to the Annual Maintenance/Support fees.
- 4. LEVEL TWO OPTIONAL EXTENDED SUPPORT.
- 4.1 Selection of Optional Service Level. Level TWO, Optional Extended Support, is optional and will only be provided if selected by Client.

- 4.2 Basic Level of Service, Level ONE Included. All of the maintenance services and related terms stated in Section 3 and included in Level ONE, Basic Level of Service, are provided and apply in Level TWO, Optional Extended Support.
- 4.3 Semi-Annual System Tuning and Site Visitation.

 Manatron support technicians will perform the following system tuning and ramnitoring in conjunction with two semi-annual site visits.

(a) Scope of Work.

- (i) Database Maintenance. Evaluate free space, assign additional disk space (as required), review and evaluate log files, and remove unneeded (clean up) log file information.
- (ii) Hardware (Server) Maintenance. Perform system performance diagnostics; review setup and procedures for system backup; run tape drive diagnostics; clean tape drive read-write heads; test restore from backup; check uninterrupted power supply, modem/dial-in capability, heat sink and cooling fan; check all server cable connections and performance of server monitor and input devices; and test PC restore from ghost image stored on server.
- (iii) Install Software Maintenance Releases or Updates. As time and resources permit, install software releases provided pursuant to maintenance services.
- (iv) Documentation. Produce a report of the tasks accomplished and the results achieved/observed.
- (v) Other Services. Any additional support and professional services will be mutually agreed and performed under a separate Service Agreement and will be priced and billed as provided for in the such agreement, in addition to the face charged for Level TWO, Optional Extended Support.

(b) Scheduling.

- (i) Client's designated PSM will coordinate and schedule system tuning and site visits with Client.
- (ii) The initial site visit should be accomplished within six (6) months of the effective date of an agreement to provide Level TWO maintenance services. The PSM and Client will schedule subsequent semi-annual site visits jointly.
- (iii) When possible, system tuning and site visit work will be accomplished during Client's normal business hours,
 - 4.4 Semi-Annual Report of Client CSC Activities.
- (a) The CSC staff will produce semi-annual reports concerning Client's interaction with the CSC.

- (b) Reports will include: number of CSRs mitiated by Client, status of CSRs, trends in type of support requested, CSR response and problem resolution performance, and Client satisfaction.
- 5. LEVEL THREE OPTIONAL EXTENDED SUPPORT.
- 5.1 Selection of Optional Service Level. Level THREE, Optional Extended Support, is optional and will only be provided if selected by Client.
- 5.2 Basic Level of Service. Level ONE included. All of the maintenance services and related terms stated in Section 3 and included in Level ONE, Basic Level of Service, are provided and apply in Level THREE, Optional Extended Support.
- 5.3 Quarterly System Tuning and Site Visit. Manatron support technicians will perform the following system tuning and monitoring in conjunction with four (4) quarterly site visits:

(a) Scope of Work.

- (i) Database Maintenance. Evaluate free space, assign additional disk space (as required), review and evaluate log files, and remove unneeded (clean up) log file information.
- (ii) Hardware (Server) Maintenance. Perform system performance diagnostics; review samp and procedures for system backup; run tape drive diagnostics; clean tape drive read-write heads; test restore from backup; check uninterrupted power supply, modem/dial-in capability, heat sink and cooling fan; check all server cable connections and performance of server monitor and input devices; and test PC restore from ghost image (stored on server).
- (iii) Install Software Maintenance Réleases or Updates. As time and resources permit, install software releases provided under this Service Level Agreement.
- (iv) Documentation. Produce a report of the tasks accomplished and the results achieved/observed.
- (v) Other Services. Any additional support and professional services will be under a separate Services Supplemental Agreement and will be priced and billed as provided for in the Services Supplemental Agreement, in addition to the fees for Level THREE, Optional Extended Support.

(b) Scheduling.

- (i) Client's designated PSM will coordinate and schedule system taning and site visits with Client.
- (ii) The initial site visit should be accomplished within two (2) months of the effective date of an agreement to provide Level THREE

maintenance services. The PSM and Client will schedule subsequent quarterly site visits jointly.

- (iii) When possible, system ranking and site visit work will be accomplished during Client's normal business hours.
 - 5,4 Quarterly Report of Client CSC Activities.
- (a) The CSC staff will produce quarterly reports concerning Client's interaction with the CSC.
- (b) Reports will include: number of CSRs initiated by Client, status of CSRs, trends in type of support requested, CSR response and problem resolution performance, and Client satisfaction.

6. EXCLUSIONS

SECTIONS 3.8 AND 3.11 OF THIS DESCRIPTION OF MAINTENANCE SERVICES AND GERTAIN SECTIONS OF THE LICENSING AGREEMENTS LIMIT THE MAINTENANCE SERVICES TO BE PERFORMED BY MANATRON.

7. CLIENT RESPONSIBILITIES.

- 7.1 Systems Operation. Client retains responsibility for the day-to-day management of the system and software, including the backup system.
- 7.2 Specific Responsibilities. Client is responsible for its obligations under the Licensing Agreements and the following items:
- (a) Glient Contact Point ("CCP"). Client will designate, in writing, a primary and at least one (1) alternate Client Contact Point who will serve as the primary interface between Manaron's support team and Client. The responsibilities of the CCP include the following:
- (i) Provide Client contact information and inform Manatron of any changes before they occur.
- (ii) Insure basic troubleshooting and a complete analysis of system problems using internal Client resources prior to referring a problem to Manatron.
- (iii) Before submitting a support request to the CSC, gather and record the information needed to fill out a CSR.
- (iv) Contact the CSC and provide the CSR information and any amplifying data to the CSC consultant.
- (v) Coordinate Client activities required to assist the CSC in resolving the problem.
- (vi) Serve as a liaison and primary point of Client contact for the PSM.

- (vii) Complete Change Request Forms and provide them to the PSM to initiate system or software modifications.
- (viii) Insure a Purchase Order (PO) or other suitable form of Client financial obligation authorization is generated and approved prior to requesting additional support not specifically included in the maintenance service level purchased pursuant to the License Agreements.
- (b) System Access, Security, and Software Licenses.
- (i) Client will insure that appropriate primary and alternate means are available for Manatron support personnel to gain remote dial-in access to Client's system (when appropriately coordinated with Client).
- (ii) Client will maintain system passwords and will notify Manatron, prior to implementation, of any changes that may affect Manatron's ability to provide support under the License Agreements and this Description of Maintenance Services.
- (iii) Client will maintain a record of all user workstations running any portion of the licensed or sublicensed software (including any associated Internet applications). Client will provide this information to Manatron upon request and will advise Manatron of any changes in the system that affect the currency of this information.

(The rest of this page has been intentionally left blank)



Maintenance and Support Schedule and/or Note Schedule

For Enternal Use Only: 0018967 10/15/2012 MANIA7100FF

ACCT# 4308100

FORT BEND COUNTY CLERK ATTN: PURCHASING DEPT 301 JACKSON

RICHMOND TX 77469

OCT-SEPT

		QUARTERLY MAINTENANCE & SUPPORT	\$0.00	\$0.00
1.00	MQ			,
1.00	PER	THE DERIOD 10/1/12-12/31/12 FOR	\$0.00	\$0,00
1.00	CON	CONTRACT #TX110311FBC FOR	\$0.00	\$0.00
	A2-S	GRM Recorder Support	\$8,053.00	\$8,053.00
		eRecording Support	\$4,131.25	\$4,131.25
		Public Access Spt	\$4,315.50	\$4,315.50
1.00		Wital Records Spt	\$600.00	\$600,00
1.00		alindexing sys spt	\$4,220.00	\$4,220.00
		A2 Redaction Spt *INCLUDED*	\$0.00	\$0.00
	A2-REDACTION-S	•	•	\$1,903.25
1.00	nethinutes-s	NetMinutes SW Spt	\$1,903.25	
1 80	AGENDA LINK-S	AgendaQuick Link SW Spt *INCLUDED*	\$0.00	\$0.00
1.00		Verity Lic SW Spt *INCLUDED*	\$0.00	\$0.00
1.00		ViewOne Sys Spt (2)	\$243.00	\$243.00
	TO STATE OF THE PARTY OF THE PA	Print Accelerator Spt (2)	\$137.75	\$137.75
1.00		Lead Tools SW Spt (55) *INCLUDED*	\$0.00	\$0.00
1.00		Pix Tools Software Spt (26) *INCLUDED*	\$0.00	\$0.00
1.00	PIXTOOL-S		• *	•
1.00	MARRIAGE LIC-S	Marriage Lic SW Spt	\$800.00	\$600.00
	SLA-4	Service Level 4 Support	\$1,200.00	\$1,200.00

Suprocas	\$25,603.75
Tax	\$0.00
Total	\$25,603.75

Exhibit B

STATEMENT OF WORK TX020416FBC

FORT BEND COUNTY, TEXAS

HIPA PLUS HOSTING RENEWAL

Version 1.1

February 10, 2016

Prepared by: John Rickerby

Manatron, Inc. - A Thomson Reuters Business



Document submitted by Manatron, Inc. - A Thomson Reuters Business

Thomson Reuters Tax & Accounting, Government

Records Management Division

1807 Braker Lane, Suite 400

Austin, Texas 78758

Tel.: (866) 917-4354

Fax: (512) 833-8343

Thomson Reuters Tax & Accounting, Government

Corporate Headquarters

510 East Milham Avenue

Portage, Michigan 49002

Tel.: (866) 567-2900

Fax: (269) 567-2930

@ 2016 Thomson Reuters. All rights reserved.

Republication or redistribution of Thomson Reuters content, including by framing or similar means, is prohibited without the prior written consent of Thomson Reuters. "Thomson Reuters" and the Thomson Reuters logo are registered trademarks and trademarks of Thomson Reuters and its affiliated companies.



PROJECT SUMMARY

Project:

Fort Bend County, Texas HIPA Plus hosting services for transactional replication of production Aumentum Recorder and AgendaQuick servers, hosting of Aumentum Recorder and AgendaQuick Internet Public Access with eCommerce,

and eRecording submissions.

Project Site:

Fort Bend County 301 Jackson Street Richmond, Texas 77469

County Contact:

Laura Richard, County Clerk

(281) 341-8686

Term of Renewal:

January 1, 2016 - December 31, 2018

Scope:

This project is a renewal for the replication of the current Aumentum Recorder and AgendaQuick software and production versions of Aumentum Public Access, AgendaQuick Public Access, and eRecording.

Technology:

The County will provide all local production hardware and infrastructure. TRTA Gov will provide all data center replication and web server hardware and

infrastructure.

Software -

Any software deliverables from TRTA Gov are listed in this Statement of Work. Deliverables outside the current functionality of the standard Aumentum Recorder suite and/or as installed currently in the County's production environment may be subject to additional development and configuration, which may be accompanied by additional charges. Any changes or additions to these deliverables will need to adhere to TRTA Gov's change management process.

This Statement of Work outlines deliverables, assumptions, and a high-level scope of work mutually agreed upon by Manatron, Inc. - a Thomson Reuters Business ("Thomson Reuters Tax & Accounting, Government" or "TRTA Gov") and Fort Bend County, Texas (the "County").



PROJECT OVERVIEW

This Statement of Work (SOW) defines the areas or scope of work for this project and identifies TRTA Gov and County project responsibilities. It also describes the project management processes for performing this work, the criteria for determining that the work is successfully completed, and the methods to be used to control scope, quality, and costs. Mitigation is defined for any significant project risks that have been identified.

SOFTWARE

TRTA Gov will provide database-to-database replication of the data and images to the hosted data center. This will include replication of the current Aumentum Recorder suite of application software and the Microsoft SQL Server 2012 database and replication of AgendaQuick. The County's Internet Aumentum Recorder Public Access with eCommerce module, AgendaQuick Internet Public Access and eRecording module will also be directly hosted from the data center and will include real-time communication to the local Aumentum Recorder and AgendaQuick production servers located in the County.

16217	Description	Number of Users	Comments
Aumentum Recorder Application Software	No changes to current Aumentum Recorder functionality	N/A	Replicated to data center
AgendaQuick Application	No changes to current AgendaQuick functionality	N/A	Replicated to data center
Software Cold Fusion, Version 10	Required for AgendaQuick	N/A	License required for data center and production system
AgendaQuick	Public Access for Internet-based researchers	Site license (unlimited users)	Hosted at data center
Public Access Aumentum Recorder Public Access with	Public Access for Internet-based researchers	Site license (unlimited users)	Hosted at data center
eCommerce Aumentum Recorder	Receive electronic filing of land records documents	County-approved submitters	Hosted at data center
eRecording Microsoft SQL Server 2012	Database software, including transactional replication functionality	N/A	Data center license provided by TRTA Gov
Database Software SSL Certificate	128 SSL Certificate for eCommerce	N/A	SSL certificate provided by County



RECOMMENDED HARDWARE

TRTA Gov will provide all data center server hardware, software and infrastructure required for replicating the County's Aumentum Recorder system, AgendaQuick, and hosting internet Public Access and eRecording. The County will be responsible for local production server hardware, software, infrastructure for Aumentum Recorder and AgendaQuick. The County will provide the network and bandwidth.

ltem	Description	Quantity	Comments
Data center servers and infrastructure	Aumentum Recorder application/ database/image replication server and AgendaQuick server and production web server for Internet Public Access and eRecording	1	TRTA Gov will provide
Local production server and infrastructure	Aumentum product server, AgendaQuick server and back- up web server for internet Public Access and eRecording	1	County will provide

PROJECT IMPLEMENTATION

PURPOSE STATEMENT

The purpose of this renewal is to provide full off-site transactional replication of the County's Aumentum Recorder and AgendaQuick software, indexes, and images at TRTA Gov's data center located in the Time Warner facility. In Indianapolis, Indiana. The data center will provide layered redundancy to the County's local production system and fallover in the event of an interruption in access to the production system or a full disaster.

In addition to Aumentum Recorder replication, the data center will host County's Internet Public Assess and eRecording modules to manage Internet researcher inquiries of County indexes and images stored in Aumentum Recorder and the receiving of eRecording submissions for transmission to the County production systems. This configuration will ensure high availability for these Internet-based functions within consistently managed response times, thus insulating the local production configuration and ensuring consistent access even in the event of a disaster at the County's site.

ASSUMPTIONS AND CONSTRAINTS

- A 36-month commitment is required for service for TRTA Gov applications. If the County decides to terminate the service at any point prior to the end of the 36-month commitment, the County will be responsible for the fees for the TRTA Gov applications for the balance of the commitment.
- TRTA Gov will replicate the existing indexes, images, Aumentum Recorder software, and Microsoft SQL Server database to the remote data center configuration. The County's current Aumentum Recorder release level and configuration will not be changed.



- TRTA Gov will replicate the existing AgendaQuick data, images, and software to the remote data center and configure for disaster recovery. Public users will access AgendaQuick documents via the remote data center just like Aumentum public users. The County's current AgendaQuick release level and configuration will not be changed. Version 10 of the production and data center Cold Fusion licenses will continue to be used. If the County chooses to upgrade to a newer version, an addendum to this SOW will be required, and the County will be charged an additional fee.
- Public Access for Aumentum Recorder and AgendaQuick will allow researchers to view documents using the HIPA Plus replication server. Requests to purchase Aumentum Recorder document copies will be redirected to the County/production server to complete the transaction. For eRecording, documents will be received from submitters in the hosted application and will be pushed to the production server for normal processing as batches are accessed by users. All database changes occur on the production server. Replication is from the production server to the hosted server. In the event of a disaster event impacting the County production server, the hosted replication server can be enabled to allow ecommerce Public Access purchases, eRecording processing, and any other standard County Clerk business transaction. When the production server is back online, TRTA Gov will coordinate through County iT to update the production server.
- The County will be provided with 2TB of data storage, which should be sufficient based on current
 estimates. If usage exceeds this estimate and additional storage has to be added, the County will be
 notified and asked to approve the cost of adding additional storage for the balance of the contract.
 Space will be monitored as needed to ensure there is no limitation in functionality.
- The term "real-time replication" can take a few minutes to complete as network/server traffic will affect
 the timing of the replication between sites.
- Data replication can be impacted by the telecommunication provided between the County and the data center.
- The County is responsible for providing broadband communication access and any necessary firewall from the production site to an internet service provider in the case of a LAN-to-LAN VPN tunnel.
- TRTA Gov will provide the County with a one-time delivery of images and database stored in the
 hosted facility at no charge to the County. This will include both database and images for the TRTA
 Gov AgendaQuickapplications. If the County requests that TRTA Gov Technical Services restore the
 delivered backup data and images on the local server, then a separate billable work order would be
 created based on the time and materials required to support the event.
- Data and images housed in the Aumentum Recorder system are of a non-proprietary nature. All data
 will be housed in a standard SQL database structure, and images will continue to be stored as Group
 4 tiff. All data and images hosted in the data center are the property of the Fort Bend County Clerk.
- The County will be responsible for providing all Aumentum Recorder production system hardware, infrastructure and environmental space, power requirements, and access by TRTA Gov for installation testing to the data center and operation.
- The County will provide the SSL Certificate for use with eCommerce and eRecording.
- High-speed remote access to application and database server(s) must be provided to the TRTA Gov
 project team members at agreed upon times for set-up, configuration and testing. The date and times
 of the required access will be scheduled between TRTA Gov and the County in advance by the
 respective project managers.
- The County will be responsible for testing and reviewing the data on the remote server and providing approval for cut-over production.
- All documentation provided by TRTA Gov is provided "as-is."
- TRTA Gov recommends running 100mbs Ethernet connections to the desktop.



Future technology refreshes for the County have not been included in the scope of this SOW.
 Technology refreshes include future peripheral hardware upgrades or replacement of equipment.

DEFINITION OF SEVERITY LEVELS

TRTA Gov shall respond to any errors reported by the County based on the severity level assigned to such error. Severity levels and TRTA Gov's responses are detailed in the Records Management Master Agreement dated October 9, 2007 between the County and TRTA Gov.

IMPLEMENTATION DELIVERABLES

Hardware and Software Deliverables

Any hardware and software deliverables are listed at the beginning of this SOW.

Remote Data Center Deliverables

Remote Data Center deliverables are described in "Attachment 2 - Managed Services Hosting Agreement" located at the end of this document.

All remote hardware and software and data center infrastructure are provided within the proposed HIPA Plus hosting services described in this SOW. The County will be provided with 2 TB of data storage which should be sufficient based on current estimates. TRTA Gov will be responsible for system set-up, configuration, and establishing communication with the County's network.

COUNTY RESPONSIBILITIES

Access

The County will provide a secure connection for remote access to facilitate TRTA Gov's support of the system. This can be accomplished via VPN connection for TRTA Gov personnel on a 7-day/24-hour basis during the installation or as otherwise arranged by both parties.

LAN

The County shall provide a Local Area Network (LAN) with Internet Protocol (IP) switched 100Mb Ethernet with wiring to the Network Interface Card (NIC).

Notwork Administration and Operating System Security

The County shall provide trained personnel to administer its network and manage its domain, including security access.



PRICES

The following outlines the HIPA Plus hosting services proposed.

TRTA GOV HIPA PLUS HOSTING SERVICES	TOTAL PRICE
HIPA Plus transactional data replication of Aumentum Recorder and AgendaQuick production server to the Indianapolis data center HIPA Plus Hosted Internet Public Access with eCommerce and eRecording County to provide SSL certificate for eCommerce and eRecording TB of storage Includes all Aumentum Recorder and AgendaQuick server and web server data storage hardware located at the data center, Microsoft SQL Server database software, and all supporting infrastructure to ensure high availability. Recurring TRTA Gov applications monthly fee for 36 months (January 1, 2016 – December 31, 2018): \$3,632.00 per month, or \$43,584.00 per year	\$130,752.00



ATTACHMENT 1 - AGREEMENT TO STATEMENT OF WORK

STATEMENT OF WORK AGREED AND ACCEPTED

We, the undersigned, accept this document as a stable work product to be used in the delivery of the services described herein. Any deviation from this Statement of Work is to be handled through TRTA Gov's change management process.

This Statement of Work will confirm all requests for services as outlined and at the price indicated. This SOW will be an addendum to the "Records Management Master Agreement" dated October 9, 2007 between the County and TRTA Gov. All the terms and conditions of that agreement will pertain.

BILLING FOR HIPA PLUS

Monthly billing of \$3,632.00 for HIPA Plus will begin as of January 1, 2016.

HIPA Plus hosting fees are \$3,632.00 per month for 36 months (January 1, 2016 – December 31, 2018). The total fee for the full 36 months is \$130,752.00.

SIGNATURES

Fort Begd County	TRTA Gov
301 Jackson Street	1807 Braker Lane, Suite 400
301 January 77/60 1	Austin, TX 78758
Richmond, TX 77469	BY:
BY JANOUA LLEVER	or a a a
	PRINTED NAME:
PRINTED NAME:	- I was a second
Robbet HODEVT	Brian Wilson
TITLE	TITLE:
TILLE.	
I County Judge	Vice President, Finance
DATE: J	DATE:
May 10, 2016	May 3, 2016

This agreement is not effective until executed by both parties.



ATTACHMENT 2 - MANAGED SERVICES HOSTING AGREEMENT

An incident is defined as any time the system is inaccessible to the public. The County will own the telecommunication (VPN) line that provides the data to the data center and will be first-line support; however, there may be an issue that is out of TRTA Gov's control that will temporarily affect the database replication. TRTA Gov will initiate a support call, monitor progress, and monitor the telecommunication line. Database replication is not included in the credit percentage section of this SOW. The telecommunication within the data center that provides availability to the records/website once on the server is included in the credit percentages and is owned by the data center.

The SOW applies only when the County is agreeing to a term commitment of thirty-six (36) months for the TRTA Gov applications. If the County decides to terminate the service, the County will be responsible for data center charges through the TRTA Gov application's thirty-six-month commitment. TRTA Gov reserves the right to amend the SOW from time to time effective upon notice to County, provided that in the event of any amendment resulting in a material reduction of the service levels, the change request sets forth County's sole remedies for any claim relating to the services, including any failure to meet any guarantee set forth in the change request. The Data Center records and data shall be the basis for all change request calculations and determinations. Notwithstanding anything to the contrary, the maximum amount of credit in a month shall not exceed 10% of the monthly recurring fee.

Without limiting the foregoing, County agrees that neither it nor its employees, agents, contractors, or representatives shall attempt in any way to circumvent or otherwise interfere with any security precautions, procedural controls, Acceptable Use Policy, Change Management, or other Data Center policies relating to the Data Center service offering. Any such actions may cause a disruption in service. Any disruption of services which results in the violation of these provisions shall be excluded from the infrastructure availability, and the County will have no right to any service credit or other remedy under a change request or otherwise with respect to such disruption. The County will pay TRTA Gov at the current published rates for reasonable remedial services resulting from the County's actions.

TRTA Gov will provide the County with written notice before performing scheduled maintenance and as much notice as possible when performing emergency change controls. The Data Center designs its services around an N+1-redundancy philosophy that is used to minimize outages during planned change controls. The Data Center will use commercially reasonable efforts to minimize the impact of any change control and, when possible, schedule the change control to eliminate or minimize impact to the County's service. However, the Data Center reserves the right to proceed with any change control if it is defermined by the Data Center, in its sole discretion, that a change control is necessary to maintain the overall integrity of the services and/or the change control will not cause harm to the County.

INFRASTRUCTURE AVAILABILITY

Infrastructure shall consist of the following components that are provided as part of the Data Center services:

- Co-location space reserved for use by the County in the Data Center, including quarter, half, or full cabinets and cage space;
- Power provided to County's co-location space; and
- Internet access provided to the County from the Data Center Internet Backbone.

In the event that TRTA Gov falls to meet its stated infrastructure Availability in any given month during the service term. County shall be entitled to receive a credit (as the sole and exclusive remedy hereunder) by providing TRTA Gov with a written request for a service credit within sixty (60) days after receipt of an invoice for the period in which the infrastructure was unavailable. The amount of the service credit shall be determined by the percentage in accordance with the table shown below; the credit percentage will be off the monthly recurring fee.



INCIDENT MANAGEMENT AVAILABILITY MATRIX

Infrastructure Availability Percentage	Credit Percentage
100%	0%
<99.99% > 99.5%	1%
<99.5% > 99.0%	2%
<99.0% > 98.5%	3.5%
<98.5% > 95.0%	5%
<95.0%	10%



Complete Nos. 1 - 4 and 6 if there are interested parties. Complete Nos. 1, 2, 3, 5, and 6 if there are no interested parties. Name of business entity filling form, and the city, state and country of the business entity's of business. Manatron, Inc A Thomson Reuters Business Portage, MI United States Name of governmental entity or state agency that is a party to the contract for which the febeing filed. Fort Bend County Provide the identification number used by the governmental entity or state agency to track description of the services, goods or other property to be provided under the contract.	Date Filed: 05/03/2016 Date Acknowledged:
Name of business entity filing form, and the city, state and country of the business entity's of business. Manatron, Inc A Thomson Reuters Business. Portage, Mi United States Name of governmental entity or state agency that is a party to the contract for which the fobeing filed. Fort Bend County Provide the identification number used by the governmental entity or state agency to trace description of the services, goods or other property to be provided under the contract.	CERTIFICATION OF FILING s place Certificate Number: 2016-48652 Data Filed: 05/03/2016 Date Acknowledged:
Name of business entity filing form, and the city, state and country of the business entity's of business. Manatron, Inc A Thomson Reuters Business. Portage, Mi United States. Name of governmental entity or state agency that is a party to the contract for which the febeing filed. Fort Bend County. Provide the identification number used by the governmental entity or state agency to trace description of the services, goods or other property to be provided under the contract.	Date Filed: 05/03/2016 Date Acknowledged:
Name of governmental entity or state agency that is a party to the contract for which the to being filed. Fort Bend County Provide the identification number used by the governmental entity or state agency to trace description of the services, goods or other property to be provided under the contract.	Date Acknowledged:
description of the services, goods or exist property to be provided and	k or identify the contract, and provide a
HIPA Plus Hosting Renewal	
	Nature of interest (check applica
Name of Interested Party City, State, Country (place of busi	Controlling Intermedian
5 Check only if there is NO Interested Party.	
MARY AMMAR Notes Public, State of Michigan County of Kalarrazoo 28y Commission Dolles Aug. 21, 2021 Acting in the County of Labora 220	erjury, that the above disclosure is true and correct
AFFIX NOTARY STAMP / SEAL. ABOVE Sworn to and subscribed before me, by the said Brian Wilson 20_16 to certify which, witness my hand and seal of office.	is the 3rd day of May
Mary Armhar, Set. Signature of officer administering path Printed name of officer administering path	

Records Management	150358	24	9415
Health and Human			
Services	150951	24	CARE COORDINATION
Health and Human			
Services	150951	24	EXPANDED HOURS
Health and Human			
Services	150951	24	SBIRT
Health and Human		~ <i>i</i>	COMMUNITY PARAMEDIC
Services	150951	24	HP 410A-BLACK-ORIGINAL-LASERJET
Jones Creek Ranch Park	151211	24	The state of the s
Jones Creek Ranch Park	151211	24	HP-410A-CYAN-ORIGINAL-LASERJET-T
Jones Creek Ranch Park	151211	24	HP 410A-YELLOW-ORIGINAL-LASERJET
Jones Creek Ranch Park	151211	24	HP 410A-MAGENTA-ORIGINAL-LASERIE
County Treasurer	151406	24	INSTALLATION CHARGE
Clinical Health Services	151668	24	WEST HOUSTON MEDICAL
Clinical Health Services	151669	24	OAKBEND MEDICAL
Juvenile		seper asarah rahi (1901-19	
Probation/Detention	151824	24	LICENSE RENEWAL
Animal Services	151881	24	SNAP
······································	152150	24	SHARP TONER CARTRIDGE AR-621NTA
County Treasurer Health Dept-Miss City			500
Annex	152380	24	SHIPPING & HANDLING
Indigent Health Care	152625	24	PCP SERVICES OBMG
	152798	24	10
Emergency Management	152798	24	1010
Emergency Management	152798	24	2010
Emergency Management	***************************************		3010
Emergency Management	152798	24	400000000000000000000000000000000000000
Emergency Management	152798	24	4010
Health and Human	152000	24	COLONOSCOPY SERVICES
Services	152800	24	CONDITION
Health and Human	152839	24	GLOBAL FEE
Services	125023	4-4	VIEW DE 18 1 In the
Health and Human	152841	24	GLOBAL FEE
Services	2.46,074		
Health and Human	152842	24	COLONOSCOPY SERVICES
Services	152987		
Clinical Health Services	*******		
Juvenile Probation/Detention	153067	24	TREAT BAGS
Juvenile Probation/Detention	153067	24	TRASH CANS
- MCACOMPTON TO THE PARTY OF TH	153446		
Parks Department	153448		
Animal Services Emergency Management	153656		

Emergency Management	153656	24	SPLT
Emergency Management	153656	24	SPLT
County Treasurer	153761	24	GHG ANNUAL RENEWAL 1/1-12/31/18
Fire Marshal	153989	24	SHIPPING
Extension Service	154032	24	INSTALLATION OF RACK & PINION
Fairgrounds	154087	24	ENVIRONMENTAL TESTING BLDG C
Auditor	154096	24	ANNUAL SUBSCRIPTION
Fairgrounds	154157	24	MIXED SOIL BY YARD
Fairgrounds	154157	24	HARDWOOD MULCH BY YARD
Fairgrounds	154157	24	25# COLOR STAR
Fairgrounds	154157	24	FLAT PANSIES
Social Services	154158	24	12376
Social Services	154158	24	12383
Social Services	154158	24	12384
Social Services	154158	24	12386
Social Services	154158	24	12387
Social Services	154165	24	576096
County Treasurer	154171	24	TX LOCAL GOVERNMENT CODE 2018
Emergency Management	154186	24	SATELLITE SERVICE RENEWAL
320-357-X	154196	24	TX SOCIAL SECURITY PROGRAM FEE
County Treasurer	154196	24	TX SOCIAL SECURITY PROGRAM FEE
County Treasurer Juvenile		1	100000000000000000000000000000000000000
Probation/Detention	154197	24	242.23
Juvenile Probation/Detention	154197	24	SSGY
Juvenile Juvenile			
Probation/Detention	154197	24	SSGY
Juvenile			
Probation/Detention	154199	24	SKU 4790739
Parks Department	154208	24	12694
Fire Marshal	154211	24	SNT-NP20X50W15N
Fire Marshal	154211	24	KOC-HPKE25R25-45
Fire Marshal	154211	24	TFT-AYNJ-NF
Fire Marshal	154226	24	IPAD CASE
Health Dept-Miss City			and the state of t
Annex	154273		
Commissioner Pct. 4	154291	***************************************	
Commissioner Pct. 4	154291		
Commissioner Pct. 4	154291	. 24	
Commissioner Pct. 4	154291	1 24	MX-23NTYA
Juvenile		_	024
Probation/Detention	15433		
Juvenile	15433	5 24	PJ4

Probation/Detention			
Juvenile			
Probation/Detention	154335	24	23789
Juvenile Field Services	154348	24	TR506
Juvenile Field Services	154348	24	TR506
Juvenile Field Services	154348	24	TR506
Emergency Management	154366	24	DJI-Z3
Fire Marshal	154373	24	ULTIMA FIRE HOSE
Fire Marshal	154373	24	YRCS HOSE
Fire Marshal	154373	24	ULTIMA FIRE HOSE 2"
Fire Marshal	154373	24	FREIGHT
Fire Marshal	154376	24	TFT-ME1-VPGI
Fire Marshal	154377	24	80035001-PROMO
Fire Marshal	154377	24	HALO-HD
Fire Marshal	154377	24	FREIFHT
Clinical Health Services	154388	24	1385335
Clinical Health Services	154388	24	659650
Clinical Health Services	154388	24	911900
Clinical Health Services	154388	24	142651
Clinical Health Services	154388	24	271501
Clinical Health Services	154388	24	141971
Fire Marshal	154394	24	CLASS A UNIFORM PACKAGE W/ SHOES
Fire Marshal	154394	24	SHIPPING
Parks Department	154412	24	0-RH500-74282
Parks Department	154412	24	0-RH500-75057
Parks Department	154412	24	0-RH500-74654
Parks Department	154412	24	0-RG-7192
Parks Department	154412	24	SHIPPING
Parks Department	154416	24	CLAY
Parks Department	154416	24	RED
Parks Department	154420	24	FISH
Auditor	154430	24	DRUM KIT
Emergency Management	154432	24	RAPTOR LICENSE RENEWAL
Jones Creek Ranch Park	154453	24	10586
Jones Creek Ranch Park	154453	24	11572

Agreement	Number	
-----------	--------	--

RECORDS MANAGEMENT MASTER AGREEMENT

Signature Page

WHEREAS, Ft. Bend County, Texas ("Client"), a political subdivision of the State of Texas, and Hart InterCivic, Inc. ("Hart") executed the "Fort Bend County Sublicense Agreement" on December 22, 1993; and

WHEREAS, Manatron, Inc. ("Manatron"), a Michigan corporation authorized to do business in the State of Texas, acquired Hart's Records Management Solutions Business, including the "Fort Bend County Sublicense Agreement" on September 1, 2007;

WHEREAS, Client and Manatron wish for this Records Management Master Agreement to supercede the "Fort Bend County Sublicense Agreement."

NOW, THEREFORE, Client and Manatron desire to execute this Records Management Master Agreement

This Records Management Master Agreement ("Master Agreement") is entered into by and between Manatron and Client. This Master Agreement sets forth the general terms under which Client will purchase, license or sublicense products and services from Manatron, Manatron and Client will from time to time enter into one or more Integrated Systems Installation, Annual Software License and Maintenance and/or Professional Services Supplemental Agreement(s) ("Supplemental Agreements") which will refer to this Master Agreement and describe the specific products, services, pricing, and additional terms under which Client will purchase, license or sublicense products and services from Manatron. The combination of this Master Agreement and the Supplemental Agreements constitute the entire agreement between Manatron and Client.

The Effective Date of this Master Agreement is October 9

Client acknowledges it has read and understands this Master Agreement (including all Supplemental Agreements, schedules and amendments').

Agreed and Accepted:

Client

Fort Bend County, Texas Name: 301 Jackson Street, Suite 719 Address:

Richmond, Texas 77469

281-341-8608 Telephone:

281-341-8609 Facsimile:

Executed by: Robert E. Hebert Name:

County Judge Title:

Manatron

Manatron, Inc.

510 East Milham Avenue Portage, Michigan 49002

John R. Hansen

Director of Risk Management

This Agreement is not effective until executed by both parties.

Dianne Wilson, County Clerk

Page 1 of 14

Version 01-06-06

ATTESET

1. DEFINITIONS.

- 1.1 "Client" has the meaning set forth in the signature page.
- 1.2 "Description of Maintenance Services" means a document, as amended from time to time, prepared by Manatron describing the scope and terms of Maintenance Services available from Manatron under Annual Software License and Maintenance Supplemental Agreements.
- 1.3 "Effective Date" has the meaning set forth in the signature page.
- 1.4 "Hardware" means the hardware identified in one or more Integrated Systems Installation Supplemental Agreement(s).
- 1.5 "Manatron" means Manatron, Inc., a Michigan corporation authorized to do business in the State of Texas.
- 1.6 "Manatron Proprietary Software" means the executable version of computer programs and computer code owned by Manatron which are licensed to Client pursuant to one or more Annual Software License and Maintenance Supplemental Agreement(s), and all updates, upgrades, versions, new releases, derivatives, revisions, corrections, improvements, rewrites, bug fixes, enhancements and other modifications, including any custom modifications, to such computer programs and code which are provided to Client, and all copies of the foregoing. Manatron Proprietary Software also includes all documentation provided by Manatron to Client with respect to these computer programs and code, excluding maintenance diagnostics, and the source code version of the programs and code when provided pursuant to a Supplemental Agreement, and all copies of the foregoing. Manatron Proprietary Software licensed to Client is identified as "Licensed Manatron Proprietary Software" in Section 2.1 (Licensed Manatron Proprietary Software) of Exhibit A (Pricing and Inventory) of one or more Annual Maintenance Supplemental License and Software Agreement(s).
 - 1.7 "Installation Date" means (a) the date Manatron completes installation (as determined by Manatron), or (b) if the relevant Supplemental Agreement specifically designates such date, the Installation Date as specified in such Supplemental Agreement, or, (c) if Hardware or Software is to be installed by Client, the tenth calendar day following receipt by Client.
 - 1.8 "Integrated System" means Hardware and/or Software that is integrated and installed by Manatron for Client pursuant to one or more Integrated System Installation Supplemental Agreement(s).
 - 1.9 "Licensed Location" means the location where an item of Software is licensed for Use as designated in an Annual Software License and Maintenance Supplemental Agreement.
 - 1.10 "Licensed Server" means the Hardware on which an item of Software is licensed for Use as designated in an Version 01-06-06 Page 2 of 14

- Annual Software License and Maintenance Supplemental Agreement.
- 1.11 "Licensor" means the licensor(s), respectively, of the Non-Manatron Software, as listed in Section 2.2 (Non-Manatron Sublicensed Software) and 2.3 (Non-Manatron Other Software) of Exhibit A (Pricing and Inventory) of one or more Annual Software License and Maintenance Supplemental Agreement(s).
- 1.12 "Maintenance Services" means the level of maintenance service (Level One, Basic Level of Service (Mandatory); Level Two, Extended Service (Optional); or, Level Three, Extended Service (Optional) in Exhibit A (Prioing and Inventory) of an Annual Software License and Maintenance Supplemental Agreement selected by Client, which Manatron will provide to Client under such Supplemental Agreement. The Maintenance Services are more specifically described in the Description of Maintenance Services attached to such Supplemental Agreement.
- 1.13 "Master Agreement" has the meaning set forth in the signature page.
- 1.14 "Non-Manatron Software" means the executable version of computer programs owned by third parties that are provided by Manatron to Client pursuant to sublicense(s) under one or more Annual Software License and Maintenance Supplemental Agreement(s) or license(s) directly from the third party Licensor, and all updates, ripgrades, versions, new releases, derivatives, revisions, corrections, improvements, rewrites, bug fixes, enhancements and other modifications to such computer programs which are provided to Client, and all copies of the foregoing. Non-Manatron Software also includes all documentation provided to Client with respect to these computer programs. Non-Manatron Software provided to Client is identified as "Non-Manatron Sublicensed Software" or "Non-Manatron Other Software" in Sections 2.2 and 2.3, respectively, of Exhibit A (Pricing and Inventory) of one or more Annual Software License and Maintenance Supplemental Agreement(s).
- 1.15 "Non-Manatron Other Software" means Non-Manatron Software that is licensed by a third party Licensor directly to Client. Non-Manatron Other Software is identified in Section 2.3 (Non-Manatron Other Software) of Exhibit A (Pricing and Inventory) of one or more Annual Software License and Maintenance Supplemental Agreement(s).
- 1.16 "Non-Manatron Sublicensed Software" means Non-Manatron Software that is sublicensed by Manatron to Client pursuant to sublicense(s) under one or more Annual Software License and Maintenance Supplemental Agreement(s). Non-Manatron Sublicensed Software is identified in Section 2.2 (Non-Manatron Sublicensed Software) of Exhibit A (Pricing and Inventory) of one or more Annual Software License and Maintenance Supplemental Agreement(s).

1.17 "Number of Licensed Users" means the number of users licensed to use an item of Manaron Proprietary Software or Non-Manatron Sublicensed Software as set forth in Exhibit A to the applicable Annual Software License and Maintenance Supplemental Agreement.

1 1

- 1.18 "Products" means the Hardware, Software, and all other documentation provided by Manatron to Client under this Master Agreement and any Supplemental Agreements.
- 1.19 "Proprietary and Confidential Information" means Software, diagnostics, documentation (including manuals), Hardware and Software configuration, Integrated Systems design and configuration, training materials, user guides, trade secrets, source code and related documentation, and any other information confidential to Manatron or its Proprietary and Confidential suppliers or Licensors. Information shall be deemed not to include information that (a) is or becomes (other than by disclosure by Client) publicly known or is contained in a publicly available document; (b) is furnished by Manatron to others without restrictions similar to those imposed by this Master Agreement and the Supplemental Agreements; (c) is rightfully in Client's possession without the obligation of nondisclosure prior to the time of its disclosure under this Master Agreement and the Supplemental Agreements or any prior agreements between Manatron and Client; or (d) is independently developed by employees or agents of Client who can be shown to have had no access to the Proprietary and Confidential Information.
 - 1.20 "Services" means the services to be performed by Manatron for Client as identified on one or more Supplemental Agreement(s).
 - 1.21 "Software" means the Manatron Proprietary Software and Non-Manatron Software.
 - 1.22 "Supplemental Agreements" has the meaning set forth on the signature page. The types of Supplemental Agreements are: Integrated Systems Installation Supplemental Agreement, Annual Software License and Maintenance Supplemental Agreement and Professional Services Supplemental Agreement.
 - 1.23 "Use" means reading the Manatron Proprietary Software or Non-Manatron Sublicensed Software into and out of memory and the execution of such Software, in whole or in part, by the Licensed Server.

2. SUPPLEMENTAL AGREEMENTS.

2.1 Supplemental Agreements. Manatron will furnish to Client and Client will accept and pay for Products, Services and Integrated Systems itemized on Supplemental Agreements entered into by Client and Manatron, which together with the terms in the Supplemental Agreements, are an integral part of this Master Agreement. Supplemental Agreements will refer to this Master Agreement by number and will be signed by Client and Manatron. All references to Products, Services and Integrated Systems in this Master Agreement are to the Products, Services and Integrated

Systems listed on any Supplemental Agreements submitted to and accepted by Manatron pursuant to Sections 2.2 and 2.3, as modified by any Change Requests entered into by Client and Manatron pursuant to Section 2.4.

- 2.2 Additional Requests. Client may order additional Products, Services and Integrated Systems under this Master Agreement by submitting properly completed Supplemental Agreements referencing this Master Agreement, signed by an authorized representative of Client. Under no circumstances will additional orders be effective until accepted in writing by Manatron.
- 2.3 Subject to Acceptance. All Supplemental Agreements are subject to acceptance by Manatron. Manatron's acceptance will be effective when Manatron signs the Supplemental Agreements. Manatron's receipt or deposit of a Client purchase order or down payment will not constitute acceptance of a Supplemental Agreement. Manatron will return any down payment received from Client if Manatron does not accept the Supplemental Agreement.
- 2.4 Change Request. Client and Manatron may at any time modify a Supplemental Agreement by written Change Request, signed by both parties, identifying the modified Supplemental Agreement and specifying the modifications to at least the same degree of specificity as in the original specifications. The Change Request will include all changes and additions being made to the terms of the applicable Supplemental Agreement, Manatron will not be bound by any modifications to a Supplemental Agreement unless made by written Change Request signed by authorized representatives of both parties. A Change Request, when signed by both parties, will be subject to the terms of the applicable Supplemental Agreement, as modified by the Change Request, and this Master Agreement.
- 2.5 <u>Substitution</u>. With notification to Client and Client's prior approval, such approval not to be unreasonably denied, Manatron may substitute Product(s) of equivalent or superior functionality and performance in the event that any of the Product(s) ordered are not available. If Manatron reasonably determines that the substitute Product(s) would be more suitable, this substitute will be documented on the Change Request Form, modifying the Hardware or Software listed in Exhibit A (Pricing and Inventory) of the applicable Supplemental Agreement(s) and submitted to Client for approval, which approval shall not be unreasonably withheld.
- 2.6 Training. Manatron will provide training to Client's personnel as specified in one or more Integrated System Installation Supplemental Agreement(s) or Professional Services Supplemental Agreement(s). Client will be responsible for all training not specifically provided for in Supplemental Agreement(s).

3. LICENSES AND SUBLICENSES.

3.1 Supplemental Agreements. Manatron will provide Client with licenses and sublicenses under Section 3.2 to the Manatron Proprietary Software and Non-Manatron

Sublicensed Software, respectively, which is specified by Manatron and Client in Exhibit A (Pricing and Inventory) of one or more Annual Software License and Maintenance Supplemental Agreement(s). Licenses and sublicenses provided under an Annual Software License and Maintenance Supplemental Agreement will begin with the date of the Annual Software License and Maintenance Supplemental Agreement and end upon termination or expiration of such agreement. The terms of this Agreement, including but not limited to this Section 3, and the terms of the applicable Annual Software License and Maintenance Supplemental Agreement will apply to all licenses of Manatron Proprietary Software and to all sublicenses of Non-Manatron Sublicensed In some cases, Non-Manatron Software Software. (consisting of Non-Manatron Other Software) may be provided subject to a license directly from the Licensor to If a separate license agreement applies to or accompanies Non-Manatron Software, then the separate license agreement terms will apply and supersede the license terms in this Master Agreement and the Supplemental Agreements for that Non-Manatron Software. Client agrees to comply with the terms of all licenses governing Manatron Proprietary Software and Non-Manatron Software.

3.2 Licenses and Sublicenses. Manatron grants to Client a personal, non-exclusive, nontransferable limited license or sublicense to Use the Manatron Proprietary Software and Non-Manatron Sublicensed Software, respectively, which is identified as licensed or sublicensed to Client in Exhibit A (Pricing and Inventory) of one or more Annual Software License and Maintenance Supplemental Agreement(s). Client is licensed or sublicensed to Use the Manatron Proprietary Software and Non-Manatron Sublicensed Software only for Client's internal data processing requirements at the Licensed Location on the Licensed Server by the Number of Licensed Users, each as specified in the applicable Annual Software License and Maintenance Supplemental Agreement. Client may temporarily transfer the Manatron Proprietary Software and Non-Manatron Sublicensed Software to a back-up server at an alternative location within Client's county of operation if the Licensed Server is inoperative or the Licensed Location is temporarily unavailable. Unless otherwise provided in the applicable Annual Software License and Maintenance Supplemental Agreement, Client will only be provided and permitted to use the executable form of Manatron Proprietary Software and Non-Manatron Sublicensed Software and such use must be in connection with the application package provide by Manatron. Client agrees that Manatron or its representatives may periodically inspect and audit, at mutually agreed upon times during normal business hours, the computer site, Integrated Systems and appropriate records of Client to verify Client's compliance with the terms of this Master Agreement and all applicable Supplemental Agreement(s) with respect to the Software supplied by Manatron.

3.3 Protection of Software.

- (a) Client shall not, under any circumstances, cause or permit the adaptation, conversion, reverse engineering, disassembly, modification or de-compilation of any Software. (b) Client shall not modify, assign, transfer, sublicense, time-share, rent, copy or duplicate the Software; provided, Client may have in its possession a reasonable number of copies of the Software for inactive archival or back-up purposes. All copies of the Software, in whole or in part, must contain all of Manatron's and the third party Licensor's titles, trademarks, copyright notices and other restrictive and proprietary notices and legends (including government restricted rights) as they appear on the copies of the Software provided to Client. Client shall notify Manatron of the following:
- (i) the location of all Software and all copies thereof, and
- (ii) any circumstances known to Client regarding any unauthorized possession or use of the Software.
- (c) Upon termination of Client's license or sublicense of Softwere, Client shall immediately discontinue all use of the Software and return to Manatron or destroy, at Manatron's option, the Software (and all related documentation and Proprietary and Confidential Information) and all archival, back-up and other copies thereof, and provide certification to Manatron of such return or destriction.
- (d) Client shall not publish any results of benchmark tests run on any Software.
- (e) Client shall not relocate the Manatron Proprietary Software or Non-Manatron Sublicensed Software from the Licensed Location, except as permitted in Section 3.2.
- (f) Client shall maintain the Manatron Proprietary Software and Non-Manatron Sublicensed Software in confidence and comply with the terms of Section 7, Protection of Proprietary and Confidential Information, with respect to such Software.
- (g) The terms of this Section 3.3 will survive the termination or expiration of this Agreement and the applicable Annual Software License and Maintenance Supplemental Agreement.
- 3.4 No Transfer of Title. This Agreement does not transfer to Client title to any Software, intellectual property contained in any Software, or Proprietary and Confidential Information. Title to Manatron Proprietary Software and all copies thereof, and all associated intellectual property rights therein, will remain in Manatron. Title to Non-Manatron Software and all copies thereof, and all associated intellectual property rights therein, will remain in the applicable third party Licensor.
- 3.5 Inherently Dangerous Applications. The Software is not developed or licensed for use in any nuclear, aviation, mass transit, or medical application or in any other inherently dangerous applications. Client shall not use the Software in any inherently dangerous application and agrees that

Manatron and any third party licensor will not be liable for any claims or damages arising from such use.

4. MAINTENANCE SERVICES.

- 4.1 Maintenance. Manatron will provide Maintenance Services to the extent provided for in Annual Software License and Maintenance Supplemental Agreement(s) or Professional Services Supplemental Agreement(s) and pursuant to the terms and conditions of Manatron's Description of Maintenance Services, as amended from time to time. Manatron will make best commercially reasonable efforts to remedy or provide a reasonable work-around for defects, errors or malfunctions in Manatron Proprietary Software which have a significant adverse affect upon operation of the Manatron Proprietary Software or Integrated System, as applicable, and which are promptly reported by Client to Manatron. Because not all errors or defects can or need be corrected, it is possible that some errors or defects will not be corrected. Errors or defects must be reported on Manatron's Client Service Request Form, Exhibit C to the applicable Annual Software License and Maintenance Supplemental Agreement and be accompanied with sufficient detail, if available, to enable Manatron to reproduce the error and provide a remedy or suitable work-around. The remedies and obligations set forth in this Section 4.0 are the full extent of Client's remedies and the full extent of Manatron's obligations with respect to errors, defects or malfunctions of the Integrated System, Manatron Proprietary Software and Non-Manatron Sublicensed Software.
 - 4.2 Limitation. Manatron will not provide Maintenance Services if alterations to Products or Integrated Systems which are not made by or coordinated with Manatron or attachments to Products or Integrated Systems which are not provided and installed by or coordinated with Manatron directly or indirectly result in any malfunction, nonperformance or degradation of performance of Products or Integrated Systems.

4.3 Exclusions.

- (a) Maintenance Services apply only to properly configured Products. This exclusion for improper configuration is not applicable to Maintenance Services for any Hardware or Software installed by Manatron or under Manatron supervision, unless Client or any third party has changed the configuration without Manatron's supervision. Maintenance Services also apply only at the minimum Hardware and Software levels designated by Manatron for support of the applicable Product specifications.
- (b) Maintenance Services do not include correction or repair of defects, errors or malfunctions, including any related to date data functionality, in the design, manufacture, materials or workmanship of either (i) Non-Manatron Other Software, or (ii) Hardware.
- (c) Maintenance Services described in this Section 4 and under Annual Software License and Maintenance Supplemental Agreement(s), including the Description of

Maintenance Services exhibit, do not cover defects, errors or malfunctions which are not attributable to the relevant Manatron Proprietary Software or Integrated System or which are caused by any of the following: (i) de-installation, reinstallation or relocation of any item of Hardware by Client or any third party; (ii) Client's failure to follow operational or maintenance instructions as set forth in applicable documentation; (iii) the use of non-compatible media or supplies; (iv) repair, maintenance, modification or alteration of the Manatron Proprietary Software, Hardware or Integrated System by Client or third parties; (v) use of hardware or software not supplied or authorized by Manatron: (vi) external factors (e.g., power failure, surges or electrical damage, fire or water damage, air conditioning failure, humidity control failure, or corrosive atmosphere harmful to electronic circuitry); (vii) failure to maintain proper site specifications and environmental conditions; (viii) negligence, accidents, neglect, misuse or tampering; (ix) improper or abnormal use or use under abnormal conditions; (x) use in a manner not authorized by this Master Agreement and any Supplemental Agreement(s) or use inconsistent with Manatron's specifications; (xi) Client's failure to comply with Client's responsibilities under Section 6; (xii) attachments or alterations not provided and installed by Manatron as further described in Section 4.4 or (xiii) the introduction of software viruses.

- (d) Manatron reserves the right to charge on a time and materials basis for efforts expended due to problems caused by these maintenance exclusions, using Manatron's standard time and materials charges, subject to Client's prior approval.
 - 4.4 Alterations and Attachments.
- (a) Client will be solely responsible for infringement, personal injury or damage to property, Products and Integrated Systems resulting from alterations to Products or Integrated Systems that are not made by Manatron or attachments to Products or Integrated Systems that are not provided and installed by Manatron.
- (b) Client will give Manatron prior written notice of any proposed alterations or attachments to Products or Integrated Systems subject to Maintenance Services. Manatron has no obligation to provide Maintenance Services for Products or Integrated Systems containing alterations not made by Manatron or attachments not provided and installed by Manatron. If Manatron agrees to maintain, support or correct altered Products or Integrated Systems, Manatron may impose additional fees. Manatron is not responsible for a realfunction, nonperformance or degradation of performance. of Products or Integrated Systems caused by or resulting directly or indirectly from any alteration or attachment unless Manatron has provided and installed the alteration or attachment and has informed Client that such adverse conditions will not occur. For purposes of this Master Agreement "alterations" includes, but is not limited to, the incorporation of components, boards and subassemblies not provided by Manatron into Products or Integrated Systems, as

well as modifications to Products or Integrated Systems that are not made by Manatron. "Attachments" includes any hardware, software, components or devices which are connected to Products or Integrated Systems and which are not provided by Manatron.

5. CHARGES; PAYMENT.

- 5.1 Charges. Charges for Products, Services and Integrated Systems will be identified and payable in accordance with the terms set forth in the relevant Supplemental Agreement(s). All payments are to be made to Manatron at its principal office in Austin, Texas, as set forth on the signature page or to such other location as may be designated by Manatron in a notice to Client.
- 5.2 Late Charges. All uncontested charges must be paid as agreed in a Supplemental Agreement. If invoiced, all payments are due no later than thirty (30) days from the date of receipt of invoice by Client. Manatron may impose a late payment charge on past due payments equal to the lesser of (a) 1% per month or (b) the maximum rate allowed by law.
- 5.3 Additional Charges. Subject to Manatron's notification to Client and Client's prior approval, additional charges may be assessed for services rendered outside contracted hours or beyond normal coverage at Client's request, including but not limited to travel expenses, premium and minimum charges.
- 5.4 Payment Disputes. If any dispute exists between the parties concerning the amount due or due date of any payment, Client shall promptly pay the undisputed portion. Such payment will not constitute a waiver by Client or Manatron of any of their respective legal rights and remedies against each other.

5.5 Price Protection.

- (a) The prices for Products in any Supplemental Agreement will remain firm through delivery, unless through no fault of Manatron shipment takes place more than one year after the date of the Supplemental Agreement. If Manatron notifies Client that an increase in prices will apply to its order and the affected part of the order is not part of an Integrated System, Client may terminate the affected part of its order by giving written notice to Manatron within ten (10) days of the date of notification of the increase.
- (b) Fees for Maintenance Services under an Annual Software License and Maintenance Supplemental Agreement will not be increased during the twenty-four (24) month period beginning with the Installation Date, but may be increased on the second (2nd) anniversary of the Installation Date and on each subsequent anniversary of the Installation Date with thirty (30) days prior written notice to Client.
- (c) Fees for Software licenses, Software sublicenses and Services under Supplemental Agreements of at least one year will not be increased during the twenty-four (24) month period beginning with the Installation Date, but may be increased on the second (2nd) anniversary of the Installation

Date and on each subsequent anniversary of the Installation Date with thirty (30) days prior written notice to Client. If Software or Services are contracted on a month-to-month basis, the fees may be increased at any time with sixty (60) days prior written notice to Client.

- (d) Manatron will limit the increase in annual license and support fees to 5% or the CPI, whichever is greater. The base for computing the adjustment is the Consumer Price Index All Urban Consumers for the Houston/Harris County area published by the United States Department of Labor, Bureau of Labor Statistics ("Index"). The Index published most immediately preceding the Adjustment Date in question ("Extension Index") is to be used in determining the amount of the increase.
- 5.6 Taxes. If Client is tax exempt, Client will provide Manatron with proof of its tax-exempt status. If Client is not tax-exempt, (a) Client will pay any tax Manatron becomes obligated to pay in connection with this Agreement, exclusive of taxes based on the net income of Manatron, and (b) Client will pay all personal property and similar taxes assessed after shipment. If Client challenges the applicability of any such tax, Client shall pay the tax and may thereafter seek a refund.
- 5.7 <u>Delivery</u>. Unless otherwise specified in the relevant Supplemental Agreement, Manatron will arrange for delivery of Integrated Systems and Products to Client and delivery charges will be included in Manatron's pricing.
- 5.8 Installation. Unless otherwise provided in the relevant Supplemental Agreement, if the Supplemental Agreement provides for Manatron to install Integrated Systems or Products, (a) installation will be performed during Manatron normal working hours, (b) all installation will be subject to the then-current standard Manatron charges and conditions, and (c) if additional labor and rigging is required for installation due to Client's special site requirements, Client will pay those costs including costs to meet union or local law requirements.

6. CLIENT RESPONSIBILITIES.

- 6.1 <u>Independent Determination</u>. Client acknowledges that, based on Manatron's representations and Client's own due diligence, Client has determined that the Products, Services and Integrated Systems ordered under this Master Agreement and Supplemental Agreement(s) meet its requirements.
- 6.2 Cooperation. Client agrees to cooperate with Manatron and promptly perform Client's responsibilities under this Master Agreement, including but not limited to those set forth in any Supplemental Agreement. Client shall:
- (a) provide adequate working and storage space for use by Manatron personnel near Hardware and Integrated Systems;
- (b) provide Manatron full access to the Hardware, Software and Integrated Systems and sufficient computer time, subject only to Client's security rules;

- (c) follow Manatron's procedures for placing service requests and determining if remedial service is required;
- (d) follow Manatron's or manufacturer's instructions for operator maintenance and obtaining services;
- (e) provide a memory dump and additional data in machine-readable form if requested;
- (f) reproduce suspected errors or malfunctions in Software;
- (g) install all Manatron Software and Non-Manatron Software releases supplied by Manatron within ninety (90) days after receipt of such release by Client or, if the applicable Supplemental Agreement provides for Manatron to install the releases, then Client shall fully cooperate with Manatron's installation of the Manatron Software and Non-Manatron Software releases, as applicable;
- (h) provide timely access to key customer personnel and timely response to Manatron's questions; and
- (i) otherwise cooperate with Manatron in its performance under this Master Agreement and Supplemental Agreements.
- 6.3 Site Preparation. Client is responsible for compliance with all local labor concerns and building codes. If Manatron is to install Products, Client shall prepare and maintain the installation site in accordance with the manufacturer's instructions and instructions provided by Manatron and ensure that these instructions are not in violation of labor laws or building ordinances. Client is responsible for environmental requirements, electrical interconnections and modifications to facilities for proper installation. Any delays in preparation of the installation site will correspondingly extend Manatron's delivery and installation deadlines.
- 6.4 Site Maintenance. Client shall maintain the appropriate operating environment, in accordance with the manufacturer's specifications and Manatron's specifications, for the Hardware, Software and Integrated Systems and all communications hardware, telephone lines, electric lines, cabling, moderns, air conditioning and all other hardware and utilities necessary for the Hardware, Software and Integrated Systems to operate properly.
- 6.5 Use. Client is exclusively responsible for supervising, managing and controlling its use of the Hardware, Software and Integrated Systems, including but not limited to, establishing operating procedures and audit controls, supervising its employees, making daily backups, providing virus protection, inputting data, ensuring the accuracy and security of data input and data output, monitoring the accuracy of information obtained, and managing the use of information and data obtained. Client will ensure that its personnel are, at all times, educated and trained in the proper use and operation of the Products and that the Products are used in accordance with applicable manuals, instructions and specifications.

- 6.6 Backups. Client will maintain back-up data necessary to replace critical Client data in the event of loss or damage to data from any cause.
- 6.7 Compliance with Third Party Contracts. Client represents and warrants to Manatron that it does not have any contracts or other obligations to third parties, including but not limited to any license agreements or confidentiality obligations, that will be violated in any respect by Client's or Manatron's performance under this Master Agreement or any Supplemental Agreement(s).
- PROTECTION OF PROPRIETARY AND CONFIDENTIAL INFORMATION.
- 7.1 Manatron Proprietary and Confidential Information. Client shall keep in confidence and protect Proprietary and Confidential Information from disclosure to third parties and restrict its use to uses expressly permitted under this Master Agreement and the Supplemental Agreements subject to the requirements of the Texas Public Information Act. Client shall take all reasonable steps to ensure that Proprietary and Confidential Information is not disclosed, copied, duplicated, misappropriated or used in any manner not expressly permitted by the terms of this Master Agreement and Supplemental Agreement(s). Client shall keep the Software and all tapes, CDs, diskettes and other physical embodiments of the software. and all copies thereof, at a secure location and limit access to those employees who must have access to enable Client to use Client acknowledges that unauthorized the Software. disclosure of Proprietary and Confidential Information may cause substantial economic loss to Manatron or its suppliers and licensors. Client agrees not to copy Proprietary and Confidential Information, in whole or in part, except as expressly authorized by this Master Agreement and any Supplemental Agreement(s). Each permitted copy of Proprietary and Confidential Information, including its storage media, will be marked by Client to include all notices and legends (including government restricted rights) that appear on the original. Title, copyright and all other proprietary rights in and to the Software, at all times remain vested exclusively in Manatron or, as applicable, the respective third party licensors.
- Client Confidential Information. acknowledges that it and its employees or agents may, in the course of performing their responsibilities under this Master Agreement and the Supplemental Agreements, be exposed to or acquire information that is confidential to the Client. Any and all information of any form obtained by Manatron or its employees or agents in the performance of this Master Agreement and the Supplemental Agreements shall be deemed to be confidential information of the Client. Any reports or other documents or items (including software) that result from the use of the confidential information by Manatron shall be treated with respect to confidentiality in the same manner as the Proprietary and Confidential Information. Confidential information shall be deemed not to include information that (a) is or becomes (other than by disclosure by Manatron) publicly known or is contained in a publicly

available document; (b) is furnished by the Client to others without restrictions similar to those imposed by this Master Agreement and the Supplemental Agreements; (c) is rightfully in Manaron's possession without the obligation of nondisclosure prior to the time of its disclosure under this Master Agreement and the Supplemental Agreements; or (d) is independently developed by employees or agents of Manaron who can be shown to have had no access to the confidential information.

- Return of Proprietary and Confidential Information. Upon termination or cancellation of this Master Agreement, Client shall immediately destroy or return to Manatron, at Manatron's option, all Proprietary and Confidential Information in Client's possession or control (including Proprietary and Confidential Information incorporated in other software or writings) and provide certification to Manatron of such return or destruction. Upon termination of a Supplemental Agreement, Client shall immediately destroy or return, at Manatron's option, all Proprietary and Confidential Information in Client's possession or control (including Proprietary and Confidential Information incorporated in other software or writings) relating to such Supplemental Agreement, which is no longer licensed or sublicensed to Client pursuant to another Supplemental Agreement.
 - 7.4 Intellectual Properties. All ideas, concepts, know-how, data processing techniques, Software, documentation, diagrams, schematics, blueprints and trade secrets developed by Manatron personnel (alone or jointly with Client) in connection with Proprietary and Confidential Information or Products, Services and Integrated Systems provided to Client will be the exclusive property of Manatron. Manatron grants to Client a non-exclusive, royalty-free license to use of any of the foregoing in connection with Client's use of the Products and/or Integrated System as permitted by the terms of this Master Agreement and the Supplemental Agreement(s).
 - 7.5 Support Materials. Client acknowledges that all support materials, including without limitation, diagnostic software, are the property of and include Proprietary and Confidential Information of Manatron. Client will not use such materials. Manatron has the right to remove such materials from Client's facility at any time. This provision applies unless Client has specifically purchased or licensed such support materials pursuant to a Supplemental Agreement.
 - 7.6 Client Employees. Client will inform its employees of their obligations under this Section 7 to ensure such obligations are met.
 - 7.7 <u>Survival</u>. This Section 7 will survive termination or cancellation of this Agreement and the Supplemental Agreement(s).
 - 8. TITLES; RISK OF LOSS.

8.1 Hardware. Title to Hardware and risk of loss of or damage to Hardware will pass to Client upon delivery to Client or Manatron as Client's agent.

8.2 Software.

- (a) Manarron Proprietary Software. Title to Manatron Proprietary Software and all associated intellectual property rights will remain in Manatron, including but not limited to all copyrights, trade secrets, trademarks and other proprietary rights in and to the Manatron Proprietary Software.
- (b) Non-Manatron Software. Title to Non-Manatron Software and all associated intellectual property rights will remain in the applicable third party Licensor including but not limited to all copyrights, trade secrets, trademarks and other proprietary rights in and to the Non-Manatron Software.
- (c) Risk of Loss. Risk of loss to Software will pass to Client on the Installation Date, except to the extent covered by the limited warranties in Section 9 or Maintenance Services pursuant to an Annual Software License and Maintenance Supplemental Agreement, as applicable.
- 8.3 <u>Data</u>. Client will retain all title, rights, and ownership of all images and associated indexes, and other data created and/or acquired by use of the Hardware, Software and Integrated Systems as stored on magnetic disk, magnetic tape, optical disk, optical tape, CD-ROM disk (or other "like" electronic media that may be used).
- 8.4 Proprietary and Confidential Information. Title to Manatron's Proprietary and Confidential Information will remain in Manatron. Title to Proprietary and Confidential Information of Manatron's suppliers and licensors will remain in the relevant suppliers and licensors. Title to Client's confidential information will remain in Client.
 - 9. REPRESENTATIONS AND WARRANTIES.

9.1 Software.

- (a) <u>Title</u>. <u>Manatron</u> represents and warrants that it owns or has the right to license or sublicense the Manatron Proprietary Software and Non-Manatron Sublicensed Software licensed or sublicensed by Manatron to Client as provided for under Annual Software License and Maintenance Supplemental Agreement(s) or other prior Agreements.
- (b) Services. Manatron warrants to Client that Manatron has the skill and knowledge ordinarily possessed by well-informed members of its trade and profession and Manatron will apply that skill and knowledge with care and diligence to perform Services under the Supplemental Agreements in accordance with best commercially reasonable professional standards and in accordance with industry standards. If Client believes Manatron has breached this warranty, Client shall promptly, and in any event within fourteen (14) days after Client becomes aware of the breach, notify Manatron in writing of the claimed breach, with substantiating documentation, and request that Manatron re-perform the

Services subject to the breach. If there is a breach of this warranty, Manatron shall re-perform the Services for which there has been a breach. Manatron will not be responsible under this warranty if Client fails to timely notify Manatron of the claimed breach or if Client fails to substantiate the breach with documentation. The remedy set forth in this section is Client's sole and exclusive remedy, and the full extent of Manatron's liability, for Manatron's breach of this warranty.

(c) Conformance to Specifications. Manatron warrants to Client that upon the Installation Date or, if there is an acceptance testing process upon acceptance by Client, the Hardware, Software and Integrated System, as applicable, will conform and perform materially to the specifications in the related Supplemental Agreement. Client shall inspect the Hardware Software and Integrated System for compliance with this warranty promptly upon receipt and installation. If Client believes Manatron has breached this warranty, Client shall within fourteen (14) days after Client knew or should have known of the breach, and in any event no later than ninety (90) days after the Installation Date (or date of acceptance if there is an acceptance testing process), notify Manatron in writing of the claimed breach, with substantiating documentation, and request that Manatron provide compliant Hardware, Software and Integrated System, as applicable, or, at Manatron's election, refund the amount paid for the Hardware, Software and Integrated System subject to the breach. If there is a material breach of this warranty, Manatron will, at Manatron's election, either provide materially compliant Hardware, Software and Integrated System, as applicable, for which there has been a material breach or refund the amount paid by Client with respect to the Hardware, Software and Integrated System, as applicable, for which there has been a material breach. Because not all errors or defects can or need to be corrected, it is possible that some errors or defects will not be corrected. Errors or defects relating to Software must be reported on Manatron's Client Service Request Form, Exhibit C to the applicable Annual Software License and Maintenance Supplemental Agreement and be accompanied with sufficient detail, if available, to enable Manatron to reproduce the error and provide a remedy or suitable work-around, Manatron will not be responsible under this warranty if Client fails to promptly and timely notify Manatron of the claimed breach or if Client fails to substantiate the breach with documentation. Client further acknowledges and agrees that this warranty will be deemed to be satisfied upon acceptance by Client of the Hardware, Software and/or Integrated System, as applicable, pursuant to Client's inspection or pursuant to acceptance testing done with respect to the Hardware, Software and/or Integrated System, as applicable, This warranty does not apply if the failure of the Hardware, Software or Integrated System to conform to this warranty is caused by any of the exclusions from Maintenance Services that are set forth in Section 4.2, Section 4.3, and/or Section 4.4 which are incorporated herein by reference as exceptions

Page 9 of 14

to this warranty. The remedy set forth in this section is Client's sole and exclusive remedy, and the full extent of Manatron's liability, for Manatron's breach of this warranty.

- 9.2 Initial Maintenance Period at No Extra Charge. During the one-year period beginning with the Installation Date, in lieu of a one (1) year warranty, Manatron will provide to Client, at no extra charge, Maintenance Services at Level One, Basic Level of Service on the terms and conditions (including the exclusions, limitations and restrictions) set forth in the applicable Annual Software License and Maintenance Supplemental Agreement, Manatron's Description of Maintenance Services, as amended from time to time, and Sections 4, and 9.8 of this Master Agreement. Client's remedies are limited to Maintenance Services as set forth in this Section 9.2.
- 9.3 Conversion. If Manatron converts Client's legacy data pursuant to a Supplemental Agreement, Manatron warrants, for the benefit of Client only, that Manatron has converted the legacy data as extracted by Client into the Integrated System as directed by Client. Client shall review the converted data and Client's sole remedy and Manatron's sole obligation for conversion services will be to correct any errors caused by conversion of the data by Manatron, as detected by Client. Manatron will not be obligated to correct errors in the data provided to Manatron. The remedies and obligations set forth in this Section 9.3 are the full extent of Client's remedies and the full extent of Manatron's obligations with respect to errors caused by the conversion of data by Manatron.
- 9.4 Non-Manatron Software, EXCEPT AS OTHERWISE SPECIFICALLY PROVIDED IN SECTION 9.1 (c), MANATRON MAKES NO OTHER REPRESENTATIONS WARRANTIES AS TO NON-MANATRON SOFTWARE, ALL OF WHICH IS SOLD OR LICENSED TO CLIENT "AS IS." Manatron will pass through to Client, on a non-exclusive basis and without recourse to Manatron, any third party manufacturer's and Licensor's warranties covering Non-Manatron Software, but only to the extent, if any, permitted by the third party manufacturer and Licensor. Client may independently seek to obtain directly from the manufacturers or Licensors of the Non-Manatron Other Software maintenance of the Non-Manatron Other Software under any warranty or guarantee provided by such third party manufacturer or Licensor. Client agrees to look solely to the warranties and remedies, if any, provided by the third party manufacturer or Licensor. The remedies and obligations set forth in this Section 9.4 are the full extent of Client's remedies and the full extent of Manatron's obligations with respect to warranties of Non-Manatron Software.
- 9.5 Hardware. EXCEPT AS OTHERWISE SPECIFICALLY PROVIDED IN SECTION 9.1 (c), MANATRON MAKES NO OTHER REPRESENTATIONS OR WARRANTIES AS TO HARDWARE, ALL OF WHICH IS SOLD OR LICENSED TO CLIENT "AS IS."

MANATRON will pass through to Client, on a non-exclusive basis and without recourse to MANATRON, any third party manufacturer's warranties covering the Hardware, but only to the extent, if any, permitted by the third party manufacturer. Client may independently seek to obtain directly from the manufacturers of the Hardware maintenance of the Hardware under any warranty or guarantee provided by such third party manufacturer. Client agrees to look solely to the warranties and remedies, if any, provided by the manufacturer. The remedies and obligations set forth in this Section 9.5 are the full extent of Client's remedies and the full extent of MANATRON's obligations with respect to warranties of Hardware.

Compliance with Third Party Contracts. 9.6 MANATRON represents and warrants to Client that it does not have any contracts or other obligations to third parties, including but not limited to any license agreements or confidentiality obligations, that will be violated in any respect by MANATRON's or Client's performance under this Master Agreement or any Supplemental Agreement(s).

9.7 Proof of Insurance.

- Manatron shall obtain and maintain, throughout the term of the Master Agreement and the Supplemental Agreements, insurance of the types and in the minimum amounts set forth below. Manatron shall furnish certificates of insurance to Client evidencing compliance with the insurance requirements hereof. Certificates shall indicate Manatron's name, name of insurance company, policy number, term of coverage and limits of coverage. Manatron shall cause its insurance companies to provide Client with at least thirty (30) days prior written notice of any reduction in the limit of liability by endorsement of the policy, cancellation or non-renewal of the insurance coverage required under this Agreement. Manatron shall obtain such insurance from such companies having a Bests rating of B+/VII or better, licensed or approved to transact business in the State of Texas, and shall obtain such insurance of the following types and minimum limits:
- (i) Workers' Compensation insurance in accordance with the laws of the State of Texas, or state of hire/location of Services, and Employers' Liability coverage with a limit of not less than \$500,000 each employee for Occupational Disease, \$500,000 policy limit for Occupational Disease; and Employer's Liability of \$500,000 each accident.
- (ii) Employers' Liability insurance with limits not less than \$1,000,000 per injury by accident, \$1,000,000 per injury by disease, and \$1,000,000 per bodily injury by disease.
- (iii) Commercial General Liability insurance including coverage for Products/Completed Operations, Blanket Contractual, Contractor's Protective Liability Broad Form Property Damage, Personal Injury/Advertising Liability, and Bodily Injury and Property Damage with limits not less than:

\$2,000,000 annual aggregate limit

Page 10 of 14

- \$1,000,000 cach occurrence, combined single limit
- (iv) Business Automobile Liability coverage applying to owned, non-owned and hired automobiles with limits not less than \$1,000,000 each occurrence combined single limit for Bodily Injury and Property Damage combined.
- (v) Professional Liability insurance with limits not less than \$1,000,000 each claim/annual aggregate.
- Client shall be named as additional insured to all coverages required above. All policies written on behalf of Manatron shall contain a waiver of subrogation in favor of Client.
- 9.8 DISCLAIMER. MANATRON DOES NOT REPRESENT OR WARRANT THAT THE FUNCTIONS PROPRIETARY MANATRON CONTAINED IN SOFTWARE OR INTEGRATED SYSTEMS WILL MEET REQUIREMENTS THAT ARE CLIENT'S FORTH IN APPLICABLE **EXPRESSLY** SET SUPPLEMENTAL AGREEMENTS. MANATRON DOES NOT REPRESENT OR WARRANT THAT PROPRIETARY SOFTWARE OR MANATRON WILL OPERATE TN INTEGRATED SYSTEMS COMBINATIONS SELECTED FOR USE BY CLIENT WITH HARDWARE OR SOFTWARE NOT APPROVED EXCEPT FOR THE EXPRESS BY MANATRON. WARRANTIES IN THIS MASTER AGREEMENT, TO EXTENT PERMITTED MUMIXAM THERE ARE NO LAW: (A) APPLICABLE WARRANTIES, IMPLIED BY OPERATION OF LAW OR OTHERWISE, AND (B) MANATRON DISCLAIMS ALL WARRANTIES, INCLUDING BUT NOT IMPLIED TO, THE IMPLIED WARRANTIES OF LIMITED FITNESS FOR A PARTICULAR PURPOSE AND MERCHANTABILITY FOR MANATRON AND NON-MANATRON PRODUCTS AND SERVICES. EXPRESS LIMITED WARRANTIES EXTEND SOLELY TO CLIENT.

10. LIMITATION OF DAMAGES.

- 10,1 <u>EXCLUSIVE REMEDY</u>. MANATRON'S ENTIRE LIABILITY AND CLIENT'S EXCLUSIVE REMEDY FOR CONCERNING THIS MASTER CLAIM AGREEMENT AND SUPPLEMENTAL AGREEMENTS, AND THE PRODUCTS, SERVICES AND INTEGRATED SYSTEMS PROVIDED UNDER THIS MASTER AGREEMENT AND SUPPLEMENTAL AGREEMENTS, ARE SET FORTH IN THIS SECTION.
- 10.2. <u>DISCLAIMER</u>. CLIENT IS RESPONSIBLE FOR ASSURING AND MAINTAINING THE BACK-UP OF ALL CLIENT DATA. UNDER NO CIRCUMSTANCES WILL MANATRON BE LIABLE TO CLIENT OR ANY THIRD PARTY FOR THE LOSS OF OR DAMAGE TO CLIENT DATA.
- 10.3 <u>INDEMNITY</u>. MANATRON SHALL SAVE HARMLESS CLIENT FROM AND AGAINST ALL SHALL SAVE CLAIMS, LIABILITY, AND EXPENSES, INCLUDING

REASONABLE ATTORNEYS FEES FOR INJURY TO PERSONS OR DAMAGE TO TANGIBLE PROPERTY, ARISING FROM ACTIVITIES OF MANATRON, ITS EMPLOYEES. CONSULTANTS OR AGENTS. PERFORMED UNDER THIS AGREEMENT THAT RESULT FROM THE NEGLIGENT ACT, ERROR, OR OR. ANY MANATRON OMISSION OF MANATRON'S AGENTS, SERVANTS OR EMPLOYEES. MANATRON'S OBLIGATION TO SAVE HARMLESS CLIENT IS CONDITIONED UPON CLIENT: (A) GIVING MANATRON PROMPT WRITTEN NOTICE OF ALL SUCH CLAIMS FOLLOWING RECEIPT OF SUCH CLAIMS BY CLIENT, (B) PERMITTING MANATRON TO CONTROL THE DEFENSE AND SETTLEMENT OF ALL SUCH CLAIMS, AND (C) REASONABLY COOPERATING WITH MANATRON IN THE DEFENSE AND SETTLEMENT OF ALL SUCH CLAIMS. IN NO EVENT WILL MANATRON BE LIABLE FOR ANY INTO WITHOUT **ENTERED** SETTLEMENTS MANATRON'S WRITTEN CONSENT.

NOTWITHSTANDING LIMITATION. ANYTHING TO THE CONTRARY IN THIS MASTER SUPPLEMENTAL ANY AGREEMENT OR THE MAXIMUM EXTENT TO AGREEMENT. PERMITTED BY APPLICABLE LAW: MANATRON AND ANY PARTY INVOLVED IN THE CREATION, MANUFACTURE OR DISTRIBUTION OF PRODUCTS AND INTEGRATED SYSTEMS ÓR PERFORMANCE OF THE SERVICES, WILL NOT BE LIABLE TO CLIENT FOR ANY SPECIAL, INDIRECT, INCIDENTAL, PUNITIVE, OR CONSEQUENTIAL DAMAGES (INCLUDING LOST PROFITS) OR FOR LOST DATA SUSTAINED OR INCURRED CONNECTION WITH THIS MASTER AGREEMENT OR SUPPLEMENTAL AGREEMENTS, EVEN IF ADVISED OF THE POSSIBILITY OF SUCH DAMAGES, REGARDLESS OF THE FORM OF ACTION AND SUCH DAMAGES ARE WHETHER OR NOT FORESEEABLE. IN ADDITION, TO THE MAXIMUM EXTENT PERMITTED BY APPLICABLE LAW, MANATRON'S LIABILITY TO CLIENT FOR DIRECT DAMAGES ARISING OUT OF OR RELATING TO THIS SUPPLEMENTAL AND AGREEMENT MASTER AGREEMENTS WILL IN NO EVENT EXCEED THE TOTAL AMOUNT ACTUALLY PAID BY CLIENT TO SUPPLEMENTAL THE MANATRON UNDER AGREEMENT WITH RESPECT TO WHICH THE DIRECT DAMAGES WERE INCURRED TIMES 2.0.

10.5 Referrals. Manatron may direct Client to third parties having products or services that may be of interest to Client for use in conjunction with the Products, Services or Integrated Systems. Notwithstanding any Manatron recommendation, referral or introduction, Client will independently investigate and test non-Manatron products and services and will have sole responsibility for determining suitability for use of non-Manatron products and services. Manatron has no liability with respect to claims relating to or Version 01-06-06 Page 11 of 14

arising from use of non-Manatron products and services, including, without limitation, claims arising from failure of non-Manatron products to provide proper time and date functionality.

11. INFRINGEMENT INDEMNITY.

11.1 Indemnity. Manatron, at its own expense, will defend and indemnify Client against claims that Manatron Proprietary Software furnished under this Master Agreement or Supplemental Agreements infringe any patent or copyright or misappropriate trade secrets protected under United States law, provided Client (a) gives Manatron prompt written notice of such claims pursuant to Section 14.11, (b) permits Manatron to control the defense and settlement of the claims, and (c) provides all reasonable assistance to Manatron in defending and settling the claims.

11.2 Remedies. As to any Manatron Proprietary Software which is subject to a claim of infringement or misappropriation, Manatron may (a) obtain the right of continued use of the Manatron Proprietary Software for Client or (b) replace or modify the Manatron Proprietary Software to avoid the claim. If neither alternative is available on commercially reasonable terms, then, at the request of Manatron, any applicable Software license and its charges will end, Client will stop using the Manatron Proprietary Software, and Client will return to Manatron or destroy all copies of the Manatron Proprietary Software, and will certify in writing to Manatron that such return or destruction has been completed. Upon return or Manatron's receipt of certification of destruction of the Manatron Proprietary Software, Manatron will give Client a credit for the price paid to Manatron, less a reasonable offset for use and obsolescence.

11.3 Exclusions. Manatron will not defend or indemnify Client if any claim of infringement or misappropriation (a) is asserted by an affiliate of Client, (b) results from Client's design or alteration of any Manatron Proprietary Software, (c) results from use of any Manatron Proprietary Software in combination with any non-Manatron product, except to the extent, if any, that such use in combination is part of an Integrated System designed and installed by Manatron for Client, or (d) relates to a non-Manatron Product alone.

11.4 Exclusive Remedies. This Section 11 states the entire liability of Manatron and Client's sole and exclusive remedies for patent or copyright infringement and trade secret misappropriation.

12. TERMINATION.

12.1 Term. The term of this Master Agreement will be for one (1) year from the Effective Date (the "Initial Term"). This Master Agreement will automatically renew for consecutive one (1) year terms thereafter ("Renewal Terms") unless either party notifies the other of its election not to renew the terms of this Master Agreement at least sixty (60) days prior to the expiration of the Initial Term or any Renewal Term. Notwithstanding the termination of this

Master Agreement upon the expiration of the Initial Term or any Renewal Term, this Master Agreement will remain in effect with respect to any Supplemental Agreements then in progress, but not then completed, until such Supplemental Agreements terminate or another Master Agreement is entered into by Client and Manatron.

- 12.2 <u>Defaults</u>. The following events are deemed to be defaults:
- (a) A party committing a material breach of any term of this Master Agreement or any Supplemental Agreement, if such breach has not been cured within thirty days after written notice of such breach has been given by the non-defaulting party to the defaulting party;
- (b) A party filing bankruptcy, becoming insolvent, or having its business placed in the hands of a receiver; assignee or trustee, whether by voluntary act or otherwise;
- (c) A party failing to comply in any material respect with any federal, state or local laws applicable to a party's performance under this Master Agreement or any Supplemental Agreement.
- 12.3. Termination for Default. A party may terminate this Master Agreement and the relevant Supplemental Agreement(s) before expiration of their respective term(s) for default by the other party. If default occurs, the parties will have all remedies provided in this Master Agreement and otherwise available by statute, law or equity.
- 12.4. <u>Survival</u>. Any terms of this Agreement, which by their nature extend beyond its termination, remain in effect until fulfilled, and apply to successors and permitted assigns.
- 12.5 Suspension of Performance. If any payment due to Manatron under this Master Agreement or any Supplemental Agreement is past due more than thirty days, Manatron may suspend performance under this Master Agreement and any or all Supplemental Agreements until all amounts due are current.
- 12.6 Fiscal Funding. Manatron or Client may terminate any Supplemental Agreement upon thirty (30) days written notification due to the lack of fiscal funding. Client will be responsible for payment of all labor, costs and expenses incurred by Manatron through the date of the receipt of written notification.

12.7 Termination of Maintenance Services.

- (a) Client may terminate this Master Agreement or any Supplemental Agreement(s) at any time after the first anniversary of the Installation Date by providing at least sixty (60) days prior written notice of termination to Manatron.
- (b) Manatron may not terminate an Annual Software License and Maintenance Supplemental Agreement during the first thirty-six (36) months of paid maintenance. Thereafter, Manatron may terminate Maintenance Services provided to the Client for any Product or Integrated System upon written notice six (6) months prior to termination.

Page 12 of 14

- (c) If Manatron determines that any alterations, attachments, or medifications not made by Manatron will interfere with the provision of Maintenance Services, then Manatron may notify Client of its intention to terminate Maintenance Services. If Client does not cure within thirty (30) days of such notice, Maintenance Services will be terminated.
- (d) Notwithstanding anything in this Master Agreement, if Client is in default of Section 6.2(g), Manatron may terminate Maintenance Services for such Manatron Software or Non-Manatron Software for which Client is in default.

13. DISPUTE RESOLUTION.

- 13.1 <u>Disputes and Demands</u>. The parties will attempt to resolve any claim or controversy related to or arising out of this Master Agreement or Supplemental Agreements, whether in contract or in tort ("Dispute"), on a confidential basis according to the following process, which either party may start by delivering to the other party a written notice describing the dispute and the amount involved ("Demand").
- 13.2 <u>Negotiation and Meditation</u>. After receipt of a Demand, authorized representatives of the parties will meet at a mutually agreed upon time and place to try to resolve the Dispute by negotiation. If the Dispute remains unresolved after this meeting, either party may start mandatory non-binding mediation under the commercial mediation rules of the American Arbitration Association ("AAA").
- 13.3 Injunctive Relief. Notwithstanding the provisions of Section 12.2(a) and this Section 13, if either party seeks injunctive relief, such relief may be sought in a court of competent jurisdiction without the requirement for prior notice and opportunity to cure under Section 12.2(a) and without complying with the negotiation and mediation provisions of this Section 13.
- 13.4 <u>Time Limit</u>. Neither mediation under this section nor any legal action, regardless of its form, related to or arising out of this Master Agreement or Supplemental Agreements may be brought more than four (4) years after the cause of action first accrued or within the statutory limits prescribed by law.

14. GENERAL PROVISIONS:

14.1 Entire Agreement. This Master Agreement, the Supplemental Agreements and the attachments and exhibits thereto are the entire agreement and supersede all prior negotiations and oral agreements. Manatron has made no representations or warranties with respect to this Master Agreement or the Supplemental Agreements that are not included herein or therein. This Master Agreement and the Supplemental Agreements may not be amended or waived except in writing signed by an officer of the party to be bound thereby. If any conflict exists between the terms of this Master Agreement and any Supplemental Agreement, the terms of the Supplemental Agreement will control.

14.2 Preprinted Forms. The use of preprinted forms, such as purchase orders or acknowledgments, in connection with this Master Agreement and the Supplemental Agreements is for convenience only and all preprinted terms and conditions The terms of this stated thereon are void and of no effect. Master Agreement, including but not limited to this Section 14.2, and the terms of Supplemental Agreements cannot be amended, modified or altered by any conflicting preprinted terms, provisions or conditions contained in a preprinted form, such as purchase orders or acknowledgements. If any conflict exists between this Master Agreement or Supplemental Agreements and any terms and conditions on a purchase order, acknowledgment or other preprinted form, the terms and conditions of this Master Agreement and Supplemental Agreements will govern.

14.3 Interpretation. This Master Agreement and the Supplemental Agreements will be construed according to their fair meaning and not for or against either party. Headings are for reference purposes only and are not to be used in construing the Master Agreement or Supplemental Agreements. All words and phrases in this Master Agreement and the Supplemental Agreements are to be construed to include the singular or plural number and the masculine, feminine, or neuter gender as the context requires.

14.4 Governing Law. THIS MASTER AGREEMENT AND THE SUPPLEMENTAL AGREEMENTS WILL BE GOVERNED BY THE LAWS OF THE STATE OF TEXAS, WITHOUT REGARD TO ITS CONFLICT OF LAWS PROVISIONS, UNLESS CLIENT IS A GOVERNMENTAL SUBDIVISION OF ANOTHER STATE, IN WHICH CASE THE LAWS OF THE STATE IN WHICH CLIENT IS A GOVERNMENTAL SUBDIVISION WILL CONTROL.

14.5 Severability. Whenever possible, each provision of this Master Agreement and the Supplemental Agreements will be interpreted to be effective and valid under applicable law; but if any provision is found to be invalid, illegal or unenforceable, then such provision or portion thereof will be modified to the extent necessary to render it legal, valid and enforceable and have the intent and economic effect as close as possible to the invalid, illegal or unenforceable provision. If it is not possible to modify the provision will be severed from the rest of the Master Agreement or Supplemental Agreement, as applicable, and ignored. The invalidity, illegality or unenforceability of any provision will not affect the validity, legality or enforceability of any other provision of this Agreement, which will remain valid and binding.

14.6 Delays. Manatron is not responsible for failure to fulfill its obligations when due to causes beyond its reasonable control, including the failure of third parties to timely provide Software, Hardware, Services, materials or labor contemplated herein or in any Supplemental Agreement(s). Manatron will notify Client in writing of any such delay, and the time for Manatron's performance will be extended for a period corresponding to the delay. Manatron

and Client will determine alternative procedures to minimize project delays.

14.7 Force Majeure. "Force Majeure" means a delay encountered by a party in the performance of its obligations under this Master Agreement or Supplemental Agreements which is caused by an event beyond the reasonable control of the party, but does not include any delays in the payment of monies due by either party. Without limiting the generality of the foregoing, Force Majeure includes but is not restricted to the following types of events: acts of God or public enemy; acts of governmental or regulatory authorities (other than the Client and its governing entities); fires, floods, epidemics or serious accidents, unusually severe weather conditions; strikes, lockouts, or other labor disputes. If any event constituting Force Majeure occurs, the affected party shall notify the other party in writing, disclosing the estimated length of the delay, and the cause of the delay. If a Force Majeure occurs, the affected party will not be deemed to have violated its obligations under this Master Agreement or any Supplemental Agreements, and time for performance of any obligations of that party will be extended by a period of time necessary to overcome the effects of the Force

14.8 Compliance with Laws. Client and Manatron shall comply with all federal, state and local laws in the performance of this Master Agreement and the Supplemental Agreements, including those governing use of the Hardware, Software and Integrated Systems. Products provided under this Master Agreement may be subject to U.S. and other government export control regulations. Client shall not export or re-export any Products.

14.9 <u>Assignments</u>. Neither party may assign its interest in this Master Agreement or the Supplemental Agreements without the prior written consent of the other, such consent not to be unreasonably denied.

14.10 Independent Contractors. Client and Manatron are independent contractors and are not agents or partners of each other. Manatron's employees, agents and subcontractors will not be entitled to any privileges or benefits of Client employment. Client's employees, agents and contractors will not be entitled to any privileges or benefits of Manatron employment.

14.11 Notices. Any notice required or permitted to be given under this Agreement by one party to the other must be in writing and shall be given and deemed to have been given immediately if delivered in person to the address set forth on the Signature Page for the party to whom the notice is given, or on the fifth (5th) business day following mailing if placed in the United States Mail, postage prepaid, by registered or certified mail with return receipt requested, addressed to the party at their address set forth on the Signature Page. Each party may change its address for notice by giving written notice of the change to the other party.

Manamon, Inc. Master Agreement

Sig	natur	e Page
-----	-------	--------

		1 የሚያለስለስ
		1 Day Master A greenient Nimiber: FBUUUA
Agreement Number:		Re: Master Agreement Number: FB0002
A otenneni promovi	•	1
1 KEY OWELL THE STATE OF THE ST	1	

RECORDS MANAGEMENT ANNUAL SOFTWARE LICENSE

AND MAINTENANCE SUPPLEMENTAL AGREEMENT

This Records Management Annual Software License and Maintenance Supplemental Agreement ("Supplemental Agreement") is entered into by and between Manatron, Inc. ("Manatron"), a Michigan corporation authorized to do business in the State of Texas, and Fort Bend County ("Client"), a political subdivision of the State of Texas. Manatron and Client have entered into a Records Management Master Agreement (the "Master Agreement") number FB0002. This Supplemental Agreement is entered into under the terms of the Master Agreement and constitutes a "Supplemental Agreement" as defined in the Master Agreement. The terms of the Master Agreement are incorporated herein by reference and are an integral part of this Supplemental Agreement. The Master Agreement and this Supplemental Agreement constitute the entire agreement between Manatron and Client with respect to the subject matter of this Supplemental Agreement.

The following Exhibits are attached to this Supplemental Agreement and made a part hereof:

Exhibit A Pricing and Inventory

Exhibit B Description of Maintenance Services

Exhibit C Client Service Request Form

Exhibit D Change Request Form

Exhibit E Non-Manatron Sublicensed Software License Agreements

2007. Client acknowledges it has read and The date of this Supplemental Agreement is October 9 understands the Master Agreement and this Supplemental Agreement (including all exhibits, schedules and amendments).

Agreed and Accepted:

Client

Name: Address: Fort Bend County, Texas

Primary Phone:

Facsimile:

Executed by:

Name: Title:

201 Jackson Street, Suite 719

Richmond, Texas 77469

541-8608 -341-8609

Robert E. Hebert County Judge_

Manatron

Manaron, Inc.

510 East Milham Avenue Portage, Michigan 49002

John R. Hansen

Director of Risk Management

This Agreement is not effective until executed by both parties.

ATTEST:

Dianne Wilson, County Clerk

Page 1 of 12

Manatron, Inc. Annual Software License and Maintenance Agreement

1. DEFINITIONS.

Defined terms used in this Supplemental Agreement will have the same meaning given to such terms in the Master Agreement, except to the extent modified or otherwise defined herein. As used in this Supplemental Agreement, the following additional definitions apply:

- 1.1 "Anniversary Date" means each anniversary of the Installation Date.
- 1.2 "Annual Fee" means the combined annual license, sublicense and Maintenance Services fees payable by Client to Manatron as described in Section 4.
- 1.3 "Maintenance Services" means the level of maintenance service selected (Level One, Basic Level of Service (Mandatory); Level Two, Extended Service (Optional); or, Level Three, Extended Service (Optional)) in Exhibit A (Pricing and Inventory) of this Supplemental Agreement that Manatron will provide to Client under this Supplemental Agreement. The Maintenance Services are more specifically described in Exhibit B, Description of Maintenance Services.
- 1.4 "Maintenance Phase Start Date" means the date the Client signs off on the final Project Acceptance Criteria. This final sign-off begins the Maintenance Phase of the Agreement.

2. SOFTWARE LICENSES AND SUBLICENSES.

- 2.1 Manatron Proprietary Software License. The Manatron Proprietary Software licensed to Client under this Supplemental Agreement is identified in Section 2.1 (Manatron Proprietary Software) of Exhibit A (Pricing and Inventory). The Number of Licensed Users, Licensed Server and Licensed Location with respect to each item of Manatron Proprietary Software are also specified in Exhibit A. The Manatron Proprietary Software is licensed to Client on the terms and conditions set forth in the Master Agreement and this Supplemental Agreement. Client is permitted to use the Licensed Software only in connection with the Integrated System with which the Manatron Proprietary Software is provided. 2.2 Non-Manatron Sublicensed Software.
- (a) The Non-Manatron Sublicensed Software sublicensed to Client under this Supplemental Agreement is identified in Section 2.2 (Non-Manatron Sublicensed Software) of Exhibit A (Pricing and Inventory). The Licensor, Number of Licensed Users, Licensed Server and Licensed Location with respect to each item of Non-Manatron Sublicensed Software are also specified in Exhibit A. The Non-Manatron Sublicensed Software is sublicensed to Client on the terms and conditions set forth in the Muster Agreement and this Supplemental Agreement. Client is permitted to use the Non-Manatron Sublicensed Software only in connection with the Integrated System with which the

Non-Manatron Sublicensed Software is provided. Client acknowledges that this sublicense is subject to the terms of the Licensors' respective license agreements for the Non-Manatron Sublicensed Software, which are incorporated heroin by reference, and attached hereto as Exhibit E. The Licensor(s) of the Non-Manatron Sublicensed Software are third party beneficiaries of the sublicense terms of the Master Agreement and this Supplemental Agreement to the extent permitted by applicable law.

- 2.3 Non-Manatron Other Software Not Under This Agreement. The Non-Manatron Other Software identified in Section 2.3 (Non-Manatron Other Software) of Exhibit A (Pricing and Inventory) is not included in or covered by this Supplemental Agreement. The Non-Manatron Other Software is listed in Exhibit A solely as a matter of record keeping convenience and to identify the Integrated System with which the Manatron Proprietary Software and Non-Manatron Sublicensed Software may be used. If Client has any rights with respect to the Non-Manatron Other Software, such rights would be under a separate agreement with Manatron or the Licensor of such software.
- 2.4 Hardware Not Under This Agreement. The Hardware identified in Exhibit A (Pricing and Inventory) is not included in or covered by this Supplemental Agreement. The Hardware is listed in Exhibit A solely as a record keeping convenience and to identify the Integrated System with which the Manstron Proprietary Software and Non-Manatron Sublicensed Software may be used. If Client has any rights with respect to the Hardware, such rights would be under a separate agreement with Manatron or the manufacturer or supplier of such Hardware.

3. MAINTENANCE SERVICES.

3.1 First Year's Maintenance Services at No Extra Charge. Manatron will provide to Client Maintenance Services for no extra charge for one (1) year beginning with the Installation Date on the terms and conditions set forth in Section 9 of the Master Agreement. The Maintenance Services to be provided during this one year (1-year) period are described in Section 3 of Exhibit B (Description of Maintenance Services) as Level ONE, Basic Level of Service. The Maintenance Services described in Section 3 of Exhibit B as Level ONE, Basic Level of Service constitute the complete list of services provided by Manatron during the first year, unless Client elects to pay for a higher level of Maintenance Services, which payment will be due on the Installation Date. If Client pays Manatron on or before the Installation Date for a Level Two or Level Three level of service during the first year, Manatron will provide Cliem during such year the level of service paid for by Client. Manatron will provide one (1) week of on-site support upon commencement of the first year's Maintenance Services.

- 3.2 Selection of Maintenance Service Level. Pursuant to the terms and conditions set forth in the Master Agreement and this Supplemental Agreement, if the term of this Supplemental Agreement renews as provided in Section 5.1(b), beginning with the first Anniversary Date Manatron will provide Client with Maintenance Services at the level selected and paid for by Client (Level One, Level Two, or Level Three) as described in Exhibit B, upon payment of the Annual Fee. The Maintenance Services as described in Exhibit B for the level selected and paid for by Client constitute the complete list of maintenance and support to be provided by Manatron. Manatron will not be obligated to provide any Maintenance Services not described in the level of Maintenance Services selected and paid for by Client.
- 3.3 Additional Services on Time and Materials Basis. If Manatron with Client's prior written approval, provides maintenance and support or other services requested by Client that are not covered by the level of Maintenance Services selected and paid for by Client, Client shall pay Manatron for all such maintenance, support and services on a time and materials basis, plus expenses, at Manatron's then prevailing rates, unless otherwise agreed in writing by Manatron and Client.

4. ANNUAL FEE.

- 4.1 Date Payable. An Annual Fee is due and payable by Client to Manatron on each Anniversary Date of this Agreement. Client must pay each invoiced Annual Fee by the due date.
- 4.2 Invoicing. Manatron will invoice Client for the Annual Fee at least ninety calendar days before the Anniversary Date on which the Annual Fee is due. Manatron may periodically review and adjust the amount of the Annual Fee, subject to the terms of Section 5.5 of the Master Agreement. Manatron will notify Client of any changes in the Annual Fee with the invoice.
- 4.3 <u>Nonrenewal</u>. If Client fails to timely pay the Annual Fee, this Supplemental Agreement will not automatically renew in accordance with Section 5.1(b) of this Agreement.
- 5. TERM OF AGREEMENT; TERMINATION.

5.1 Term

(a) The initial term of this Supplemental Agreement will commence on the date of this Supplemental Agreement, with the licenses and sublicenses granted herein becoming effective on the

- date of this Supplemental Agreement. This Supplemental Agreement and the licenses and sublicenses granted herein will automatically terminate on the first Anniversary Date unless renewed as provided in Section 5.1(b).
- (b) This Supplemental Agreement will renew for a one year renewal term on each Anniversary Date if Manatron has received, on or before such Anniversary Date, payment of the invoiced Annual Fee due on such Anniversary Date. Unless this Supplemental Agreement renews at the end of a renewal term as provided in the previous sentence, this Supplemental Agreement and the licenses and sublicenses granted herein will automatically terminate upon the end of such renewal term.
- 5.2 Termination. This Agreement may be terminated as provided in Section 12 of the Master Agreement. Maintenance and support services under this Agreement may be terminated as provided for in Section 12.7 of the Master Agreement.
- 5.3 Effect of Termination. Upon termination or cancellation of this Supplemental Agreement, Client shall return to Manatron or destroy all Manatron Proprietary Software, Non-Manatron Sublicensed Software and Proprietary and Confidential Information in accordance with Sections 3.3(c) and 7.2 of the Master Agreement.

6. LIMITED WARRANTY TERMS.

THE DISCLAIMERS SET FORTH IN SECTION 9 OF THE MASTER AGREEMENT ARE INCORPORATED HEREIN BY REFERENCE.

7. Limitation of Damages.

THE LIMITATION OF DAMAGES SET FORTH IN SECTION 10 OF THE MASTER AGREEMENT ARE INCORPORATED HEREIN BY REFERENCE.

8. CHANGE REQUESTS:

Any amendment of this Supplemental Agreement must follow the Change Request procedure stated in Section 2.4 of the Master Agreement.

Manatron, Inc.

Annual Software License and Maintenance Agreement

AUDITOR'S CERTIFICATE

I hereby certify that funds in the amount of \$92,010.00 are available to pay the obligation of fort bend county within the foregoing agreement.

Robert Ed Stardivant, County Auditor

(The rest of this page is intentionally left blank.)

Manatron, Inc. Exhibit A - Pricing and Inventory

1. ANNUAL FEE

1.1 License Fees and Basic Level ONE Service - Mandatory

Client shall pay Manatron the fees set forth below for the license of Manatron Proprietary Software, sublicense of Non-Manatron Sublicensed Software and Manatron's Level ONE Basic Level of Service:

Software License and Sublicense Fees plus Maintenance Level ONE Basic Level of Service Anthem™ Licenses (\$16,880.00) Anthem™ Official Public Records (also includes UCC, Assumed Names, Export/Archive, and Cash Management Reporting) — (105 users) - Anthem™ Vitals for Birth and Death - Anthem™ Maprilat - Anthem™ Maprilat - Anthem™ Commissioners Court - Anthem™ Courts (for Inventle, Criminal, Civil & Probate) - Anthem™ Public Access (local and internet) - Imagen™ Licenses (\$27,365.00) - Web Support (\$15,190.00) - Anthem™ Recording (\$12,700.00) - Anthem™ Toll Road (\$300.00) - Anthem™ Toll Road (\$300.00) - Anthem™ Texas Online Support (\$700.00) - Lead Tools ImageViewer (105 users included) - Java Viewer	Ceuzed 2011/Ame and parameters are	Total Price	
Maintenance Level ONE Basic Level of Service Anthem™ Licenses (\$16,880.00) Anthem™ Official Public Records (also includes UCC, Assumed Names, Export/Archive, and Cash Management Reporting) — (105 users) - Anthem™ Vitals for Birth and Death - Anthem™ Marriage Licenses - Anthem™ Map/Plat - Advanced Backfile - Anthem™ Commissioners Court - Anthem™ Courts (for Juvenile, Criminal, Civil & Probate) - Anthem™ Public Access (local and internet) - Imagen™ Licenses (\$27,365.00) - Web Support (\$15,190.00) - Anthem™ Automated Indexing (\$18,375.00) - Anthem™ Automated Indexing (\$18,375.00) - Anthem™ Toll Road (\$800.00) - Anthem™ Tolls Road (\$800.00) - Lead Tools ImageViewer (105 users included) - Java Viewer	Service Option	\$92,010.00	
Anthem TM Official Public Records (also includes UCC, Assumed Names, Export/Archive, and Cash Management Reporting) — (10S users) - Anthem TM Vitals for Birth and Death - Anthem TM Marriage Licenses - Anthem TM Map/Plat - Advanced Backfile - Anthem TM Counts (for Invenile, Criminal, Civil & Probate) - Anthem TM Courts (for Invenile, Criminal, Civil & Probate) - Anthem TM Public Access (local and internet) - Imagen TM Licenses (\$27,365.00) - Web Support (\$15,190.00) - Anthem TM eRecording (\$12,700.00) - Anthem TM Automated Indexing (\$18,375.00) - Anthem TM Toll Road (\$800.00) - Anthem TM Texas Online Support (\$700.00) - Lead Tools Image Viewer (105 users included) - Java Viewer	Software License and Sublicense rees plus		
Anthem™ Official Public Records (also includes UCC, Assumed Names, Export/Archive, and Cash Management Reporting) — (105 users) - Anthem™ Vitals for Birth and Death - Anthem™ Marriage Licenses Anthem™ Map/Plat - Advanced Backfile - Anthem™ Courts (for Invenile, Criminal, Civil & Probate) - Anthem™ Public Access (local and internet) - Imagen™ Licenses (\$27,365.00) - Web Support (\$15,190.00) - Anthem™ eRecording (\$12,700.00) - Anthem™ Automated Indexing (\$18,375.00) - Anthem™ Toll Road (\$800.00) - Anthem™ Texas Online Support (\$700.00) - Lead Tools ImageViewer (105 users included) - Java Viewer	Maintenance Level ONE Basic Level of Service		
	Anthem TM Licenses (\$16,880.00) Anthem TM Official Public Records (also includes UCC, Assumed Names, Export/Archive, and Cash Management Reporting) — (105 users) - Anthem TM Vitals for Birth and Death - Anthem TM Marriage Licenses - Anthem TM Map/Plat - Advanced Backfile - Anthem TM Counts (for Juvenile, Criminal, Civil & Probate) - Anthem TM Courts (for Juvenile, Criminal, Civil & Probate) - Anthem TM Public Access (local and internet) - Imagen TM Licenses (\$27,365.00) - Web Support (\$15,190.00) - Anthem TM eRecording (\$12,700.00) - Anthem TM Automated Indexing (\$18,375.00) - Anthem TM Toll Road (\$800.00) - Anthem TM Texas Online Support (\$700.00)		The second secon
Total for Level ONE Service (10/1/07 – 9/30/08)	• 1849 Alexei		
A WANTED TO THE PARTY OF THE PA	Total for Level ONE Service (10/1/07 - 9/30/08)		

1.2 Extended levels (optional - select one)

Client shall pay Manatron the fees set forth below for extended levels of service selected by Client:

int small pay intananon me account	Check one	Additional Price	
Optional - Select One	Tolle City (31)	The state of the s	
Mandatory fees above plus Level TWO -			
Extended Service (On site 2x per year)			ĺ
Mandatory fees above plus Level THREE			
Extended Service (On site 4x per year)		**************************************	81.

1.3 Total Annual Fee

The Annual Fee consists of the sum of the fees in Section 1.1 and 1.2 above. Manatron will bill Client quarterly for the Annual Fee. Manatron may periodically adjust the amount of the Annual Fee, subject to the terms of Section 5.5 of the Master

2. LICENSED SOFTWARE INVENTORY

2.1 Manatron Proprietary Software

Z. I Manatron Frontistary South	
	Version: 6.8.x
Licensed Software: - Anthem Official Public Records (also includes UCC, Assumed Names,	1
- Anthem Official Public Records (also includes octo, Assumed	Providence of the Control of the Con
Export/Archive and Cash Management Reporting)	ply the state of t
- Anthem Vitals for Birth and Death	
- Anthem Marriage Licenses	1
- Withen Marked Disame	
- Authem Map/Plat	
- Advanced Backfile	
The state of the s	

Manatron, Inc., Annual Software License and Maintenance Agreement Exhibit A - Pricing and Inventory

- Anthem Commissioners Court - Anthem Courts (for Juvenile, Criminal, Civil & Probate) - Anthem Toll Road	
Licensed Server: (Central Processing Unit) Anthem production server Licensed Location: Fort Bend County Number of Licensed Users: 105	
Ticensed Software:	Version: 6.8.x

ì	Licensed Schware:		
1	- Anthem TM Public Access (local and internet)	Ì	
	- Anthen TM eRecording		
	- Anthem TM Texas Online		
-	· ·		
	Licensed Server: (Central Processing Unit) Anthem™ PA internet on Anthem web	server, Anthem	
	eRecording and Anthem TX Online on Anthem eRecording server	and the second s	-
	Licensed Location: Fort Bend County	A STATE OF THE STA	
	Number of Licensed Users: unlimited		
	Number of Licensey Osers, unminted	· · · · · · · · · · · · · · · · · · ·	

Licensed Software: - Anthem™ Automated Indexing	Version: 6.8.x
Licensed Server: (Central Processing Unit) Anthem AI server	
Licensed Location: Fort Bend County Number of Licensed Users: 6 users plus 1 supervisor	

2.2 Non-Manatron Sublicensed Software

(a) Database Software

Licensed Software	Version
Licensed Server (Central processing Unit)	
Licensed Location	
Number of Licensed Users	

(b) Image Software

web server

Licensed Software: Lead Tools ImageViewer	Version:
Licensed Server: (Central Processing Unit) Fort Bend County Clerk PC workstal	ions
Licensed Location: Fort Bend County	
Number of Licensed Users: 105	
Alleman and the second and the secon	*1
Licensed Software: Java Viewer	Version:
Licensed Software: Java viewer Licensed Server: (Central Processing Unit) Fort Bend County Clerk Anthem	production server and Admient

2.3 Non-Manatron O	ther Software
--------------------	---------------

Licensed Location: Fort Bend County

Number of Licensed Users: 2 processor licenses

(a) Shrink-wran software

	(a) Sill lik. Mich sommer		
į	1 1 d C. Banner Name	Version	Quantity of Licenses
	Licensed Software Name		

Manatron, Inc., Annual Software License and Maintenance Agreement Exhibit A - Pricing and Inventory

(b) Operating systems shipped with hardware

(i) Desktop operating systems

(i) Desktop operating systems		
	Version	Quantity of Licenses
Licensed Software Name		,
	., <u></u>	
(ii) Server operating systems		
	I Vareiari	Touantity of Licenses
Licensed Software Name	I. A CI PIOT	1 Kneetter) or see the see of the
L10011040		

Quantity	Onantity
Manatrou, Inc., Annual Software License and Maintenance Agreement Schibit A – Pricing and Inventory Hardware Purchased From Manatron: Model Number Description Subsystem Component Model Number Description	3.2 Hardware Purchased Directly From Hardware Manufacturer or Other Supplier. Subsystem Component Model Model Number Description

Manatron, Inc. Exhibit B - Description of Maintenance Services

Manatron, Inc. Exhibit C - Client Service Request Form

\$ \$ 1 m

CLIENT SERVICE REQUEST FORM

(FAX # 800-396-4278) (HELP DESK # 800-750-4278)

T 2727 A 7716 387-			(CLIEN	T) REGISTER OF DEEDS
DATE	IME:	[A.	м./Р.М.)	
LOCATION:T DATE:T NAME OF USER/DEPT, HAVING TI	HE PRO	BLEM:		the state of the s
NAME OF USER/DEPT, HAVING TO PHONE NUMBER: SEVERITY PRO	******************	onniger segmented dest dest regional régissa consentation de region conceptant e conceptant des registres des	per contract	
SEVERITY PRI	ORITY	Ole. ASAP, HIGH, ROUTINE, FYT);	
If Software - Check one of the following:				
DOCUMENT RECEPTION INDEXING REPORTS MARRIAGE SYSTEM ADM.	0	SCANNING CODE MAINT/FEE SETUP UCC PUBLIC OTHER		ENHANCEMENT
If Hardware - Check one of the following:	: -			
U HP LASER PRINTER U LABEL PRINTER U RECEIPT PRINTER U OTHER (specify)	0	SCANNER PC CASH DRAWER	0 0 0	MONITOR SERVER
to the problem.)	**************************************			and the state of t
Resolution: (i.e. who was contacted, situa		resolved in what way, etc.)		

			······································	
				MMC 15
				Management (1985)
		A CONTRACTOR OF THE PARTY OF TH		wywi-webi
				na marana y ni 199 0). In la const ensión del costo persona constante de la costo della costo della costo della costo de la costo della costo della costo della costo della co
		and the second s	***************************************	A STATE OF THE STA

* - * - A

Manatron, Inc. Exhibit D – Change Request Form

CHANGE REQUEST FORM

Part I - Request

Check one)	
☐ Initial Project Requirement ☐ Project Scope Change ☐ Post-Acceptance Enhancement	
Post-Acceptance Enhancement	
Project Name:	The state of the s
Product / Release / Version:	Received Date:
Request Number:	Phone:
Request Originator: Name:	Emails
Title / Role	A. (1.) 16.11
Solution area / module / component:	Requirement Priority:
Requirement Severity:	ACCEPTED AND ACCEPTED ACCEPTED AND ACCEPTED ACCEPTED AND ACCEPTED ACCEPTED AND ACCEPTED AND ACCEPTED AND ACCEPTED AND ACCEPTED AND ACCEPTED ACCEPTED ACCEPTED AND ACCEPTED
Request Abstract or Title:	the state of the s
The second secon	
the same of the sa	
and the same of th	
	1 1
Dequest description: (Objective to be satisfied	ed or issue to be resolved. Please attach any relevant examples.)
CCI UCA: Market Property and Commission of the C	
the state of the s	
	the state of the s
	That we the banefite?
Justification: (What is the significance of the	is request? What are the benefits?)
	14 - 14 - 14 - 14 - 14 - 14 - 14 - 14 -
Name of the Control o	
OPTIONAL Solution Proposal (Used to cl	larify issue description)
OPTIONAL Solution Proposal (Osed to of	10.111 15340 A1504 1544
and the second s	
· · · · · · · · · · · · · · · · · · ·	
appearance of the same of the	The state of the s
Implementation Acceptance Criteria: (W.	hat will you use as your acceptance criteria if this change of scope is
implemented?)	
ISLUANITY CONTROL OF THE PROPERTY OF THE PROPE	
Marie Control of the	
	A STATE OF THE STA
Harry State of the	
4.4444.4444.4444.4444.4444.4444.4444.4444	

ş 1 §

Manatron, Inc., Annual Software License and Maintenance Agreement Exhibit D - Change Request Form

Part 2 - Response

ם ם ם	(One of the following. Also shown as status once decision is made) Proposal created, implementation not proposed (Cost) = The cost of implementing this request would far exceed the value to be added to the solution implementation not proposed (Conflict) = The implementation would conflict with current or future fundions or objectives implementation not proposed (Conflict) = The issue to be addressed, and/or the implementation for this request is outside triplementation not proposed (Domain) = The issue to be addressed, and/or the implementation for this request is outside the domain of this solution. It is proposed solution: (Detailed implementation proposal attached)

'stima'	ted cost to modify the solution: (HW/SW infrastructure, solution documentation, training materials,
aining	, and support requirements)
Sinteres and a second	
,	
rss e/j/ *****************	
	
*525	ated schedule impact to accommodate the solution:
ESTI III	HEC SCHOOL See See See See See See See See See Se

· ·	tron response approval:
Mana	
Mana Signa Title:	
Signa Title:	nents:
Signa Title: Com	ments:
Signa Title: Com ———————————————————————————————————	ments: nt acceptance of change to project scope, schedule, and cost:
Signa Title: Com Clier Sign	nt acceptance of change to project scope, schedule, and cost: ature: Phone:
Signa Title: Com Clier Sign	nt acceptance of change to project scope, schedule, and cost: ature: Phone:
Signa Title: Comm Clier Signa Title Purc	nt acceptance of change to project scope, schedule, and cost: ature: Phone:

1. PURPOSES AND SCOPE.

* 1, 1

This Service Level Agreement sets forth the terms of the maintenance services available to Client from Manatron, Inc. ("Manatron"), a Michigan corporation authorized to do business in the State of Texas. Manatron will perform the maintenance services selected in and on the terms and conditions set forth in the Records Management Master Agreement and the Records Management Annual Software License and Maintenance Supplemental Agreement between Manatron and Client (the "License Agreements") and as described herein.

2. AVAILABLE SERVICES.

Manatron offers three (3) levels of maintenance services.

- 2.1 Level ONE. Basic Level of Service. This is the most basic level of maintenance services available to Client under this Service Level Agreement and is further described in Section 3, Level ONE, Basic Level of Service.
- 2.2 Level TWO, Optional Extended Support. Manatron provides optional on-site maintenance services. Maintenance services under Level TWO, Optional Extended Support, include all maintenance services under Level ONE, Basic Level of Service, and the additional maintenance services described in Section 4. Level TWO, Optional Extended Support.
- 2.3 Level THREE, Optional Extended Support. Manatron provides optional on-site maintenance services. Maintenance services under Level THREE Optional Extended Support, include all maintenance services under Level ONE, Basic Level of Service, and the additional maintenance services described in Section 5, Level THREE, Optional Extended Support.

3. LEVEL ONE - BASIC LEVEL OF SERVICE.

- 3.1 Base Mandatory Services. Level ONE, <u>Basic Level of Service</u>, is included in the Annual License/Maintenance/Support Fees and Annual Sublicense/Maintenance Fees payable pursuant to the License Agreements. During the term of the License Agreements, Manatron will provide Level ONE, <u>Basic Level of Service</u>, with respect to the Manatron Proprietary Software and Sublicensed Software.
- 3.2 Project and Support Manager. Manatron will designate a Project and Support Manager ("PSMPSM"). The PSM will maintain close contact with Client through frequent communication. The PSM will be responsible for managing delivery of the maintenance services.
- 3.3 <u>Client Support Center.</u> The Client Support Center ("CSC") is the primary point of Client contact

for all support. CSC consultants provide responses to support requests received from system users and system administration personnel. When initiating a support request, Client should communicate to the CSC the information in the Client Service Request ("CSR")

- (a) The primary means of contacting Manatron's CSC during normal operating hours is via telephone through the toll-free client support line. Outside of normal operating hours or if all CSC consultants are busy, the client support line will prompt callers to leave a voice mail message that will, in turn, acrivate a page to a CSC consultant.
- (b) A dedicated, toll-free client support fax line is available 24 hours a day, 7 days a week, as is e-mail access.
- 3.4 CSC Hours of Operation. Normal operating hours for the CSC are 7:00 AM to 7:00 PM Central Time, Monday through Friday, except for Manatron company holidays.

3.5 CSC Response Goals

(a) Upon receipt of a CSR, a CSC consultant will review the information and assign a severity for urgency of response according to the following list:

*	
Severity	Type of Problem/Request
1/Critical	A system-wide problem, one that
	prevents the recorder's office from
	continuing fundamental business
	processes. Some examples might be the
	system servers being down, users unable
	to record documents, unable to view
	images on the Clerk system, etc.
2/High	A problem that affects one or more
	modules of the Manatron system. A
	problem that prevents the recorder's
	office from performing an important
	function of the office's normal business
	processes.
3/Medium	System feature or minor hardware is
	malfunctioning or inoperative, but a
	alternative procedure exists to achieve
	business needs. A problem that impacts
	individual users or workstations.
	Examples would include receipts
	requiring adjustment, users receiving
	error messages that do not otherwise
	prevent business activities, data
	corrections, etc.
4/Low	The "Low" category includes cosmetic
	issues such as misspellings, parts of
	letters falling off the screen or report
	print outs, incorrect punctuation, etc.
	"Low" also includes problems that
	happen intermittently, for which root
	causes are being determined or which
	cannot be reproduced. This category is

Wersion 01-06-06

Page 1 of 6

also used to characterize information requests.

(b) A CSC consultant will communicate to Client a Response based upon the severity of the problem. "Response" is defined as a communication with Client of the status of problem, analysis or potential remedies, or workarounds. The Response goals for a CSR received during normal working hours are shown in the following table:

Severity ::	Response Goal Within 1 hour
2/High	Within 4 hours
3/Medium	Within 2 business days
4/1.ow	Assumption is these will be fixed in
	the next Maintenance Release.

(c) Responses to a CSR received via voice mail/pager, fax, or e-mail during other than normal operating hours may be delayed unless previous arrangements have been made for standby support resources.

3.6 CSC Request Escalation.

* 1 . *

- (a) Upon receipt of a Severity 1/Critical CSR, the CSC manager will be notified to insure that appropriate Manatron resources are focused on returning the affected system to operation as soon as possible.
- (b) A severity 2/High CSR not resolved within eight (8) hours of notification to the CSC will be escalated for assistance by other subject matter experts or Manatron functional area supervisor/manager to determine next steps.
- (c) Client will be notified of the current status and projected closure target on each unresolved CSR, which will be tracked and reported until resolved.
- 3.7 Remote Diagnostics. The CSC consultant, subject matter expert, PSM, or other Client support personnel may utilize remote dial-in capability to assist with system diagnosis and/or corrective action. Client direct participation may or may not be required during remote dial-in operations. However, in either case, all use of remote dial-in capability will be coordinated with the Client in advance.
- 3.8 <u>Supplements and Custom Programming are Excluded.</u>
- (a) From time to time, Manatron may make available computer programs that are compatible with the Manatron Proprietary Software and that supplement the Manatron Proprietary Software. Also, third parties may make available computer programs that are

compatible with the Sublicensed Software and that supplement the Sublicensed Software. Supplements do not include programs necessary for the proper functioning of the Hardware, Software, or Integrated System. SUPPLEMENTS ARE NOT LICENSED OR UNDER LICENSE THE SUBLICENSED AGREEMENTS AND WILL NOT BE PROVIDED WITH MAINTENANCE RELEASES. Subject to availability and compatibility, Client may license or sublicense supplements by written amendment to such License Agreements. All licenses and sublicenses of supplements will include additional charges.

- (b) Maintenance services do <u>not</u> include custom programming.
- 3.9 Manatron Proprietary Software Maintenance and Support. The terms of this section apply to maintenance of Manatron Proprietary Software.
- (a) Client's designated PSM will manage delivery of Manatron Proprietary Software maintenance releases or updates in accordance with the provisions of the applicable License Agreements and this Description of Maintenance Services.
- (b) Maintenance releases will be deployed on an "as-required" basis as determined by Manatron. Maintenance releases for Manatron interface programs and/or supplementary applications, that are not part of the main application (e.g., interfaces with mainframe programs, index or image conversion programs, export programs, etc.), will also be developed and deployed on an "as required" basis as determined by Manatron.
- (c) Manatron may include, at its sole discretion, in its maintenance releases, software modifications, and enhancements, which enhance the functionality of the software.
- (d) Release notes will be provided simultaneously with delivery of the release to Client to include all issues and corresponding resolutions contained in the maintenance release.
- (e) Client may submit recommended software application enhancements to be considered for inclusion in future software maintenance releases.
- (f) Manatron reserves the right to decline acceptance of software modifications recommended or requested by Client. Manatron also reserves the right to determine the conditions under which approved modifications will be delivered.
- (g) Manatron Proprietary Software maintenance includes the correction of material defects, malfunctions, or failures that result in the Manatron Proprietary Software failing to perform substantially according to the performance specifications provided by

Manatron when used properly under normal use and conditions.

v ; *

- (i) Client shall fully inform Manatron immediately of any such defects, malfunctions or failures. [Upon receipt of such notice, Manatron will make best commercially reasonable efforts to fix or replace the Manatron Proprietary Software or provide a suitable workaround, as herein provided. Manatron will make best commercially reasonable efforts to provide the fix, replacement, or workaround as soon as is reasonably possible, taking into consideration the applicable Severity level.].
- (ii) Client shall provide Manatron with a list of output and any other data, including databases and backup systems, that Manatron reasonably may request to reproduce operating conditions similar to those present when the error occurred.
- (iii) Client shall provide Manatron and its agents access to all Client's facilities, hardware, personnel, and data, physically at the hardware site and, if requested by Manatron, through modem telephone connection, to permit Manatron to perform its maintenance services.
- (h) Manatron Proprietary Software maintenance includes any updates to the Manatron Proprietary Software developed by Manatron Updates consist of any enhancements, corrections, modifications, and additions to the Manatron Proprietary Software. Use of updates with or in place of the Manatron Proprietary Software will be fully governed by and subject to the terms of the applicable License Agreements and this Description of Maintenance Services. Any portion of the Manatron Proprietary Software replaced by updates, and all copies thereof, will be destroyed (with certification of destruction provided to Manatron) or returned to Manatron, at Manatron's option.
- 3.10 Non-Manatron Sublicensed Software Maintenance and Support. The terms of this section apply to maintenance of Sublicensed Software. Manatron does not provide maintenance of any other non-Manatron software.
- (a) Client's designated PSM will manage delivery of Sublicensed Software maintenance releases or updates in accordance with the provisions of the applicable License Agreements and this Description of Maintenance Services.
- (b) Maintenance of Sublicensed Software will be accomplished on an "as required" basis as determined by Manatron and the software licensor.
- (c) Sublicensed Software maintenance will be provided only to the extent offered by the licensor of the Sublicensed Software. Manatron will not be responsible for any software programming with respect to the

Sublicensed Software or for software fixes or replacements except to the extent available from the licensor.

's , e

- (d) Client shall fully inform Manatron immediately of any defects, malfunctions, or failures in the Sublicensed Software. Upon receipt of such notice, Manatron will contact the licensor and seek a fix or replacement of the Sublicensed Software. Client shall provide Manatron and the licensor with a list of output and any other data, including databases and backup systems, that Manatron reasonably may request to reproduce operating conditions similar to those present when the error occurred. Client shall provide Manatron, the Licensor, and their agents access to all Client's facilities, hardware, personnel and data, physically at the hardware site and, if requested by Manatron, through modern telephone connection, to permit Manatron and the licensor to perform the maintenance services.
- (e) If a Sublicensed Software failure occurs, Manatron will make best commercially reasonable efforts to obtain a fix, replacement, or suitable workaround of the Sublicensed Software from the Licensor as soon as is reasonably possible, taking into consideration the applicable Severity level.
- (f) Sublicensed Software maintenance includes any updates to the Sublicensed Software developed by the Licensor and that are made available to the Client. Updates consist of any enhancements, corrections, modifications, and additions to the Sublicensed Software. Use of updates with or in place of the Sublicensed Software will be fully governed by and subject to the terms of the License Agreements and this Service Level Agreement. Any portion of the Sublicensed Software replaced by updates, and all copies thereof, will be destroyed (with destruction certified to Manatron) or returned to Manatron, at Manatron's option.
- 3.11 Exclusions. Manatron will not provide maintenance or support of any hardware or non-Manatron software (unless it is Sublicensed Software or hardware purchased directly by Manatron which is still under manufacturer's warranty).
- 3.12 Other Services. Any additional support and professional services will be under a separate Services Agreement and will be priced and billed as provided for in the Services Agreement, in addition to the Annual Maintenance/ Support fees.
- 4. LEVEL TWO OPTIONAL EXTENDED SUPPORT.
- 4.1 <u>Selection of Optional Service Level</u>. Level TWO, <u>Optional Extended Support</u>, is optional and will only be provided if selected by Client.

- 4.2 Basic Level of Service, Level ONE Included. All of the maintenance services and related terms stated in Section 3 and included in Level ONE, Basic Level of Service, are provided and apply in Level TWO, Optional Extended Support.
- 4.3 Semi-Annual System Tuning and Site Visitation.

 Manatron support technicians will perform the following system tuning and monitoring in conjunction with two semi-annual site visits.

(a) Scope of Work.

1 % 1, %

- (i) Database Maintenance. Evaluate free space, assign additional disk space (as required), review and evaluate log files, and remove unneeded (clean up) log file information.
- (ii) Hardware (Server) Maintenance. Perform system performance diagnostics; review setup and procedures for system backup; run tape drive diagnostics; clean tape drive read-write heads; test restore from backup; check uninterrupted power supply, modem/dial-in capability, heat sink and cooling fan; check all server cable connections and performance of server monitor and input devices; and test PC restore from ghost image stored on server.
- (iii) Install Software Maintenance Releases or Updates. As time and resources permit, install software releases provided pursuant to maintenance services.
- (iv) Documentation. Produce a report of the tasks accomplished and the results achieved/observed.
- (v) Other Services. Any additional support and professional services will be mutually agreed and performed under a separate Service Agreement and will be priced and billed as provided for in the such agreement, in addition to the fees charged for Level TWO, Optional Extended Support.

(b) Scheduling.

- (i) Client's designated PSM will coordinate and schedule system tuning and site visits with Client.
- (ii) The initial site visit should be accomplished within six (6) months of the effective date of an agreement to provide Level TWO maintenance services. The PSM and Client will schedule subsequent semi-annual site visits jointly.
- (iii) When possible, system tuning and site visit work will be accomplished during Client's normal business hours.
 - 4.4 Semi-Annual Report of Client CSC Activities.
- (a) The CSC staff will produce semi-annual reports concerning Client's interaction with the CSC.

(b) Reports will include: number of CSRs initiated by Client, status of CSRs, trends in type of support requested, CSR response and problem resolution performance, and Client satisfaction.

6 j. F

- 5. LEVEL THREE OPTIONAL EXTENDED SUPPORT.
- 5.1 <u>Selection of Optional Service Level</u>. Level THREE, <u>Optional Extended Support</u>, is optional and will only be provided if selected by Client.
- 5.2 Basic Level of Service, Level ONE Included. All of the maintenance services and related terms stated in Section 3 and included in Level ONE, Basic Level of Service, are provided and apply in Level THREE, Optional Extended Support.
- 5.3 Quarterly System Tuning and Site Visit. Manatron support technicians will perform the following system tuning and monitoring in conjunction with four (4) quarterly site visits;

(a) Scope of Work.

- (i) Database Maintenance. Evaluate free space, assign additional disk space (as required), review and evaluate log files, and remove unneeded (clean up) log file information.
- (ii) Hardware (Server) Maintenance. Perform system performance diagnostics; review setup and procedures for system backup; run tape drive diagnostics; clean tape drive read-write heads; test restore from backup; check uninterrupted power supply, modem/dial-in capability, heat sink and cooling fan; check all server cable connections and performance of server monitor and input devices; and test PC restore from ghost image (stored on server).
- (iii) Install Software Maintenance Releases or Updates. As time and resources permit, install software releases provided under this Service Level-Agreement.
- (iv) Documentation. Produce a report of the tasks accomplished and the results achieved/observed.
- (v) Other Services. Any additional support and professional services will be under a separate Services Supplemental Agreement and will be priced and billed as provided for in the Services Supplemental Agreement, in addition to the fees for Level THREE, Optional Extended Support.

(b) Scheduling.

- (i) Client's designated PSM will coordinate and schedule system tuning and site visits with Client.
- (ii) The initial site visit should be accomplished within two (2) months of the effective date of an agreement to provide Level THREE

maintenance services. The PSM and Client will schedule subsequent quarterly site visits jointly.

- (iii) When possible, system runing and site visit work will be accomplished during Client's normal business hours.
 - 5.4 Quarterly Report of Client CSC Activities:
- (a) The CSC staff will produce quarterly reports concerning Client's interaction with the CSC.
- (b) Reports will include: number of CSRs initiated by Client, status of CSRs, trends in type of support requested, CSR response and problem resolution performance, and Client satisfaction.

6. EXCLUSIONS

SECTIONS 3.8 AND 3.11 OF THIS DESCRIPTION OF MAINTENANCE SERVICES AND CERTAIN SECTIONS OF THE LICENSING AGREEMENTS LIMIT THE MAINTENANCE SERVICES TO BE PERFORMED BY MANATRON.

7. CLIENT RESPONSIBILITIES.

- 7.1 Systems Operation. Client retains responsibility for the day-to-day management of the system and software, including the backup system.
- 7.2 Specific Responsibilities. Client is responsible for its obligations under the Licensing Agreements and the following items:
- (a) Client Contact Point ("CCP"). Client will designate, in writing, a primary and at least one (1) alternate Client Contact Point who will serve as the primary interface between Manatron's support team and Client. The responsibilities of the CCP include the following:
- (i) Provide Client contact information and inform Manatron of any changes before they occur.
- (ii) Insure basic troubleshooting and a complete analysis of system problems using internal Client resources prior to referring a problem to Manatron.
- (iii) Before submitting a support request to the CSC, gather and record the information needed to fill out a CSR.
- (iv) Contact the CSC and provide the CSR information and any amplifying data to the CSC consultant.
- (v) Coordinate Client activities required to assist the CSC in resolving the problem.
- (vi) Serve as a liaison and primary point of Client contact for the PSM.

- (vii) Complete Change Request Forms and provide them to the PSM to initiate system or software modifications.
- (viii) Insure a Purchase Order (PO) or other suitable form of Client financial obligation authorization is generated and approved prior to requesting additional support not specifically included in the maintenance service level purchased pursuant to the License Agreements.
- (b) System Access, Security, and Software Licenses.
- (i) Client will insure that appropriate primary and alternate means are available for Manatron support personnel to gain remote dial-in access to Client's system (when appropriately coordinated with Client).
- (ii) Client will maintain system passwords and will notify Manatron, prior to implementation, of any changes that may affect Manatron's ability to provide support under the License Agreements and this Description of Maintenance Services.
- (iii) Client will maintain a record of all user workstations running any portion of the licensed or sublicensed software (including any associated Internet applications). Client will provide this information to Manatron upon request and will advise Manatron of any changes in the system that affect the currency of this information.

(The rest of this page has been intentionally left blank)



CERTIFICATE OF LIABILITY INSURANCE

DATE (MM/D/D/YYYY) 03/28/2016

THIS CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CONFERS NO RIGHTS UPON THE CERTIFICATE HOLDER. THIS CERTIFICATE DOES NOT AFFIRMATIVELY OR NEGATIVELY AMEND, EXTEND OR ALTER THE COVERAGE AFFORDED BY THE POLICIES BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTITUTE A CONTRACT BETWEEN THE ISSUING INSURER(S), AUTHORIZED REPRESENTATIVE OR PRODUCER, AND THE CERTIFICATE HOLDER.

IMPORTANT: If the certificate holder is an ADDITIONAL INSURED, the policy(les) must be endorsed. If SUBROGATION IS WAIVED, subject to the terms and conditions of the policy, certain policies may require an endorsement. A statement on this certificate does not confer rights to the

the cer	terms and conditions of the policy, tilicate holder in lieu of such endors	certa! emeni	n policies may require an ei i(s).		tement on th	is certificate does not co	nter rights to th	16
PRODI	JCER			CONTACT NAME:				
	rsh USA, Inc. 36 Avenue of the Americas			PHONE (A/C. No. Ext):	7.77	FAX (AJC, No):		
Nev	v York, NY 10036			E-MAIL ADDRESS:		***************************************		
Aftr	n: NewYorkCerts@marsh.com Fax: 212-345	-3695		T	SURER(S) AFFOR	DING COVERAGE	NAIC#	
				INSURER A : ACE Amer			22667	***************************************
INSUR	ED	-m-manprop	212-201-201-4-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1	INSURER B : ACE Propi			20699	*************
Me	nairon, Inc.,			INSURER C : ACE Fire			20702	
	homson Reuters Business East Milhem Avenue			INSURER D :	***************************************		******	**********
	lage, Mi 49002			INSURER E :)+dooseooseoseoseoseeseespecc ap ygo			***************************************
				INSURER F:		**************************************	**************************************	www.wa
COV	ERAGES CER	TIFIC	ATE NUMBER:	NYC-007550945-14		REVISION NUMBER:6		************
TH	IS IS TO CERTIFY THAT THE POLICIES DICATED. NOTWITHSTANDING ANY RE RITIFICATE MAY BE ISSUED OR MAY CLUSIONS AND CONDITIONS OF SUCH	QUIRE PERTA POLICI	MENT, TERM OR CONDITION IN, THE INSURANCE AFFORD ES. LIMITS SHOWN MAY HAVE	OF ANY CONTRACT ED BY THE POLICIE	FOR OTHER I	DOCUMENT WITH RESPEC D HEREIN IS SUBJECT TO	T TO WHICH TH	iis I
INSR LTR	TYPE OF INSURANCE	ADDL S	UBR POLICY NUMBER	POLICY EFF (MM/DD/YY)Y)	POLICY EXP (MM/DD/YYYY)	LIMITS	aramakritiroori kikkaa maan kilirily, optoppii eesoo obaliisi ekseen	
	X COMMERCIAL GENERAL LIABILITY	WINDLY V	HDO G27405125	03/31/2016	03/31/2017		~~~~~~ ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	0,000
-	CLAIMS-MADE X OCCUR					DAMAGE TO RENTED	and the state of t	0,000
}	- CONTROLINATE - OCCUPA					Participal Control of the Control of	·	0,000
1	**************************************						*	0,000
į						**************************************		0,000
	GEN'L AGGREGATE LIMIT APPLIES PER: X POLICY PRO LOC		1			derlaindadascascoacoacoacoacoacoacoacoacoacoacoacoacoac		0,000
1						president properties and the second s	\$ 6,00	0,000
	OTHER: AUTOMOBILE LIABILITY	-	ISA H09040705	03/31/2016	03/31/2017	COMPANIES OF STREET		0,000
^				1			5 \$	·········
	X ANY ALTO SCHEDULED				1		\$ ************************************	***************************************
. 1	AUTOS AUTOS NON-OWNED AUTOS AUTOS					55558575575	\$	***************************************
	HIRED AUTOS AUTOS						3 ^.4************************************	
		├ ──	XOO G27963675 001	03/31/2016	03/31/2017		**************************************	••••••••••••••••••••••••••••••••••••••
В	X UMBRELLA LIAB X OCCUR		ACAC G27903015 001	00/3/1/2010	03/01/2017		7	0,000
	EXCESS LIAB CLAIMS-MADE							0,000
	DED RETENTIONS	-	WLR C48601650 (AQS)	03/31/2016	03/31/2017	·	\$	
A	WORKERS COMPENSATION AND EMPLOYERS' LIABILITY Y/N		RSC C48601662 (WI)	03/31/2016	03/31/2017	X PER CTH-		
Ç	ANY PROPRIETOR/PARTNER/EXECUTIVE N	N/A	THE CHOOL BOX (114)	09/0/1/2010	00/01/2011	330/300446-300407-44-4-4-4-4-4-4-4-4-4-4-4-4-4-4-4-4		0,000
	ANY PROPRIETOR/PARTNER/EXECUTIVE NOFICER/MEMBER EXCLUDED? (Mandatory in NH) it yes, describe under DESCRIPTION OF OPERATIONS below					E.L. DISEASE - EA EMPLOYEE		0,000
	DESCRIPTION OF OPERATIONS below	-	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,			E.L. DISEASE - POLICY LIMIT	\$ T ₁ 00	0,000
Ì								
Ì								
			, and the same and			MANAGEMENT TO THE REAL PROPERTY OF THE PARTY		**************
Fort 8	CRIPTION OF OPERATIONS / LOCATIONS / VEHIC Bend County, Texas is included as an Additional insu er of Subrogation is applicable where required by wri	red (exce	apt workers' compensation) where requir	ule, may be attached if mo red by written contract.	re space is requi	red)		
	RTIFICATE HOLDER	·····		CANCELLATION			-	
F(ort Bend County, Texás D1 Jackson Street Ichmond, TX 77499	***************************************	**************************************	SHOULD ANY OF	THE ABOVE D	DESCRIBED POLICIES BE CA EREOF, NOTICE WILL B CYPROVISIONS.		
				AUTHORIZED REPRES of Marsh USA Inc.	ENTATIVE		·	
	1			Michaela Grasshoff		mas anium	Judde	
lasar-								

© 1988-2014 ACORD CORPORATION. All rights reserved.

AGENCY CUSTOMER ID: 101	lß	1	m	1	ID:	₩ P	OB.	197	C	CV	CEN	۸
-------------------------	----	---	---	---	-----	-----	-----	-----	---	----	-----	---

LOC#: New York

A	$CORD^{\circ}$	
~ {		

ADDITIONAL REMARKS SCHEDULE

Page 2 of 2

AGENCY Marsh USA, Inc.	**	NAMED INSURED Manairon, Inc., a Thomson Reufers Business 510 East Milham Avenue
POLICY NUMBER		510 East Miham Avenue Portage, MI 49002
CARRIER	NAIC CODE	
ADDITIONAL REMARKS		EFFECTIVE DATE:
THIS ADDITIONAL REMARKS FORM IS A SCHEDULE TO ACO	RD FORM.	
FORM NUMBER: 25 FORM TITLE: Certificate of Lia	bility Insura	nce
Each of the Insurance policies referenced above provides that should such policy be care	celled by the insur- der, but failure to p	er before the expiration date of the premium, thereof for any reason other than non-payment of premium, the provide such notice shall impose no obligation or liability of any kind upon the insurer or its agents or



CERTIFICATE OF LIABILITY INSURANCE

DATE (MM/DD/YYYY) 04/07/2016

THIS CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CONFERS NO RIGHTS UPON THE CERTIFICATE HOLDER. THIS CERTIFICATE DOES NOT AFFIRMATIVELY OR NEGATIVELY AMEND, EXTEND OR ALTER THE COVERAGE AFFORDED BY THE POLICIES BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTITUTE A CONTRACT BETWEEN THE ISSUING INSURER(S), AUTHORIZED DEDDESENTATIVE OR PRODUCER. AND THE CERTIFICATE HOLDER.

REPRESENTATIVE OR PRODUCER, AND THE CERTIFICATE HOLDER.				
HADDITANT, If the codificate helder is an ADDITIONAL INSURED, the	policy(les) must be	endorsed. If SUBROGATION IS	WAIVED, subject to	
the terms and conditions of the policy, certain policies may require an electificate holder in lieu of such endorsement(s).	ndorsement. A stat	ament on this conficate does no	ot conter rights to the	
certificate holder in field of such endorsement(s).	CONTACT		20 40 to	
Marsh USA, Inc.	NAME: FAX			
1166 Avenue of the Americas New York, NY 10036	(A/C, No, Ext):	- Commence Add Market	ND):	
Attn: NewYork.Certs@marsh.com Fax: 212-345-3695	ADDRESS:	URER(S) AFFORDING COVERAGE	MAIN	
The state of the s	INSURER A : ACE Ameri	22867		
THE THE PARTY OF T		ACCI Algeria and Aministra	Sealth to ur e	
Maraton, Inc.,	INSURER B			
a Thomson Reuters Business 510 East Milham Avenue	INSURER C:			
Portage, MI 49002	INSURER D:			
	INSURER E :			
COVERAGES CERTIFICATE NUMBER:	NYC-007909913-14	REVISION NUMBER	**************************************	
TUO IO TO CERTIEV THAT THE POLICIES OF INSURANCE LISTED BELOW HA	AVE BEEN ISSUED TO	THE INSURED NAMED ABOVE FO	R THE POLICY PERIOD	
INDICATED. NOTWITHSTANDING ANY REQUIREMENT, TERM OR CONDITION CERTIFICATE MAY BE ISSUED OR MAY PERTAIN, THE INSURANCE AFFORD EXCLUSIONS AND CONDITIONS OF SUCH POLICIES, LIMITS SHOWN MAY HAVE	I OF ANY CONTRACT DED BY THE POLICIE E BEEN REDUCED BY	OR OTHER DOCUMENT WITH RES S DESCRIBED HEREIN IS SUBJECT PAID CLAIMS,	SPECT TO WHICH THIS I	
INSR TYPE OF INBURANCE INSD WYO POLICY NUMBER	POLICY EFF (MM/DD/YYYY)	I DOUGOVEAN I	LIMITS	
COMMERCIAL GENERAL LIABILITY		EACH OCCURRENCE	\$	
CLAIMS-MADE OCCUR	April 1	DAMAGE TO RENTED PREMISES (Ep. occurrence) <u>S</u>	
410,00	1	MED 6XP (Any one person) \$	
94444		PERSONAL & ADVINJUR	Y 8	
GEN'L AGGREGATE LIMIT APPLIES PER:		GENERAL AGGREGATE		
POLICY PRO: Loc		PRODUCTS - COMP/OP A	GG \$	
DTHER:			\$	
AUTOMOBILE LIABILITY		COMBINED SINGLE LIMIT (Ea acoldent)	\$	
ANY AUTO	1	BODILY INJURY (Per pers		
ALLOWNED SCHEDULED AUTOS AUTOS		BODILY INJURY (Per accid	dent) \$	
HIRED AUTOS AUTOS AUTOS		PROPERTY DAMAGE (Per accident)	\$	
First State			\$	
UMBRELLA LIAB OCCUR	,	EACH OCCURRENCE		
EXCESS LIAB CLAIMS-MADE		AGGREGATE		
DED RETENTIONS			\$	
WORKERS COMPENSATION		PER STATUTE EF	H	
AND EMPLOYERS' LIABILITY Y/M ANY PROPRIETOR/PARTNER/EXECUTIVE		E.L. EACH ACCIDENT	\$	
OFFICER/MEMBER EXCLUDED?		E.L. DISEASE - EA EMPLO	DYEE \$	
If yes, describe under DESCRIPTION OF OPERATIONS below		E.L. DISEASE - POLICY L	IMIT \$	
A Prof Liab. E & O EON G21635029 016	03/31/2016	03/31/2017 Limits	5,000,000	
		SIR	1,500,000	
			•	
DESCRIPTION OF OPERATIONS / LOCATIONS / VEHICLES (ACORD 101, Additional Remarks School	dule, may be attached if mo	re space is required)		
CERTIFICATE HOLDER	CANCELLATION			

Fort Bend County, Texas 301 Jackson Street Richmond, TX 77469	SHOULD ANY OF THE ABOVE DESCRIBED POLICIES BE CANCELLED BEFORE THE EXPIRATION DATE THEREOF, NOTICE WILL BE DELIVERED IN ACCORDANCE WITH THE POLICY PROVISIONS.			
	AUTHORIZED REPRESI	•		
	Morgan Frick	Morgan	Soick.	

© 1988-2014 ACORD CORPORATION. All rights reserved.
The ACORD name and logo are registered marks of ACORD

Exhibit 3



2290 Lucien Way #330, Maitland FL, 32751. 330-319-5393 rdelia@harriscomputen.com

ATTN: Paul Herrera / Alicia Losoya
Fort Bend County Clerk
301 Jackson, Suite 201
Richmond, TX
paul.herrera@fortbendcountytx.gov
alicia.losoya@fortbendcountytx.gov

To Whom it May Concern:

On behalf of Manatron, Inc., and the greater family of Harris Recording Solutions I wish to thank Fort Bend County for its continued use of our suite of records management software and services.

Manatron's Aumentum Recorder suite of records management software is currently deployed in the Fort Bend County Clerk's office.

Manatron utilizes a proprietary integration program to tightly integrate workflow processing within Aumentum Recorder under a common user interface. Support of these integrated components is available only through Manatron to Fort Bend County on a sole source basis. Enhancements and augmentation to the core software or its integration components may only be obtained from Manatron on a sole source basis. Manatron does not support authorized distributor or dealership programs and supports the end users of its product directly.

Harris acquired Manatron, Inc and all its intellectual properties in 2019; where we continue to proudly operate with our wonderful team of dedicated associates from the great states of Texas and Florida.

In addition, Manatron offers a menu of specialized services that utilize domain knowledge to access Aumentum Recorder software modules and program code.

Thank you again for the opportunity to serve Fort Bend County.

Yours truly,

Q.ADQ.
ROSS A. D'ELIA
VP, OPERATIONS

This message was sent electronically via e-mail to the addressed recipient